A KD Framework in Football Data Analytics: A Value Co-creation Framework for the Use of Knowledge Discovery Technologies in the Football Industry

A doctoral thesis submitted in partial fulfillment of the requirements for the award of Doctor of Philosophy of Loughborough University

By Khaled Shaban Department of Computer Science Loughborough University

Supervisors

Doctor Christian W Dawson And Doctor Shaheen Fatima

February 2019 © by Khaled Shaban 2019

ACKNOWLEDGEMENT

To Allah, God, thank you for helping me with strength, willingness, efforts and patience during my studies.

To my mother, father, wife, kids, friends, and family for all their prayers, support and help.

To my country and sponsors for allowing me to have this opportunity, experience, trusting me, and supporting me.

To my supervisor Doctor Christian W Dawson for his continuous support and valuable feedback.

To the Sport authorities, football federations, coaches and individuals who supported me. To Loughborough University for offering me the best research and study experience.

To my Friends, I can not name them all, the ones who aid and helped me reaching the expertise individuals needed for the success of my research.

To everyone that supported me during my research studies by any means.

I would like to specially thank,

My mother, Shadyah, for her prayers and continuous support, and I wish my success returns just part of her kindness and love that surrounded me. My lovely soulmate, Doctor Abeer Almakky for her support, advice and courage in my life. My father Professor Ghazy Almakky for his advice, courage and support during my studies. His words and wisdom were always there to support me. To my brothers and sisters, especially Ibrahim. I would like to thank my greatest boys, Mr Rayan and Mr Omar for allowing me the time to work on my thesis, and hopefully compensating them for the time and effort focusing on my studies away from them.

Thank you all very much.

ABSTRACT

Investment in sport technologies are expected to grow by 40.1% during 2016-2022 reaching approximately \$3.97 billion by 2022. As well the recent changes in technology regulations by The Federation Internationale de Football Association (FIFA) since the 2018 World Cup created promising football technologies. This research questions addressing the issue of what is the value of such technologies for professional football teams? and what are the benefits of these technologies? This is achieved by developing a framework for understanding the value co-creation process from the knowledge discovery systems in the football industry. The framework aids in mapping the resources, pinpointing the outputs, identifying the competencies leading into capabilities, and finally in realisation of the value of the final outcomes in that journey. On another words, different teams have different resources that allow them to achieve certain outputs. These outputs enable the coaching team to achieve and maintain certain abilities. By changes in practice the will improve the team ability and enhance their analytical capabilities. Therefore, that will allow and aid the coaching team to gain new outcomes such as improving training strategies, transferring players, and informative match strategies. Additionally, improved understanding of the value co-creation process from the knowledge discovery systems in the football industry answering, why are some teams better able to gain value from investment in knowledge discovery technologies than other teams in the football industry. The framework has been developed in three phases in which semi-structured interviews where used in the first and second phases for developing and validating the framework respectively. The third and final phases is verifying the framework by developing a knowledge discovery maturity model as an online assessment's tool in operationalising the research findings. The main contributions of this research are the adaptation and customisation of Melville *et al.* (2004) to develop a value co-creation process form knowledge discovery resources. Moreover, applying Agile (APM, 2015) artefacts and techniques and tools in improving the value co-creation process between coaches and data analysts. That's aided in developing the value cocreation knowledge discovery framework in football analytics. Additionally, the development of a key performance indicators balanced scorecard and its adaptation as a in understanding the relationships between the key performance indicators (i.e. physical, psychological, technical and tactical performance indicators). Finally, the development of the knowledge discovery maturity model in football analytics which was used in understanding and pinpointing areas of strength and weakness in the utilisation of the various football resources used in football analytics (human resources, technological resources, value co-creation resources and analytical models used).

GLOSSARY

Term	Details	Definition (as defined by this research)
KD	Knowledge Discovery	A systematic iterative procedure intended to enable the development of models and frameworks that can be used to study specific phenomena.
KDR	Knowledge Discovery Resources	KD resources used in the KD framework (i.e. technological resources, human resources, value co-creation resources and their sub-models.
КРІ	Key Performance Indicators	Are used to measure players performance actions based on the related events or actions during a match or training.
MM	Maturity Model	A multi-phases assessment model to reflect on current practices.
KDMMFDA	Knowledge Discovery Maturity Model	The KDMMFDA developed in this research to assess the coaching team FDA strength and weakness.
СТ	Case Team	-
FDA	Football Data Analytics	-
PA	Performance Analysis	-
KDV	Knowledge Discovery Value	-
User stores	-	are a technique for the development of the features required to meet the specific goals of users
Sprint	-	A method used to improve communication and gathering requirements.
Retrospective	-	It a technique for maintaining the iterative need between the questions, research and result so that level or collaboration is achieved as will leading enhancing maturity level in the collaboration process over time.
Transfer Strategy	-	The mechanisms that a team manager utilises to select and recruit players in fulfilment of the aim of achieving his overall objectives.
Match strategy	-	The process of using a match model to understand the strength, weaknesses, opportunities and threats (SWOT) of opponents in the match environment, in order to help the team winning the match.
Tactical KPIs	-	Are metrics that are intended to measure the ability of players to position themselves

		effectively in such a way the probability of passing, possessing, scoring and intercepting are improved.
Technical KPI	-	It is the ability to control the ball for the sake of accomplishing the required tasks effectively and efficiently.
Physical KPIs	-	Are those physiological and fitness measures for the players' abilities. Some of them are traits that cannot be changed, such as the height and ambidexterity while others can be improved by training such as speed, high/moderate intensity running and recovery rate.
Psychological KPIs	-	Referees to the ability to play in the standard performance under different psychological pressures, which can be called "resilience indicator".
Predictive models	-	Are equations or estimations used to estimate the probability of scoring in a variety of different situations.
Context- based modelling		Is defined in this research as the identification and measurement of the players' KPIs in different training and match context
Comparative modelling		Is a statistical method for the comparison between players or teams utilising different KPIs.
Synergetic modelling		Is a technique used to identify the correlation in a player performance with others in the team

Table of Case Studies

Wave One – Developing the framework

Code	Team / Bodies	Role
W1TD	Football National Teams	Technical Director & Expert
		(Coach, Player in Different
	· · · · · · · · · · · · · · · · · · ·	Leagues)
W1FC2	Football Club - 1 st Team	Coach
W1DA3	Football Club - 1 st Team	Analysts
W1DA4	Football Club 1 st Team	Analysts
W1FC5	Football Club - Olympic Team	Head Coach
W1DA6	Football Club - Olympic Team	Analyst
W1FC7	National Olympic Team	Coach
W1FC8	Football Club - 1 st Team	Coach
W1FC9	Football National Teams	Assistant Coach
W1DA10	Football Club 1 st Team UK	Analysts
W1BM11	Football Organisation - UK	Performance Analysis Team
		Member
W1BM12	Football Organisation - UK	Performance Analysis Team Member
W1BM13	Football Organisation - KSA	Technical Committee
W1RS14	Rugby Club	Director of Performance Analysis
W1PSC15	Sports Data Consultancy	Data/Video Analysts
W1BM16	Football Organisation - KSA	Technical Committee
W1PSC17	Sports Data Specialists - UK	Representative - Sports Data Specialist
W1PSC18	Sports Data Specialists - International	Representative - Sports Data Specialist
W1PSC19	Sports Consultancy - KSA	Manager – Football Data Specialist
W1PSC20	Sports Consultancy - KSA	Representative - Football Data
WIDGC21	Sports Data Specialists	Specialist
W1P5C21	Sports Data Specialists	Live Scouling Administration

Code	Team / Bodies	Role
W2TD1	Football National	Team Director – Former Player – Former
	Teams	Coach
W2EM2	Football Federation	Executive Manager of the Technical
W2TCM3	Football Federation	Technical Committee Member – Professional coach
W2FC4	Football Federation - Football Club Academy	Professional Coach – Academy Director – Former National Team Coach
W2FC5	Football Federation - Football Club Academy	Professional Coach – Academy Director – National Youth Team Coach – Scouting & Talent Identification
W2FC6	Football Federation - Football Club Academy	Professional Coach – Professional Player Mentor
W2FC7	The UK Football Association	Professional Coach
W2FTM8	The UK Football Association	UK Team Manager
W2SC9	University	Principal Lecturer in Sports Coaching Science
W2TD10	Football Federation	Ethics Discipline Committee

Wave Two – Validating the framework

Wave Three – Verifying the framework

Case	Code	Team	Role	Team / Bodies
Case 1	C1P1	CT1	Video Analyst	Football Club - 1 st Team
Case 2	C2P1 C2P2	CT2	Assistant Coach Data Analyst	U19 National Team
Case 3	C3P1	CT3	Data Analyst	National Olympic Team
Case 4	C4P1	CT4	Assistant Coach	Football Club - 1 st Team
Case 5	C5P1	CT5	Football Coach	Football Academy Club

KEYWORDS

Football data analysis, Performance analysis, value co-creation in football, competences and capabilities in football team, Resource basest view in football, football coaching, football data analysis, football data analytics, KPI Balanced scorecard.

Table of Contents

Chapter 1	I Introduction, Research Question, Aim and Objectives 1
1.1	Introduction
1.2	Research Problem 1
1.3	Research Motivations
1.4	Research Question
1.5	Research Aim 4
1.6	Research Objectives 4
1.7	Research Main Contributions 4
1.7. Ana	1 Knowledge Discovery Value Co-Creation Framework for Football Data lytics - KDVCCFFDA
1.7.2 KDI	2 Knowledge Discovery Maturity Model for Football Data Analytics - MMFDA
1.8	Thesis Structure
Chapter 2	2 Literature Review
2.1	Introduction
2.2	Definition of Knowledge
2.3	Knowledge Discovery (KD) Definition
2.4	KD in the Football Industry
2.4.	1 Objectives of KD in Football 12
2.4.2	2 Performance Analysis in Football 14
2.5	Theoretical Development: Resource-Based View
2.5.	Proposed KD Value Framework in Football Data Analytics
2.5.2	2 Modelling Development Approached: Agile Value Co-creation Approach
2.5.	3 Agile Value Co-creation 39
2.6	KD Maturity Models (MMs) 40
2.6.	1 Maturity Model Purposes 41
2.6.2	2 Maturity Models Structure
2.6.	3 MM Approaches
2.6.4	4 Capability Maturity Model (CMM) 42
2.6.	5 Data Analytics Maturity Models 43
2.6.	6 Criteria for a "Good" Maturity Model 46
2.7	Synthesis of the Literature Review
2.8	Summary
Chapter 3	3 Research Methodology
	IX

3.1 Introduction	49
3.2 Research Paradigm	49
3.3 Research Strategy	51
3.3.1 First wave of interviews: Developing the framework	52
3.3.2 Developing the Maturity Model	54
3.3.3 The Second Wave of Interviews: Validating the Framework	55
3.3.4 Third Wave of Interviews: Verifying the framework	56
3.4 Research Quality	56
3.5 Research Ethics	58
3.6 Chapter Summary	58
Chapter 4 Knowledge Discovery Resources	60
4.1 Introduction	60
4.2 The KD Technological Resources Model	60
4.2.1 Hardware required to collect the data	61
4.2.2 Software: Knowledge Discovery System	62
4.3 The KD Human Resources Model	64
4.3.1 Consumer of the Knowledge: Coach Competences	66
4.3.2 Producer of the Knowledge: Data Analysts Competences	68
4.4 Proposed Value Co-creation Process Model	74
4.4.1 User Stories / User Question	
4.4.2 Stories mapping / Questions board (mapping)	
4.4.3 The Story/Question/ Analysis Card	
4.4.4 Sprint	81
4.4.5 Retrospective	87
4.5 Summary	88
Chapter 5 Knowledge Discovery Value	
5.1 Introduction	
5.2 Performance metrics of the Team manager	
5.3 The KD Outputs: The Role of KD in Developing Effective Strategies	
5.3.1 Transfer Strategy	
5.3.2 Training Strategy	
5.3.3 Match Strategy	
5.4 The KD Outputs: Balanced Key Performance Indicators Roadmap	100
5.4.1 Tactical KPIs	101
5.4.2 Technical KPIs	106
5.4.3 Physical KPIs	108
	Х

5.4.4	Psychological KPIs11	0
5.5 Th	e KD Outcomes: Analytic Models 11	1
5.5.1	Predictive and Simulation Models11	2
5.5.2	Context based Modelling 11	3
5.5.3	Comparative Modelling11	4
5.5.4	Synergetic Modelling 11	4
5.6 Kn Analytics	owledge Discovery Value Co-Creation Framework for Football Dat (KDVCCFFDA)11	ta 6
5.7 Kn (KDMMF	owledge Discovery Maturity Model for Football Data Analytic DA)	28 20
5.7.1	Ad hoc	20
5.7.2	Defined 12	21
5.7.3	Managed 12	23
5.7.4	Optimised 12	24
5.8 KI	DMMFDA – Table Overview 12	27
5.9 Su	mmary	1
Chapter 6 Verification	Knowledge Discovery Maturity Model – Analysis, Evaluation an 132	ıd
6.1 Int	roduction13	2
6.2 Ch	apter Methodology	52
6.3 Ca	ses Background 13	3
6.3.1	Case 1: CT1 – Background 13	4
6.3.2	Case 2: CT2 – Background 13	5
6.3.3	Case 3: CT3 – Background 13	5
6.3.4	Case 4: CT4 – Background 13	6
6.3.5	Case 5: CT5– Background 13	57
6.4 Ma	aturity Model Validation and Verification13	57
6.4.1	Data Analyst Competences 13	7
6.4.2	Coach Competences 14	1
6.4.3	Technological Resources 14	3
6.4.4	Value Co-creation	-5
6.5 Th	e use of KPIs in the Cases 14	-8
6.5.1	Physical Indicators14	-8
6.5.2	Technical KPIs14	.9
6.5.3	Tactical KPIs 15	1
6.5.4	Psychological KPIs15	2

6.5.5	KPI Balanced Scorecard	153
6.6	Use of Analytical Models	159
6.7	Validating the role of KD in creating effective strategies	160
6.7.1	Transferring Strategies	160
6.7.2	2 Training Strategies	160
6.7.3	Match Strategies	161
6.8	Case Studies Feedback 1	161
6.9	Summary1	161
Chapter 7	Discussion and Conclusion	164
7.1	Introduction 1	164
7.2	Revisiting the Research Question, Aim and Objectives	164
7.2.1	Research Question	164
7.2.2	Research Aim	165
7.2.3	Research Objectives	166
7.3	Contribution to Knowledge	168
7.4	Academic Implications	168
7.5	Recommendations	169
7.6	Research Challenges	171
7.6.1	Acceptance of research practices	171
7.6.2	2 The collaborative culture between higher education and sports bodies	171
7.6.3	Collaborations, Connections and Transparency	172
7.6.4	Cost: Inability to meet all coaches/data analysts	172
7.7	Research Limitations	172
7.7.1	Generalisability/applicability of findings	172
7.7.2	2 Objective testing of results	173
7.7.3	Quantifications of the impacts of different factors on performance	174
7.8	Future Research	174
7.8.1	Testing and Quantification of the research models	174
7.8.2	2 Replicability of the study on other contexts	175
7.8.3 logic	Investigating the institutionalisation process of the knowledge discoves in the football industry	ery 175
7.8.4 lesso	Designing a new software application to manage the communications a new software discovery process	and 176
7.8.5 Disco	Benefits management techniques and approaches in Knowled	dge 176
Reference	es1	177

Appendix (Questionnaire)	

List of Figures

Figure 2-1: Proposed KD Value Framework in Football – © by the researcher	25
Figure 2-2: KD Technological Resources – © by the researcher	26
Figure 4-1: IT Resources model for enabling knowledge discovery	61
Figure 4-2: A model to show the relationship between different software applicat	ions in
discovering knowledge	64
Figure 4-3: A model to explain the role of different competences for disco	overing
knowledge for football teams	65
Figure 4-4: Pros and Cons of data interpretation by the data analyst	74
Figure 4-5: The proposed Value Co-creation process	76
Figure 4-6: Sprint Model for KD value co-creation framework	83
Figure 5-1:KD Value	90
Figure 5-2: The role of the Knowledge Discovery in formulating team manager str	ategies
Model 92	
Figure 5-3: The role of the Knowledge Discovery Indicators on the match model	100
Figure 5-4: KPIs BSC	101
Figure 5-5: Value Co-creation KD Framework for football coaching	119
Figure 6-1: Chapter Structure	132
Figure 6-2: Data Analysis Competences	141
Figure 6-3: Coach Competences	143
Figure 6-4: Technologies Used in Knowledge Discovery in Football	145
Figure 6-5: Use of Value Co-Creation Tools	148
Figure 6-6: Use of Physical KPI	149
Figure 6-7: Use of Technical KPIs	151
Figure 6-8: Use of Tactical KPIs	152
Figure 6-9: Psychological Indicators	153
Figure 6-10:The Use of Analytic Tools	159

List of Tables

Table 2-1: Sample of events from (Stein et al., 2017c)	20
Table 2-2: Push versus value co-creation production created by the researcher	34
Table 3-1: Wave 1 - Initial Participants – Framework Development	53
Table 3-2: Wave 2 - 2nd Participants (Framework Validation)	55
Table 3-3: Wave 3- 3rd Participants (Verification and MM - Application)	56
Table 3-4: Summary of research qualities	58
``Table 4-1: Summary of key findings regarding the role of the coach competence	ces on the
effectiveness of knowledge discovery process	66
Table 4-2: List of competences required by the data analyst to have an	effective
knowledge discovery process	69
Table 4-3: Generic Agile Story Template	77
Table 4-4: Proposed Agile Story Template	77
Table 4-5: Example 1 - Story/Question Card	80
Table 4-6: Example 2 - Story/Question Card	80
Table 4-7: Example 3 - Story/Question Card	81
Table 4-8: Example of Backlog file	86
Table 5-1: Summary of value-based process to transfer players	93
Table 5-2: Value based transfer process: Transfer for match results	94
Table 5-3: Sample of noted tactical KPIs from the interviews	105
Table 5-4: Sample of noted technical indicators from the interviews	108
Table 5-5: Sample of noted physical indicators from the interviews	110
Table 5-6: Sample of noted psychological KPIs from the interviews	111
Table 6-1: Summary of Cases Strength vs Weaknesses points	134
Table 6-2: Experts Evaluations regarding KPIs BSC	158

Chapter 1 Introduction, Research Question, Aim and Objectives

1.1 Introduction

Investments in sports technology exceeded \$1 billion in 2015 (Magee, 2015). However, it is unclear whether football teams are able to realise the expected value from these technologies, or what reasons could explain some teams obtaining a greater benefit than their competitors. This chapter introduces the current research project into this issue. It begins by identifying the research problem, based on real life needs, then provides a brief outline of the key literature to identify what is known and the gaps in formal understanding of the variation in realising value from investment in technologies. Based on the knowledge gaps and the research problem, the following section the research question, aim and objectives are presented, and supplemented with an overview of the overall structure of this thesis.

1.2 Research Problem

Sports technology is believed to improve team performance by enabling superior measurement, monitoring and planning of performance. According to the Analytical Research Cognizance (Infoholic Research, 2016), the Worldwide Sports Analytics market is expected to grow by 40.1% during 2016–2022, reaching approximately \$3.97 billion by 2022. Other studies have predicted that the sports analytic field will experience growth from \$764.3 million in 2016 to \$15.5 billion by 2023 (Infoholic Research, 2016), with wearable technologies predicted to grow annually by 20% to reach \$29 billion market value by 2022 (CCS Insight, 2018). PlayerTek a company, recently acquired by Catapult, experienced a 365% increase in the market of amateur and semi-professional football clubs in Europe and Asia during 2017 (Lemire, 2018). Catapult estimates that 10,000 elite teams are being targeted by football wearables devices in 2018. Indeed, hundreds of start-up companies have been established to serve this market, including major players like Nike, Adidas, Amazon and Microsoft at the Hype Foundation event (BBC News, 2017). Although the producer side of these technologies has already demonstrated exceptional profitability, with many companies doubling their profits semiannually, the value of such technologies is less clear from the consumer side. What is the value of such technologies for professional football teams? How can teams benefit from such technologies? Can all teams obtain the same value from investment in such technologies? This research seeks to provide new insights into these questions.

1.3 Research Motivations

In order to address the aforementioned questions, this research adopts a Knowledge Discovery (KD) approach, predicated upon the main argument that knowledge is a key power that could enable coaches to better understand the weaknesses and strengths of their teams, as well as those of their opponents. This kind of understanding would facilitate and support more effective planning for player transferring, training, and match strategies. These various insights may provide a way for teams to gain a competitive advantage. In other words, new knowledge may be crucial in optimising the performance of individual players and entire teams.

According to the information technology value framework, investment in technology only provides meaningful value to organisations when they have the suitable organisational resources, which are called organisational complementary resources (Melville and Kraemer, 2004). Accordingly, realising the expected value from the investment in technology requires two types of resources to be synchronised and orchestrated — technological and organisational resources (Piccoli and Ives, 2005; Badewi et al., 2018). Technological resources are the hardware and software required to enable changes in the practices. In this study, KD resources can be defined as the hardware and software applications necessary to discover new knowledge that could help coaches to improve their team match strategies. The technologies primarily addressed in the literature are wearable sports devices (Dellaserra, Gao and Ransdell, 2014), cameras (Ding and Fan, 2006; Sugimoto et al., 2012), GPS technologies (Rangsee, Suebsombat and Boonyanant, 2013), and body sensors (Marin-Perianu et al., 2013). All of these technologies create a massive amount of data every day (Rein and Memmert, 2016). These technologies have been studied in terms of their usability (Marin-Perianu et al., 2013), usefulness (Dellaserra, Gao and Ransdell, 2014), and effectiveness (Clemente et al., 2013; Rein, Raabe and Memmert, 2017). The summary of such findings is that no one universally agreed upon answer why some teams are better in using these kinds of technological resources. Additionally, to date, no dedicated research has been undertaken to provide a detailed explanation of the different wearable technologies that could help in discovering knowledge.

Technology alone is not sufficient to create value without the support of human resources. Relative few studies have examined human resources (in terms of skills, competences, and knowledge) as the main producer of such knowledge (i.e. the data analyst) (Wright *et al.*, 2013; O'Donoghue and Holmes, 2014; Wright, Carling and Collins, 2014; Scheider, Ostermann and Adams, 2017). Limited papers have focused on the resources required by the consumer of the knowledge (e.g. the coach). Clarification is also required about the relationship between the producer and consumer of this knowledge. Is it a 'give-and-take' selling process (i.e. data analyst produces the knowledge based on specifications and sell it to the coach)? Or it is a value co-creation process (i.e. data analyst shall work closely with the coach to develop new knowledge)? If it is a value co-creation process, what are the tools and methodology that could frame such process? The previous literature has not dealt with these issues.

Although KD relies on human and technological resources to create value to the coach by increasing the probability of winning matches, the mechanisms connecting resources and value are unclear. One way in which to connect the relationship between resources and benefits is through the use of benefits map frameworks (Ward and Daniel, 2012; Jenner and APMG International., 2014; Serra and Kunc, 2015; Badewi, 2016). However, benefits maps have not been used in sports analytics studies to explain the relationship between resources, mechanisms to deliver benefits, the required changes in practices, and the potential benefits of investment in such technologies.

This research aims to use software engineering tools in KD, as a knowledge gap also exists in maturity models for data analytics in football information systems. One of the first maturity models developed in information systems literature was by Churchill *et al.* (1969), which was subsequently improved by Nolan (1975) and McFarlan *et al.* (1983) to explain the ability of organisations to effectively adopt the applications of different information systems. Maturity models were then used in more than 150 information systems sub-domains (Chen and Nath, 2018), ranging from classic fields, like software development (CMMI), to contemporary areas, such as e-business (Pranicevic, Alfirevic and Stemberger, 2011), business analytics and KD (El-Gayar *et al.*, 2011). However, these studies focus on business applications, rather than on sports knowledge discovery applications and technologies. This contribution to knowledge could help technologists in sports science to develop tailored roadmaps for teams to improve their abilities to

compete through KD resources. Accordingly, this research aims to utilise the knowledge maturity model to assess the ability of teams to realise value from KD systems.

1.4 Research Question

Why are some teams better able to gain value from investment in knowledge discovery technologies than other teams in the football industry?

1.5 Research Aim

To develop a framework for understanding the value co-creation process from the knowledge discovery systems in the football industry.

1.6 Research Objectives

- 1- To operationalise the expected value of KD to the coaching team.
- 2- To identify and taxonomise the KD resources and design a model to understand the role of each class of football technologies in improving coaching performance.
- 3- To identify and frame the role of different knowledge, skills and competences required from the producer (i.e. data analyst) and consumer of the knowledge (i.e. coach) to enable the expected value from the KD to be realised.
- 4- To frame the value co-creation process and augment it with different tools to improve its value.
- 5- To develop resource-based maturity model to identify the weaknesses and strengths in the augmentation of the resources to get value from knowledge discovery activities.

1.7 Research Main Contributions

This section presents the two major contributions of the research which are, the Knowledge Discovery Value Co-Creation Framework for Football Data Analytics (KDVCCFFDA) and the Knowledge Discovery Maturity Model for Football Data Analytics (KDMMFDA). The detail of the initial development of the proposed KDMMFDA were introduced and discussed in Chapter 2, section 2.5. Then, the proposed KDVCCFFDA is improved and enhanced after interviewing experts in the sports industry as discussed in Chapter 4 and Chapter 5. The KDMMFDA then developed to operationalise the research findings in aiding and pinpointing areas of strength and weakness for teams or players in improving football data analytics. The following

sections is to present the main contributions of the research to aid in navigating to related materials within the thesis.

1.7.1 Knowledge Discovery Value Co-Creation Framework for Football Data Analytics - KDVCCFFDA

The first contribution of the research is in the development of the KDVCCFFDA which consists of 4 main models; the technological resources model, the human resources model, the value co-creation model, and the key performance indicators model. The framework was developed in three phases. The framework development phases are discussed in detail in sections 3.3. First, the initial framework which was developed from literature (see section 2.5). Then, the improvement and enhancement of the framework is discussed in Chapter 4 and Chapter 5 A compacted version of the final framework - KDVCCFFDA is presented as hyperlinked diagram for easily presentation and navigation to the related sections within the thesis as shown in **Error! Reference source not found.**



Figure 1-1: Hyperlinked Figure of the Framework - © by the researcher

The figures highlight the main models of the framework and the shows the transformation of utilising the KD resources, leading to outputs, aiding in improving the capabilities, to

outcomes that enable the realisation KD value. The detailed framework is presented here in Figure 1-2, as well to show an in-depth view of the models and sub models within framework.



Figure 1-2: Detailed KDVCCFFDA - © by the researcher

1.7.2 Knowledge Discovery Maturity Model for Football Data Analytics -KDMMFDA

The next contribution of this research in the development of the KDMMFDA in which aided in operationalising the research findings as well in assessing teams football data analysis. The KDMMFDA were proposed in Chapter 2 in section 5.7. it is based on 4 maturity levels (i.e. Ad-hoc, Defined, Managed and Optimised. It aids in understanding areas of improvements in human resources, technological resources, knowledge discovery co-creation and analytical models developed. An overview of the final Model can be seen in Chapter 5 section 5.7.

1.8 Thesis Structure

This thesis is organised into seven chapters. After introducing the research question, aim, and objectives in chapter 1, chapter 2 reviews the relevant literature and sets the theoretical foundations of this research. Chapter 3 presents the research methodological approach, strategy, and tools that this research uses to answer the research question and fulfil the research objectives. The analysis and findings are presented in three chapters. Two chapters outline the development and final versions of the chosen analytic approaches (knowledge discovery framework and its subset models and the maturity model) using data obtained from interviews with representatives of 7 teams in Saudi Arabia (13 coaches, team managers, and data analysts). Chapter 4 focuses on the identification and role modelling the technological and human resources required in delivering value. Then, chapter 5 operationalises the definition of KD value in the football sector. The main practical output of chapter 4 and 5 is the proposal of a knowledge discovery maturity model, which is validated using findings from 14 data analysts, team managers, and coaches from 5 teams. In fulfilment of this aim, chapter 6 discusses the application and validation of the KD maturity model. Finally, chapter 7 summarises and consolidates the research findings.

Chapter 2 Literature Review

2.1 Introduction

This chapter seeks to develop the theoretical basis for the current research through a comprehensive review of salient literature. The aim of this chapter is to provide a coherent and active review that not only engages with the relevant papers to illustrate the knowledge gap and the potential contribution of this study, as well as to critically engage with the literature to enable the construction of a new analytic framework and its concepts. In this way, the chapter provides a sound basis with which to inform data collection, particularly in terms of creating interview and codes for data analysis, and the discussion of findings. The literature approach tends to be a mixture of qualitative systematic review/qualitative evidence synthesis (Grant, M. J. and Booth, A. 2009) in which to understand and enrich and meet the scope and depth of the research.

In the following sections (sections 2.2 and 2.3), this literature review defines the concept of 'knowledge' within the context of the current research, enabling the discussion of Knowledge Discovery (KD) to be well-articulated and framed. This is followed, in section 2.4, by an overview of the main theories underpinning KD and its value in the context of modern sport. Based on these theoretical foundations, section 2.4 provides a focused review of the use of KD in the football industry, in order to clearly illustrate the knowledge gap and importance of the current research. As the literature does not agree on one specific framework with which to investigate the value realisation from KD in the football industry, a critical discussion is provided of the three analytic frameworks proposed for data collection and analysis: the value framework, which is adopted from IT Business literature (Melville and Kraemer, 2004); the value co-creation framework, from IT marketing literature; and the Agile framework, adapted from software engineering literature. Section 2.5 examines the value framework for KD in the specific football context, then outline the applicability and theoretical underpinnings of value-creation and Agile methods for KD in this context. Finally, the chapter (section 2.6) reviews the literature on the use of maturity models in software engineering and data analytics in order to inform the design of the model used in this research. Section 0 concludes the chapter with a summary of findings that outlines the critical review of literature in the context of the current study.

2.2 Definition of Knowledge

Comprehensive engagement with the process of "Knowledge Discovery" (KD) requires a thorough understanding of the hierarchy of knowledge. This concept refers to a means of conceptualising and defining the relationships between information. One such approach to categorising and analysing information in a hierarchical system is the DIKW (Data, Information, Knowledge, and Wisdom) pyramid (Rowley, 2007, p. 163). Data are objective, discrete, facts and consequently have no intrinsic meaning without being processed or organised (Awad and Ghaziri, 2004; Baškarada and Koronios, 2013; PhridviRaj and GuruRao, 2014). They are simple records of observations, events, things or actions (Boddy, Boonstra and Kennedy, 2005; Turban, Rainer and Potter, 2005; Laudon and Laudon, 2006), meaning that they require interpretation in order to gain meaning or applicability (Baškarada and Koronios, 2013).

Broadly speaking, many published sources agree that information is distinct from data. Essentially, data that has been aggregated and processed to offer insights and understanding of a particular topic can be thought of as information (Bocij *et al.*, 2015). In addition to inherent advantage of being synthesised and consolidated, information is valuable for comparison and prediction (Laudon and Laudon, 2006), as well as facilitating decision making (Awad and Ghaziri, 2004). As a specific example of this, information can be a useful tool to predict behaviour in a given context (Laudon and Laudon, 2006), or making data retrieval and processing simpler or more effective (Hevner and Chatterjee, 2010). The conversion of data into useful information is most often achieved through statistical, arithmetic, or algorithmic models (Laudon and Laudon, 2017), which produce information that is structurally different and is therefore useful in different kinds of applications.

Knowledge enables experts to understand and interpret information, thereby enabling them to give more accurate and meaningful opinions (Turban, Rainer and Potter, 2005; Bocij *et al.*, 2015). Unlike data, which is a property of things, knowledge is a property of people, and therefore leads to certain behaviour (Boddy, Boonstra and Kennedy, 2005, p. 9), which means that the development of knowledge is invariably closely related to the interpretation of people and information. The construction of knowledge from information therefore requires some degree of subjective interpretation of background and communication (Chi, Slotta and De Leeuw, 1994). Data can be conceptualised using

information theory reasoning (Shannon, 1948; Mishra, Akman and Mishra, 2014). In effect, data can be thought of as patterns that have no meaning, while information is data that has been consolidated to give it meaning, and knowledge refers to "information incorporated in an agent's reasoning resources" (Aamodt and Nygård, 1995, p. 197). Finally, wisdom describes accumulated knowledge, which has changed behaviour, capabilities and practices of an individual, group or population (Jessup and Valacich, 2003). In other words, wisdom is the manifested capacity to utilise knowledge (Jashapara, 2011, pp. 17–18), realising the translation of knowledge into new practice. This is an abstract level that gives foresight (Awad and Ghaziri, 2004, p. 40).

2.3 Knowledge Discovery (KD) Definition

The current research defines the process of KD as a systematic iterative procedure intended to enable the development of models and frameworks that can be used to study specific phenomena. The KD process involves exploration, observation, description, analysis, synthesis and testing (Fayyad, Piatetsky-Shapiro and Smyth, 1996; Kwasnik, 1999). There are three principle approaches to KD: descriptive statistics, which are utilised to quantify facts; visual data mining, which enables relationships to be explored in depth; and machine learning, which aims to create the conditions, predictions and rules for given contexts, enabling specific scenarios to be simulated and tested (Bandaru, Ng and Deb, 2017).

In essence, KD describes the results obtained from extracting useful information from large quantities of data (Agrawal and Shafer, 1996). This has resulted in KD being perceived as an approach (Mariscal, Marbán and Fernández, 2010), a methodology (Alsultanny, 2011), and a technique (Hegland, 2001). However, KD differs from other approaches, including performance analysis (PA) (Garganta, 2009b), data mining (Chen and Liu, 2009), and big data (Kalambe, Pratiba and Shah, 2015; Rein and Memmert, 2016). KD seeks to discover and investigate knowledge (Bandaru, Ng and Deb, 2017), usually in general terms rather than in a single dataset. In contrast, data mining is concerned with investigating specific databases (Bramer, 1999; Larose and Larose, 2014), while big data focuses on the speed and volume of the data extracted from diverse sources (Sagiroglu and Sinanc, 2013). Therefore, KD can be understood as a higher level than big data, data mining and performance analysis, as it seeks to unearth knowledge about the adoption of technological platforms and systems (Wu *et al.*, 2014), in addition

to the creation or development of new models for defining, testing or understanding concepts or phenomena (Larose and Larose, 2014). In this sense, KD is a way for organisations to find ways to outperform their competitors (Manco *et al.*, 2016). In summary, KD describes a process for using tools and instruments to find knowledge and, in this sense, has a broader concept and function than data mining, performance analysis, or big data analysis.

2.4 KD in the Football Industry

The KD approach has had a profound impact on the performance of sports teams in numerous fields (Ofoghi *et al.*, 2013), through the use of a range of algorithms, including neural network, genetic algorithm, machine learning, and heuristic analysis (Rein and Memmert, 2016; Constantinou and Fenton, 2017; Hoch *et al.*, 2017). However, it is important to note that the current research is predicated upon a different understanding of KD than the majority of extant football literature, which principally focus only on the use of data mining tools, rather than specifically examining KD as a tool to inform data analysis and the improvement of coaching practices. This approach can be seen in a recent study that utilised a hierarchical KD approach to compute and measure key team performance indicators at the individual, group, tactical and team level (Hoch *et al.*, 2017). This enabled the development of a single unified analytic model may be more relevant in the field of data mining than KD, as the latter requires a focus on both qualitative and quantitative data, as well the process by which data and information are utilised.

2.4.1 Objectives of KD in Football

The literature on football coaching has examined the value of KD from a host of different perspectives. Analysis of team performance has been shown to a competitive advantage to the participating team (Kuper and Szymanski, 2018), resulting in higher win ratios (Wooster, 2013) due to the ability of informed coaches to make superior decisions (Groom, Cushion and Nelson, 2011; Wright, Atkins and Jones, 2012; Wright *et al.*, 2013). This section seeks to synthesise the literature and obtain a holistic view of the potential objectives of the KD approach. KD can support the effective storage and retrieval of historical data, granting the head coach access to new KPIs and even creating opportunities for additional, previously unknown, performance indicators to be

uncovered. These outcomes can enable the development of more effective, tailored coaching practices and strategies, improving team capacity and ultimately offering competitive advantage to the team.

2.4.1.1 To develop decisions based on Objectives and Evidence-based

KD can offer a systematic, objective documentation of data, which is essential to help coaches make better decisions, by removing individual bias, perception errors and subjectivity. Without this systematic documentation, analysing and documenting the data, coaches will use their own experience, which is called the critical incident technique because coaches remember only critical events not objectively all events (Flanagan, 1954).

The integration of technology into the coaching process is recognised as one way in which to foster objectivity and encourage fact-based decision-making. Proponents of this approach cite the fallibility of human memory, with some studies suggesting that most coaches can only accurately recall 30-50% of the events in the most recent games that they have watched (Franks and Miller, 1986, 1991; Laird and Waters, 2008). This limited recollection illustrates the inherent value of having a system like KD, which includes data collection and database technology, meaning that it is able to gather and store data about key events that occur throughout specific matches (Nicholls and Worsfold, 2016). Access to this kind of data can increase the effective memory of coaches to incorporate all KD-captured events in the team history. This can potentially encapsulate long periods of time (Franks and Miller, 1986, 1991), reducing subjectivity (Horton *et al.*, 2014) and the selective interpretation and perception of past events (Hughes and Franks, 2004), thereby enabling more accurate, effective decisions to be made.

2.4.1.2 To Create Change in Practices, and Match Strategies

The integration of the discipline of KD through data analysis into football has led to numerous examples of positive changes to the philosophies, approaches, and practices of coaches (Reep and Benjamin, 1968), with studies of statistical analysis showing measurable improvements in coaching practices and capabilities (Pollard, 2002; Doyle, 2007). An example of the positive impact of data analysis can be seen in the rejection of 'possession football', with analysis leading teams to adopt more direct strategies that involve fewer passes per team possession (Bate, 1988). This is evidenced in the new approaches used by professional and national teams, as illustrated by Olsen and Larsen

(1997), who studied the playing style of the Norwegian national team. They found that performance analysis had drastically changed the playing style of the team. This importance has been recognised by new research, such as the studies by the Carling research group, which documented the emergence of approaches and practices among professional clubs (Carling and Dupont, 2011a; Collet, 2013a; Carling *et al.*, 2014). For instance, the knowledge that players should maintain a minimum constant speed of 24 km/h during a game has made speed a preliminary selection criteria during recruitment (Carling and Dupont, 2011b).

2.4.1.3 To develop new KPIs and New Coaching Perspectives

KD resources track and measure a wide range of events and contexts, opening new opportunities for analysis. This has led to the design of new key performance indicators, which offer coaches deeper, more accurate insights into match performance dynamics (Hughes *et al.*, 2012; Mackenzie and Cushion, 2013; Wright, Carling and Collins, 2014; McLean *et al.*, 2017). New data models can also grant access to new coaching perspectives, potentially uncovering correlations between superficially unrelated factors and a better understanding of their influence at an individual and group level, leading to the effective development and improvement of overall team performance (Bampouras *et al.*, 2012; Sarmento, Pereira, *et al.*, 2014; Wright, Carling and Collins, 2014).

In essence, KD provides enhanced performance analysis to coaching teams, enabling new perspectives into match events to be captured and analysed, then processed in terms of different scenarios and contexts. This approach is based upon the notion that better understanding of player performance and the performance of their competitors, the more effective plans that can be developed, leading to improved results at a match level. Given this perspective, the following section will review the literature on performance analysis, after which a discussion will be provided of the kinds of strategic improvement that football teams might be possible to achieve through KD performance analysis.

2.4.2 Performance Analysis in Football

KD resources improve the effective use of Performance Analysis (PA) to improve coaching practices and obtain better match results. The inception of PA can be traced to the 1960s, with the analysis of team performance in American football and basketball, albeit through annotation and coding notes rather than technological approaches (Carling *et al.*, 2005; Hughes and Franks, 2007). However, PA has evolved from a simple, linear

process utilising descriptive schemas and flow charts (Cushion, Armour and Jones, 2006) to an advanced science that represents a complex, complementary interdisciplinary approach involving the use of sport, behavioural and data sciences are to obtain detailed insight into sport performance (Stein *et al.*, 2017a).

PA is a well-established tool in the world of elite sport (Wright *et al.*, 2013) and has become integral to modern football coaching approaches (Hodges and Franks, 2002; Stratton *et al.*, 2004; Carling, Williams and Reilly, 2005; Groom, Cushion and Nelson, 2011; Cushion and Lyle, 2016). This has led to substantial growth in research in this discipline (Lago-Ballesteros and Lago-Peñas, 2010). The majority of extant research in this discipline is focused on analysis methods (e.g. Lorains et al. 2013) or outputs (e.g. Liu et al. 2016), with PA becoming increasingly sophisticated and complex (Groom, Cushion and Nelson, 2011). Research trends are increasingly focused on modelling the interactions between specific KPIs, which will be discussed in more detail in the following section (Cushion, Armour and Jones, 2006). The complexity of these parameters and needs necessitate the use of KD, as a systematic analytical approach with which to explore interactions between known (in databases) and unknown variables (derived through special sensors or through deriving the current database). The next sections discuss and expound upon these known variables, that is the KPIs, and introduce the analytical models that are currently used.

2.4.2.1 Key Performance Indicators (KPIs)

Each key moment or action in a game that can be numerically documented is referred to as an event. Events record the action, location and time of each event, in addition to any other information deemed important. As an example of this, a football pass would be recorded as the action (passes), the location in the field (X, Y coordinates) and the time that the pass was made, as well as potentially any relation or effect of the pass on the game. Researchers refer to each of these events as a Key Performance Indicator (KPI). These parameters enable individual or team attributes to be described and evaluated (O'Donoghue, 2008; Parmar *et al.*, 2017), and are most commonly used to compare players or teams (Hughes and Bartlett, 2002; Collet, 2013b; Liu, M. A. Gómez, *et al.*, 2016). In some scenarios, KPIs can also predict the outcomes of matches (Min *et al.*, 2008; Huang and Chang, 2010), with the difference between teams that win and lose being visible in terms of differences in certain game-related KPIs (e.g. quality of opposition, total number of shots on goal, or quantity of ball possession) (Hughes and Franks, 2005a; Collet, 2013c; Winter and Pfeiffer, 2016; Tenga, Mortensholm and O'Donoghue, 2017; Fernandez-Navarro *et al.*, 2018a). There is a close relationship between KPIs and KD, with KD being used to analyse, understand and control phenomena identified from new attributes, and KPI informing the performance and behaviour of players during a given match. The identification of these KPI provides a better understanding and therefore improvement of individual and team performance, including valuable insights into opponents. It has been claimed that more sophisticated, customised KPI can also grant insights into the behaviour and performance of players, leading to more effective match planning and performance. Synthesis of literature indicates that KPIs are typically classified as being physical, technical, tactical or psychological indicators, all of which can help a coach to better understand a team and their needs (Parmar *et al.*, 2017).

2.4.2.1.1 Physical KPIs

Traditionally, physical attributes and fitness are the main indicators in the analysis of sports performance (Bloomfield, Polman and O'Donoghue, 2004; Woods *et al.*, 2016; Yang *et al.*, 2018). In terms of football match results, high intensity running with the ball is a critical factor (Randers *et al.*, 2010), although it should be noted that different positions have specific bioenergetics, physical, physiological requirements (Di Salvo *et al.*, 2010; John Moores and Reilly Building, 2010). As an example of this, match performance is influenced by the ability of the midfield players to run quickly over long distances, because of their vital link role between attackers and defenders (Hughes *et al.*, 2012; Vilar *et al.*, 2013). This difference can be seen in terms of the distance covered by elite midfield players (11.5km) in comparison to elite attackers and defenders (10-10.5 km) (Mohr, KRUSTRUP and BANGSBO, 2003).

Physical KPIs are concerned with profiling and understanding players in terms of a wide range of factors. These include, but are not limited to: maximum speed and recovery time (Carling, Le Gall and Dupont, 2012; Ndlec *et al.*, 2012; Collet, 2013c); and low speed, medium speed, or high speed running (Di Salvo *et al.*, 2009; Gregson *et al.*, 2010). Studies have examined player behaviour under fatigue, in order to examine whether, and to what extent, speed and reaction time are affected (Lyons, Al-Nakeeb and Nevill, 2006; Min *et al.*, 2008; Rampinini *et al.*, 2009; Ndlec *et al.*, 2012), including examination of factors

that include fatigue from overfilled calendars (Bradley *et al.*, 2014), from fasting during Ramadan for Muslim football players (Zerguini *et al.*, 2007), and the effects of fatigue among substitutes and replaced players (Carling *et al.*, 2014). Analysis suggests that the performance of football players is affected by a number of factors including short recovery periods between matches and even the playing formation, which had a larger effect on results than the impact of fatigue (Bradley *et al.*, 2011; Carling and Dupont, 2011b).

2.4.2.1.2 Technical KPIs

Technical capabilities describe the ability of players to control the ball (Hoernig *et al.*, 2016). Extensive investigations have been conducted into the relationship between this factor and physical KPIs (Kempton *et al.*, 2015), with studies identifying a positive correlation between technical KPI and physical KPIs in the prediction of match performance (Gibson Moreira *et al.*, 2015). Consistent with this logic, technical performance tends to significantly decline during the second half of a match (Rampinini *et al.*, 2009; Carling and Dupont, 2011b), with the last five minutes of games seeing the largest fall in technical skills due to the impact of fatigue on physical performance (Sarmento, Marcelino, M. Teresa Anguera, *et al.*, 2014).

Annotations capabilities are essential in understanding this relationship between physical KPIs and technical KPIs. This approach enables considerations like different dribbling styles to be annotated and analysed in terms of the success ratios of each KPI per player or per team. Another factor, which is one of most important technical KPI in football, is the ability to control the ball. Analysis shows that higher levels of control yield significant gains in the space and time available to perform a given action with the ball (Tenga, 2010; Hoernig *et al.*, 2016).

2.4.2.1.3 Tactical KPIs

In studies of sport, tactical KPIs are those indicators focused on match tactics and the successful implementation of planning (Tenga, 2010), including such considerations as successful constructive passes, team possession, ability to regain possession, and conversion rate (e.g. shots on goal to scoring a goal). There is a high correlations between the tactical KPIs and other KPIs. Hughes and Franks explained the relationship between the tactical KPIs and other KPIs in their book as follow:

"Tactical performance indicators seek to reflect the relative importance of the use of pace, space, fitness and movement, and how players use these aspects of performance, of themselves and their opponents, targeting the technical strengths and weaknesses of the respective performers" (Hughes and Franks, 2004, p. 175).

Tactical KPIs such as ball possession, (Collet, 2013c; Barreira, J. J. Garganta, *et al.*, 2014), crosses (De Baranda and Lopez-Riquelme, 2012), or interceptions and clearance (Lago-Peñas, Lago-Ballesteros and Rey, 2011; Barreira, J. Garganta, *et al.*, 2014; Woods *et al.*, 2016) can potentially be strong predictors for match results (Clemente *et al.*, 2014; Kempton *et al.*, 2015; Winter and Pfeiffer, 2016). For example, a study of the Norwegian professional football league found that tactical KPIs relating to possession, including such considerations as directness of offence or ability to penetrate opposite defence, were an effective way to distinguish teams (Tenga *et al.*, 2010b). These insights also provided meaningful strategic insights, such as the relative strength of counter-attacks versus complex attacking patterns when playing against an imbalanced defence. Meaningful gains in understanding can also be obtained from the use of KPIs to study certain actions in the context of success, like quantitative KPIs that study the importance of ball possession by examining the relative probabilities of goals being scored or conceded (Pollard and Reep, 1997).

Perhaps the most important tactical KPI in analysis of football is the passing accuracy and success (Hughes and Franks, 2005b; Tucker *et al.*, 2005; Taylor *et al.*, 2008a; Gyarmati, 2016; Rein, Raabe and Memmert, 2017). This is particularly important in the context of fatigue, with studies showing poorer tactical performance of Italian league players, as measured by differences between first and second half performance in terms of involvement with the ball, number of short passes and number of successful passes (Rampinini *et al.*, 2009), which negatively affected the amount of time that players spent in defence, midfield or attack (Lago, 2009; Ndlec *et al.*, 2012; Sarmento *et al.*, 2017).

2.4.2.1.4 Psychological KPIs

Physical, technical or tactical performance can vary widely depending on context and the mental pressure on players. These factors KPIs are known as psychological KPIs. As an example of this, game location has been shown to have a profound impact on technical, tactical and physical performance (Carling and Dupont, 2011b; Hughes *et al.*, 2012; Gibson Moreira *et al.*, 2015). Psychological KPIs can be used to explain why players
demonstrate significantly higher performance levels when playing at home, due to the 'home advantage' (Jacklin, 2005; Hughes *et al.*, 2012; Škegro, Milanović and Sporiš, 2012; Collet, 2013b; Liu, M. A. Gómez, *et al.*, 2016; Den Hartigh *et al.*, 2018; Fernandez-Navarro *et al.*, 2018a). The reduced stress involved in playing at home can be reflected through higher numbers of goals scored (Poulter, 2009; Armatas and Pollard, 2014; Bialkowski *et al.*, 2014); more shots being taken on goal (Taylor *et al.*, 2008b; Lago-Peñas, Lago-Ballesteros and Rey, 2011; Sampaio *et al.*, 2012); more crosses (Lago-Peñas, Lago-Ballesteros and Rey, 2011; Sampaio *et al.*, 2012), more successful passes, more successful dribbles and more corners (Lago and Martín, 2007; Poulter, 2009; Collet, 2013b), as well as improved discipline, in terms of fewer fouls (Poulter, 2009) and fewer yellow cards (Lago-Ballesteros and Lago-Peñas, 2010; Liu *et al.*, 2015)..

2.4.2.2 Analytics Models

Analytic models are algorithms that can describe or explain, certain solutions or behaviours (Hazır, 2015). The three main type of analytic models are: descriptive analytics, which focuses exclusively on frequency and ratios; context analytics, which measures sensitivity to different contexts; and predictive analysis, which utilises the other approaches to simulate likely match outcomes in particular scenarios (Sarmento, Marcelino, M Teresa Anguera, *et al.*, 2014). A brief discussion of these models is given below.

2.4.2.2.1 Descriptive Analytics

Event data is the lowest level of analysis in football performance management. This kind of data focuses on those actions that are "match-relevant and happening during the match" (Stein *et al.*, 2017b). From a technical perspective, events are timestamped occurrences of known and defined categories that are deemed important in that context. They may also be annotated with spatial coordinates or additional information, such as the players involved in the event. These data are then saved in semi-automatic coding video analysis applications (Tovinkere and Qian, 2001; Xie *et al.*, 2002; Assfalg *et al.*, 2003a; Ekin, Tekalp and Mehrotra, 2003; Xu *et al.*, 2008). A popular system for the classification of events is in terms of actions taken on the ball and those that occur off the ball (Maksai, Wang and Fua, 2015; Kamble, Keskar and Bhurchandi, 2017). Another approach is to classify events in terms of time, such as the start and ends of a given period, or in terms of the ball, such as the occurrence of a foul situation during a free kick (Liu, M.-A.

Gómez, *et al.*, 2016; Stein *et al.*, 2017a; McKenna *et al.*, 2018a). Event data is typically compiled by counting the frequency of each event in a particular match or tournament (Stein *et al.*, 2017a). A descriptive summary of events in a football game are listed below (see Table 2-1: Sample of events).

Table 2-1: Sample of events from (Stein et al., 2017c)

Event	Description
Foul Penalty	Free kick on the goal defended only by the goalkeeper
Foul direct free kick	Free kick that is allowed to be directly shot into the goal
Foul indirect free kick	Free kick that is not allowed to be directly shot into the goal
Foul throw in	Throw in that is not correctly executed
Halftime Start	First or second half starts
Offside	Player is in an offside position
Out for goal kick	Ball passes the end line after an opponent touched it
Out for corner	Ball passes the endline after a player from the own team
Out for throw in	Ball passes the sideline of the soccer pitch
Goal	Awarded when the whole of the ball crosses the whole of the
Shot on target	Any shot attempt that would or does not enter the goal if left
Shot not on target	Any shot attempt that would or does enter the goal if left
Pass	Ball touch from one player with direction towards a team
Reception	Ball touch made by the player after receiving it from another
Clearance	Hard ball touch where the player tries get the ball away from
Hold of ball	Play action when the keeper takes the ball with his hands
Running with ball	Used by the player to move the ball around without passing
Cross	Hard ball touch where the executing player is positioned in
Neutral contact	Characterized by ball touch which is difficult to control
Pass assist	The last pass to a teammate in a way that leads to a goal
Cross assist	The last cross to a teammate in a way that leads to a goal
Catch	Keeper catches the ball and hold it in his hands on a
Catch drop	Keeper does not manage to hold the ball but lets it bounce
Drop of ball	Keeper drops the ball after having caught it or holds it in
Punch	Keeper punches the ball with his hand away
Diving save	Keeper jumps to a side to catch the ball
Drop kick	Kicking a ball that is dropping to the ground as it starts to
Yellow card	Displayed by referee to indicate that a player has been
Red card	Displayed by referee to indicate that a player has been

Halftime Ends Substitution First or second half ends Replacing one player with another during a match.

2.4.2.2.2 Context Analytics

As discussed above, in the overview of psychological KPIs, the factors influencing player and team performance are neither fixed nor constant. In addition to being influenced by psychological factors, there are a number of important non-psychological considerations that should also be integrated into analysis. One such factor is the impact of weather, including humidity, temperature and wetness, with some players being more vulnerable to climatic factors than others (Lago, 2009; Lago-Peñas, Lago-Ballesteros and Rey, 2011; Castellano, Casamichana and Lago, 2012; Collet, 2013b). Another important nonpsychological issue for a player is the reaction, behaviour, and performance of their opponents (Tenga *et al.*, 2010a; Sarmento, Marcelino, M. Teresa Anguera, *et al.*, 2014), with logistic regression analysis of tactical KPIs demonstrating that opponent strength is associated positively with passes, but negatively with ball possession (Tenga *et al.*, 2010a; Vilar *et al.*, 2013; Sarmento *et al.*, 2017). Other studies have supported this assertion that opponent strength influences technical skills, with dribbles being affected (Taylor *et al.*, 2008b), and physical indicators, with players tending to run for shorter distances with the ball (Carling and Dupont, 2011b; Orth *et al.*, 2014).

Recent research supports the idea that indicators are context based, meaning that analysts should situate KPIs in specific contexts (Hodges and Franks, 2002; Williams and Hodges, 2005; Koehn and Morris, 2012; Mackenzie and Cushion, 2013; Fernandez-Echeverria *et al.*, 2017). For instance, player-opponent interaction results in players adjusting their performance to reflect the capacity of opponents (Moll, Jordet and Pepping, 2010; Vilar *et al.*, 2013; Sarmento *et al.*, 2017). This observation is not completely new, with older studies suggesting that chance is not an important determinant in match outcome (Reep and Benjamin, 1968, p. 585).

2.4.2.2.3 Predictive analysis

The effective integration of multiple KPIs can produce models capable of predicting match outcomes in specific scenarios (Min *et al.*, 2008; Haghighat, Rastegari and Nourafza, 2013). However, this remains a relatively unexplored area in the literature, although certain recent studies have sought to map out likely probabilities for winning or scoring for specific teams in different match scenarios (Huang and Chang, 2010; De

Baranda and Lopez-Riquelme, 2012; Mahony, Wheeler and Lyons, 2012; Ruiz-Ruiz *et al.*, 2013; Machado, Barreira and Garganta, 2014; Ibáñez *et al.*, 2018). Perhaps the most promising avenue of predictive analysis is sensitivity analysis, which entails the construction of models with parameters that can be altered to measure different outcomes (O'Donoghue and Cullinane, 2011; Collet, 2013b). Predictive systems adopt a dynamic approach to the consideration of contextual factors (e.g. goal scoring during the first 5 minutes and last 5 minutes of the match) (De Oliveira Bueno *et al.*, 2014) and factors during the match (e.g. fatigue level and stress) (Rampinini *et al.*, 2009; Redwood-Brown *et al.*, 2012; Sarmento *et al.*, 2017). The development of such models is potentially challenging and requires in-depth investigation, as well the implementation of accurate KD models by the teams involved, resulting in relatively few papers having been published in this area.

2.5 Theoretical Development: Resource-Based View

The effective creation and application of knowledge requires a number of considerations. In the modern global context, a key issue is the investment and design of technologies that are able to obtain, process and disseminate data (Gullo, 2015). There can be a diverse range of factors to explain why certain organisations use such technologies more effectively than their competitors, enabling them to gain a competitive advantage. In the context of football clubs, it remains unclear how clubs' benefit from these technologies, as these factors influencing Return On Investment (ROI) in KD are multivariate and complex. This issue of ambiguity has also been recognised in IT literature (Chae, Koh and Park, 2018), with the result that there has been extensive investment into IT business value research (Schryen, 2013). These studies suggest that beneficial outcomes are more closely related to implementation of strategies than utilisation of key technologies (Badewi *et al.*, 2018). In other words, the success of IT-enabled business transformation is primarily attributable to issues relating to 'people' (Kotter, 1995).

This kind of question can be answered through the theoretical lens of the resource-based view, which argues that performance variation primarily occurs as a result of resources (Barney, 1991), which can be financial, human, organisational or technological (Melville and Kraemer, 2004). This approach specifically attributes variation in performance to the heterogeneous distribution of available resources between organisations (Peteraf and

Barney, 2003). Resources are technological and organizational complementary resources (Melville and Kraemer, 2004).

There are wide ranges of technological resources and approaches that can assist organisations in augmenting or leveraging data. Examples of these resources include collaboration (Serrador and Pinto, 2015), distributed learning (Lara *et al.*, 2014; Khajah, Lindsey and Mozer, 2016), KD (Mariscal, Marbán and Fernández, 2010), knowledge mapping (García-peñalvo and Conde-gonzález, 2017) or intelligence (Brooks, El-Gayar and Sarnikar, 2015). Collaboration and distributed learning technologies foster more effective and constructive collaboration between individuals in a given organisation (Duffield and Whitty, 2015). KD technologies enable the exploration of internal and external knowledge, whereas organisations can use knowledge mapping technologies to track and catalogue internal data and knowledge application technologies enable existing knowledge to be utilised in more effective ways. Finally. intelligence technologies enable performance tracking and the analysis of competitors (Sun, Sun and Strang, 2018). The current research is focused on KD technology, because of its close relationship with all other knowledge related technologies.

But technological resources alone are not sufficient to realise the promised value. In this context, value is found in the effective integration of technological resources with other complementary resources (Melville and Kraemer, 2004), using both in tandem to effectively and efficiently realise value (Nevo and Wade, 2010). This has been observed in IT-enabled customer service departments, where investment in technology yields measurable benefits to the business when IT resources complement organisational complementary resources (Nevo and Wade, 2011). Research suggests that differentiated performance and competitive advantage are largely attributable to human and organisational complementary resources (Bendoly, Rosenzweig and Stratman, 2007). In a study of 13 companies that had adopted Enterprise Resource Planning Systems, Badewi (2018) found that each technological resource required certain complementary resources for optimal functioning. In the same vein, According to a study by Stratman and Roth (2002), human resources are a differentiating factor in successfully achieving sustainable competitive advantage from Enterprise Resource Planning (ERP) systems.

Human resources are referred to as 'human capital' in sports literature (Felin and Hesterly, 2007) and are acknowledged as crucial to team performance (Dawson and

Dobson, 2002). The quality (i.e. skills, knowledge and abilities) of sportspeople and coaches is a key differentiating factor in team performance (Smart and Wolfe, 2003; Holcomb, Holmes Jr. and Connelly, 2009; Jones, Harris and Miles, 2009). However, irrespective of their value, certain human resources (e.g. knowledge or competencies) are difficult to imitate because they arise from extensive training, experience or specific contexts (Wright, McMahan and McWilliams, 1994). Other organisational resources are also difficult to imitate and cannot be transferred, such as culture (Wright, McMahan and McWilliams, 1994), or communication and collaboration (Kerr and Jackofsky, 1989). Similarly, routines in coordination and collaboration to develop strategies are organisational capabilities difficult to imitate (Grant, 1991; Winter and Szulanski, 2002; Helfat and Peteraf, 2003). Basketball studies have shown that these routines can be a differentiating factor in team performance (Berman, Down and Hill, 2002).

2.5.1 Proposed KD Value Framework in Football Data Analytics

The creation of value through investment in technology can be understood from reactive or proactive perspectives. Reactive theories primarily deal with perceptions of IT and the effect that this has on behaviour (DeLone and McLean, 1992; Petter, DeLone and McLean, 2008; Venkatesh and Bala, 2008; Venkatesh, Thong and Xu, 2012). In contrast, proactive approaches focus on the ways in which technological investment tends to shape the process of value creation (Peppard, Ward and Daniel, 2007; Venkatesh and Bala, 2008; Venkatesh, Thong and Xu, 2012; Ward and Daniel, 2012). Examples of proactive theories include value management and benefits management. A hybrid stream also exists, which seeks to better understand value creation in this context by categorising factors as organisational complementary resources and technological resources (Melville and Kraemer, 2004). In the context of this study, technological resources describes the key features, ancillary systems, technological artefacts, and characteristics of a given technology (Badewi et al., 2018). Effective integration of these technologies requires the presence of the correct organisational complementary resources, which are referred to as KD human resources in the current research. These include attitudes, perception, alignment, culture, norms, competences, skills, and organisational structure. The current research is adopting the framework proposed by Melville (Melville and Kraemer, 2004) to investigate and understand how teams can create value from KD. However, while coherent, the Melville framework can be criticised as being too generic to operate as the sole form of analysis for technological value creation. Therefore, this study will also utilise map framework as an analytic framework to better investigate the process of converting resources into outcomes, leading to changes being implemented to ultimately realise value or benefits.

The diagram below illustrates the discussion that will follow. It begins by underlining the value of the KD approach. As KD is a system, a review will then be provided of the technological platform, its technical requirements, and salient software applications. Finally, the last section discusses key human resources in terms of the various KD stakeholders and the value co-creation process as a whole.



Figure 2-1: Proposed KD Value Framework in Football – © by the researcher

2.5.1.1 KD Technological Resources in football

KD systems require the integration of technology to gather accurate, meaningful data, which can then be analysed and reported, in order to provide usable insights to coaching teams (Maren *et al.*, 2005; Mooney *et al.*, 2011). In order to establish an effective KD system, it is most important for the technological infrastructure to be able to capture and analyse. Event capture technologies seek to identify, measure, track and report changes in the subject matter under analysis (Fuss, Düking and Weizman, 2018). These include wearable technologies that track the movements of players (Li *et al.*, 2016; Mara *et al.*, 2017), those that quantify patterns of motion (Stein *et al.*, 2015), and those that chart internal body behaviour in different scenarios (Moll, Jordet and Pepping, 2010; Humana, 2011; Düking, Holmberg and Sperlich, 2017; Hoch *et al.*, 2017). In contrast, data analysis technologies identify, codify, correlate, and group data in order to process it into new and

usable forms of information, which can then be converted into knowledge (Seshadri *et al.*, 2017).



Figure 2-2: KD Technological Resources – © by the researcher

2.5.1.1.1 Data Sources Technology

In this research, data sources technologies serve to measure, track, and report issues that have the potential to improve team performance. In broad terms, these technologies can be classified into third party datasets (InStat, 2018; OPTA, 2018; STATS, 2018; STATSports, 2018); optical devices (Farin, 2005; Ronkainen and Harland, 2010; Trewin *et al.*, 2017); environmental sensors (Leser, Baca and Ogris, 2011; Rein and Memmert, 2016); and wearable sports devices (Li *et al.*, 2016; Lemire, 2018).

The third-party databases function as the source of the data. This is a comparatively costeffective stage of the process as there are many such databases available on the market, such as InStat Opta, STATSports and Stats (formerly ProZone). The main units of thirdparty databases are generally teams and the players metrics (Tunaru and Viney, 2010; Liu *et al.*, 2013). These details can then be extracted, analysed and evaluated using customised tools or applications to develop understanding about the various outcomes of a game (Cobb, Unnithan and McRobert, 2018). A number of different aspects of a match can be covered, including technical analysis, physical overview, or the ranking and contribution of a given player during a match (Casamichana *et al.*, 2014), as well as analysis of possession or passing at individual or team level (Di Salvo *et al.*, 2007)... More detailed KPI can also be extracted (J. Liu *et al.*, 2013). This could entail analysis on passes, which might produce insights into numbers of passes, ratio of successful and unsuccessful passes, or numbers of passes made backwards or over short or long range (Hughes and Franks, 2005b; Cochrane, 2011; Sampaio *et al.*, 2012; Carling *et al.*, 2014). The comparison of the Opta measurement with an independent measurement in the Spanish league found that the average difference of event time was 0.06 ± 0.04 s for OPTA, with a kappa value of 0.92, indicating a high level of reliability (H. Liu *et al.*, 2013). Other independent studies have demonstrated the reliability of Stats (formerly ProZone) (Rampinini *et al.*, 2007; Di Salvo *et al.*, 2010). Given their acknowledged reliability, this affordable and accessible data has been utilised in the current research (Cobb, Unnithan and McRobert, 2018).

Nevertheless, the technical limitations of these databases makes them insufficient for KD and data mining tools, leading to the recommendation that datasets must be developed " *further to compete with the array of additional parameters offered by new technologies such as global or local positioning system technology*" (Casamichana *et al.*, 2014, p. 701). For this reason, teams must create and maintain their own primary databases to supplement secondary databases. The second technology to collect data is the optical devices.

Optical devices, primarily in the form of automatic video tracking and electronic transmitters (O'Donoghue, 2006; Carling *et al.*, 2008; Camerino *et al.*, 2012), have a well-recognised role in the collection of team performance data (Groom and Cushion, 2004; Mara *et al.*, 2017). These optical devices are fitted with algorithms that are capable of differentiating players in terms of their faces and shirts, enabling speed and type of movement to be measured and categorised (Assfalg *et al.*, 2003b; Farin, 2005; Ding and Fan, 2006; Schlipsing *et al.*, 2017). Automatic video tracking records the movements of each player, whether during training or match conditions (Zhou and Zhang, 2017), with newer optical devices enabling real-time auto-scaling data capture (Ryoo, Kim and Park, 2018). Importantly, optical devices only capture movement indicators, rather than information on internal factors or the external environment.

Environmental sensors measure external factors that can influence individual and team performance, including temperature, altitude, pressure and humidity (Hiscock *et al.*, 2012; Bradley *et al.*, 2014; Fernandez-Navarro *et al.*, 2018b). However, although these factors have been recognised as influential over the success of business operations (Jagadish *et al.*, 2014), the literature has neglected their use in the KD process. As the current research takes a prospective approach to KD, the data collection technologies will focus on those environmental factors that may directly or indirectly affect performance, rather than focusing exclusively on player behaviour.

The final data sources technologies are wearable devices that are attached to the bodies of players to measure their external behaviour (movement) and internal behaviour (circulation) (Chambers et al., 2015). There are number of well-known examples of these technologies. For example, the Geographic Positioning System (GPS) relays data via space satellite (Rangsee, Suebsombat and Boonyanant, 2013) and Local Positioning System (LPS), which allows activities to be tracked via a nearby machine that monitors the wearable device (Frencken, Lemmink and Delleman, 2010; Leser, Baca and Ogris, 2011). Another form of wearable device is the accelerometer, which measure the number of steps taken by an individual, and heart rate monitors, which count and record heartbeats (Casamichana et al., 2014; Li et al., 2016; Bowen et al., 2017). GPS, accelerometer and heart rate monitors are often integrated into a single wearable device called an 'Integrated wearable device' (IWD) (Dellaserra, Gao and Ransdell, 2014; Li et al., 2016), many of which are capable of using a locomotor to measure the frequency of certain categories of movements (e.g. sprinting and jogging) (Dwyer and Gabbett, 2012) or measuring the frequency of contacts (e.g. closeness to other players) (Macutkiewicz and Sunderland, 2011). Finally, textile wearable devices contain integrated sensors to monitor body kinematics and physiological signals (Coyle et al., 2009; Dalsgaard and Sterrett, 2014), enabling coaches to monitor players during training sessions without requiring the use of external devices.

2.5.1.1.2 Data Analysis Technologies

Data analysis technologies generate quantitative and qualitative analyses of individual players and teams, providing rigorous feedback to athletes and coaching teams (Booroff, Nelson and Potrac, 2016). In the context of this research, the chosen data analysis technologies are software-based annotation technologies and data mining technologies. The function of annotation technologies is to enrich collected data, thereby enabling the analyst to process it with the chosen KPIs to reach the desired goals, often including enriched visual data (Cobb, Unnithan and McRobert, 2018). Annotation analysis is the most popular analysis among coaches (Booroff, Nelson and Potrac, 2016), with the majority of professional teams worldwide using this approach in the discovery and measurement of performance knowledge (James, 2006). Annotation is perceived to be relatively straightforward (Booroff, Nelson and Potrac, 2016) and able to cover numerous vital movements, including shots, corners, aerial challenges, crosses, tackles and passes

(Martin *et al.*, 2018; McKenna *et al.*, 2018b). The annotation of movements can grant coaches a deeper insight into their players, enabling them to give better, more targeted feedback (Ives, Straub and Shelley, 2002; Stratton *et al.*, 2004; Groom, Cushion and Nelson, 2011). These insights have also provided additional understanding of the relationship between injury occurrence and in-game behaviour (Carling, Gall and Reilly, 2010); the factors influencing reductions of skill in skilled physical performance (Carling and Dupont, 2011b); and monitoring load in both training and competition (Gaudino *et al.*, 2013).

As a sole performance analysis approach, this approach can become a bottleneck in the KD process (Carling *et al.*, 2013). Prior to the advent of wearable devices, non-annotated actions were not analysed or considered in terms of match strategies (Wright, Carling and Collins, 2014). Classically, notational analysis during games involved manually annotating each event recorded on camera during a match (Martin *et al.*, 2018). This traditional reliance on cameras was high inefficient, leading to the adoption of other qualitative tools, like Quintic, Dartfish, Sportscode and, MatchViewer (Wright *et al.*, 2013; O'Donoghue and Holmes, 2014; Kite and Nevill, 2017; Martin *et al.*, 2017). The objectivity of annotation analysis has been significantly improved through the introduction of specialised software applications, such as SportsCode, TRACAB, Focus X2, ProZone, and Sport Universal Process AMISCO Pro (Carling, Williams and Reilly, 2005; O'Donoghue and Holmes, 2014; Fernandez-Navarro *et al.*, 2016; Martin *et al.*, 2017).

Currently, wearable devices provide a degree of automatic annotation, although reliable capture of specific movements requires a programmer to set an appropriate algorithm. Both research and applied settings commonly utilise this approach to record incidence and outcomes of behaviour, which enables the investigation of certain technical aspects of football performance (Assfalg *et al.*, 2003b; Canales, 2014; Stein *et al.*, 2016; Xue *et al.*, 2017). However, an important criticism of annotation analysis is the inability to precisely track complex biomechanical, physiological, tactical or technical aspects of sport (Sampaio *et al.*, 2015). Annotation analysis is also reductionist, meaning that its analysis ignores important contextual factors (Groom, Cushion and Nelson, 2011; Booroff, Nelson and Potrac, 2016), and reactive, because of the unstructured focus on critical moments in a performance, (Groom, Cushion and Nelson, 2011). Because of this,

irrespective of its advantages, annotation analysis, should be supplemented with other data analytic technologies (Stein *et al.*, 2017a).

The main roles of data mining technologies are grouping and correlating data, or uncovering patterns, influences and effects manifested in a certain data set (Olson, 2018), which enable new insights to be extracted by analysts. Data mining can use a number of different programming and analytical languages, including R, Python, and Java. Tools, packages and visualisation tools typically utilise Python, R, or Structured Query Language (SQL). Meanwhile, the various packages and Application Programming Interfaces (API)s used to collect and process data are generally in Python and R, with extraction and processing normally involving XML data feeds run through analytics platforms like RapidMiner, WEKA, KNIKE RStudio, and SPSS, as well as through visualisation platforms like Tableau, QlikView, Crystal Reports, D3.js, or Alteryx (Thuraisingham, 2014; Slater et al., 2017). Statistical analyses in the literature are typically conducted using software applications and packages, such as SPSS, SAS, RapidMiner, WEKA and RStudio. These analyses can be most efficiently integrated using programming for data and analytics, such as R, Python, or MATLAB (Slater et al., 2017; Olson, 2018). Given the array of available software and options, one of the main challenge for data analysis technologies, which includes both data mining and annotation analysis, is the cost involved in the purchase, deployment and utilisation of such technologies (Martin et al., 2018)

2.5.1.2 KD Human Resources in football

KD is a socio-technical process (Ho, 2017), arising from the interaction between individuals and technology. Because of this, the successful functioning of KD requires the people involved to have the correct skills and competences, otherwise the technological outcomes will not produce the expected benefits (Groom and Cushion, 2004; Carling, Williams and Reilly, 2005). This requirement extends to all individuals involved in the analysis process, such as coaches and team managers, rather than just the analyst (Medeiros, 2017; McKenna *et al.*, 2018a). The following section will present and examine the roles involved in the KD process, then outline the way that the interaction between these roles in the agile value co-creation process results in the discovery of new knowledge.

2.5.1.2.1 Football Team Management Role

Proper teamwork requires clarity of roles (Bray and Brawley, 2002), otherwise effective teamwork will be compromised and group satisfaction is likely to fall (Eys *et al.*, 2005; Fletcher and Arnold, 2011). Clarity of roles does not mean that each role is isolated, however. Instead, it means that all members of the team should be aware of the responsibilities and requirements of the others. Analysts enquired about preferred playing style and key priorities, but have knowledge about concepts like goals, chances and set plays. This perhaps help communication and brainstorming among the analysts and the coaching team (Lyle, 2003; Nash and Collins, 2006). In the context of the management of a football team, the key roles are team manager, coach, and data analyst, each of which can be a single individual or a group, depending on the particular needs of the club (Zambon Ferraresi, Lera López and García Cebrián, 2017).

2.5.1.2.2 Team Manager Objectives and Competencies Role

The primary responsibility of a team manager is to oversee the financial performance of the team (Watts and Wruck, 1988; C. P. Barros and Leach, 2006; Kelly, 2008; Hamil and Walters, 2010; Zambom-Ferraresi, Lera-López and Iráizoz, 2017; Kilpatrick, 2018); or to maximise profit (El-Hodiri and Quirk, 1971; Szymanski and Smith, 1997; Rohde and Breuer, 2017). Extensive research has investigated the topic of maximising the financial returns of football clubs (Szymanski and Smith, 1997; Garcia-del-Barrio and Szymanski, 2009; Hamil and Walters, 2010; Kuper and Szymanski, 2018), although recent research has tended to emphasise sustainability over profit. There is an established assumption that sports managers maximise utility to achieve non-profit goals, consuming resources in pursuit of satisfaction rather than profit (Sloane, 1971). In this sense, it is important to distinguish between performance success, which refers to the ability of a coach to win matches, and profit success, which describes the ability of a team manager to ensure financial sustainability and profitability (Carlos Pestana Barros and Leach, 2006).

(Plumley, Wilson and Ramchandani, 2017) state that there are three primary cases that might arise in terms of the relationship between financial success and winning match success. In the first scenario, greater profits might improve team performance, or more successful teams might become more profitable. In this case, no conflict exists between satisfying the desire of shareholders to obtain profit and the desire of fans to win. This relationship is typical of the situation in the stock market model of ownership of privately

owned clubs (Wilson, Plumley and Ramchandani, 2013). In the second case, successful performance does not necessary result in increased profitability. A correlational study into the relationship between profit and the league positions of 40 football clubs (1978-1997) found no significant evidence that changes in league position brought a corresponding change in profit (Szymanski and Kuypers, 2000, p. 22). Finally, in the third scenario, successfully winning matches might cause or be attributed to lower profits, such as when players are overpriced or there is poor investment at the club. This relationship would require shareholders to decide upon an acceptable balance between profit and performance. In all cases, there is a fundamental link between sporting and financial activities and this association is essential for the operation and sustainability of football clubs (Szymanski and Kuypers, 2000, p. 22).

In summary, the role of the team manager is to balance performance and profit, although with an emphasis on profit in order to satisfy stakeholders. There is generally a more positive relationship between financial performance and team performance among teams listed on the stock market than teams that are either privately or publicly owned. Nevertheless, poor investment can harm teams, even when they are successful. However, examination of the team manager is beyond the scope of this study, because concentrating on financial aspects could adversely affect the quality of this research.

2.5.1.2.3 Coach Objectives and Competences

The sole objective of the team coach is to help the team to win matches (Cushion, 2001; Ingle, 2013; Vilar *et al.*, 2013; O'Donoghue and Robinson, 2016). As a consequence of this, coaches invest heavily in analysis of performance and associated problems using a range of data, reports and videos (Wright, Atkins and Jones, 2012). Some coaches utilise sophisticated, objective data, while others rely more heavily on personal intuition and judgment (Adjei *et al.*, 2013; Medeiros, 2017). Many models exist to conceptualise and investigate the myriad competences required by team coaches. The first is Santos *et al.*'s (2010) theory, which focuses on a number of competences: annual and multi-annual planning; practice and competition orientation; and personal and coaching education. Additionally, coaches who are cooperative and work effectively with data analysts invariably use data more effectively to develop advanced plans and training scenarios (Martin *et al.*, 2018).

In many clubs, data analysts, and particularly performance analysts, play an instrumental role in the coaching process (Groom, Cushion and Nelson, 2011; Nelson, Cushion and Potrac, 2013). Coaching involves guiding players, but also leading and building trust between the coach and his players, which benefits from transparent, objective performance analysis technologies (Nelson, Potrac and Groom, 2014). Coaches also sometimes act politically with the motivations and intentions of the team (Booroff, Nelson and Potrac, 2016). In this scenario, analysis can become a tool to assert power, which can harm the perceptions, relationships and culture of the club (Booroff, Nelson and Potrac, 2016).

2.5.1.2.4 Data Analyst Competences

Performance analysts work within the football coaching team to ensure that all data is utilised efficiently and effectively (Davenport and Harris, 2007). Data analysts play a critical role in team success, with research demonstrating that coaches supported by data analysts receive more information and are therefore better able to understand match dynamics than when coaches attempt to fill analytical roles themselves (Martin *et al.*, 2018). In some teams, the data analyst role involves teams of specialists, working in recruitment analysis, opposition analysis, and even academy analysis (Hatton, 2013; Wright *et al.*, 2013, 2016). However, despite this growing prevalence, the role of data analyst is relatively recent within team management as a whole. For example, according to a recent survey on Australian football, only 13% of the coaches have access to data analysts, which limits their advanced planning potential (Martin *et al.*, 2018).

It is important to note that the role of data analyst is different from that of sport scientist. A performance analyst at a large club, such as in the English Premier League, will focus on examining match strategies and team performance, whereas the sport scientist will be responsible for the physical aspects of performance (Garganta, 2009a; Wright *et al.*, 2016; Sarmento *et al.*, 2017). Data analysts must demonstrate statistical and technological competences, enabling them to effectively utilise sophisticated data analytics technologies (Wright, Carling and Collins, 2014; McKenna *et al.*, 2018a).

2.5.2 Modelling Development Approached: Agile Value Co-creation Approach

In modern football teams, data analysts produce data and the coach or team manager consumes that data. This relationship benefits from close cooperation and interaction, in order to ensure that the analyst properly understands the needs of the consumers and is therefore able to translate them into statistical models for the testing or exploration of certain aspects or issues. As summarised in Table 2-2, there are two main model development approaches: Waterfall (also called Push), which is a relatively simple process that fits clear outputs, and Agile, which is more suitable for outputs that are complicated and difficult to express.

Knowledge creation is fuzzy and challenging to articulate clearly in advance (Du Chatenier *et al.*, 2009). Therefore, development of tailored, needs-based solutions requires collaboration between the model producer and model consumer. In other words, the discovery of new useful knowledge requires constructive positive cognitive sharing and interaction among different stakeholders (e.g. data analysts, team managers, coaches, physiotherapists) (Jones, Harris and Miles, 2009; Martindale *et al.*, 2010). This collaboration is known as the value-co-creation process (Prahalad and Ramaswamy, 2004). The following sections outline the model development approach to illustrate the use and validity of the value co-creation approach. This is followed by an in-depth discussion of agile value-creation models and the various tools involved in agile models. *Table 2-2: Push versus value co-creation production created by the researcher*

	Push Production	Value Co-creation Production
Waterfall	Traditional project management approach	Client is involved in designing the solution but not involved in producing the solution
Agile	Client is involved in the production, but the scope is ambiguous, but the client is not involved in the production	Client is involved in designing the solution and involved in producing it. Knowledge Discovery Agile value co- creation approach

2.5.2.1 Knowledge Development Approaches: Waterfall versus Agile

The traditional waterfall approach is most appropriate and efficient when working with a clearly defined and deliverable scope in a stable environment, often over repeated projects (APM, 2015). In contrast, the agile approach is more suitable with vague scopes or requirements, which need flexibility and regular stakeholder feedback. Agile methodologies benefit from clear, scheduled tasks, carried out quickly, flexibly, in phases, and with adaptation following a clear well-understood plan. As an example of this, the digital services in the United Kingdom government utilise an agile approach (GOV.UK, 2018).

Development of tailored, needs-based solutions requires collaboration between the model producer and mod el consumer. In other words, the discovery of new useful knowledge requires constructive positive cognitive sharing and interaction among different stakeholders (e.g. data analysts, team or measure key metrics (Carling, Williams, et al. 2005; Hughes 2004; O'Donoghue 2006). The use of waterfall depicts a simple, linear process in which the consumer clearly articulates their requirements and the producers fulfil those needs. However, given the complexity and iterative requirements of coaching (Lyle, 2003), agile approaches may be more useful. This approach encourages close collaboration between client and producer, which is especially useful when the needs are not clearly identified and the producers therefore do not have clear guidelines. For instance, telling stories as an agile based tool can improve collaboration in football teams at the management level (Perin, Vuillemot and Fekete, 2013). This perspective argues that the job of the data analyst is therefore to listen to the coach about the stories, to tell stories and to express findings, rather than to provide an exhaustive list of statistics about games or players. This is reflected in the review of literature, which did not find explicit research to demonstrate how agile development can develop new models in the knowledge discovery approach.

2.5.2.2 Agile Based Approach

Modern dynamic digital services utilise iterative approaches to manage complex challenges, such as those discussed above. Agile offers a way to overcome the rigidity of the traditional project management approach, which suffers from "too much planning" (Boehm, 1996) and is challenged by the inability of specifications to differentiate and clarify deliverables from prototype. In addition, early specification of requirements can lead to extraneous features being added, the lack of opportunities to alter or develop functionality, and the immutability of focus, which is especially problematic given the likelihood that requirements or environments will change. In a study of 1386 projects, the relationship between planning and delivering promises was found to be an inverted U (Serrador and Rodney Turner, 2014).

Agile approaches share certain values and principles, as underlined in the Agile Manifesto (*http://agilemanifesto.org/*) (Collier and Highsmith, 2010; Sutherland and Schwaber, 2017), emphasising flexibility, leanness and engaging the clients (Conboy, 2009). Traditional project management methodologies experience difficulties in dynamic

environments (Collyer *et al.*, 2010), leading to increased reliance on Agile, which utilises minimal documentation to make the approach more flexible, while also increase responsiveness to changing conditions (Serrador and Pinto, 2015). Agile methodologies encourage iterative user engagement throughout the developmental cycle (Serrador and Pinto, 2015), with daily meetings to ensure that clients are updated, involved, and satisfied throughout the process (Mann and Maurer, 2005).

Historically, these kinds of methodologies when software developers sought better practices and key values with which to manage the development process (Beck *et al.*, 2001).

Establishing solid theoretical foundations for the definition of agile methodology in the specific context of the current research requires a review and synthesis of the three most relevant definitions in the literature. The first definition is a software development practice that seeks to manage issues relating to high levels of uncertainty, dynamic and frequent changes, short development cycle, digital deliverable and custom based systems (Abrahamsson *et al.*, 2002; Dingsøyr *et al.*, 2012; Hobbs and Petit, 2017). The second definition focuses on the 'evolutionary' aspect of the development process:

"Agile software development is an evolutionary (iterative and incremental) approach which regularly produces high quality software in a cost effective and timely manner via a value driven lifecycle" (Ambler, 2009, p. 6).

According to this definition, agile methodology is disciplined and self-organising, with a high degree of collaborative from active the active involvement of stakeholders to ensure that their diverse, evolving needs are met. The final definition transcends the narrow understanding in software engineering to encapsulate the diverse needs of the market and technology. In this sense, Agile refers to the ability of a team to respond quickly to the technological or market needs of customers or stakeholders, enabling "better project and product performance in an innovative and dynamic project environment" (Conforto *et al.*, 2016, p. 547).

As the concept of required knowledge is fuzzy in terms of usability, usefulness, and fitness, different definitions are required for the projects used to uncover or develop knowledge. Building on previous definitions, this methodology is a co-created iterative approach to the development of customised solutions to fuzzy problems. This research is informed by the Agile approach to the discovery of knowledge as a deliverable, which is

produced by data analyst in conjunction with continuous interaction with the coaching team. The Agile approach can utilise a number of different tools and techniques: user stories, on demand scheduling (Kanban), sprint, retrospective, and burn down charts. These will be discussed in more detail in the following sections.

2.5.2.2.1 User Stories

'User stories' refers to a technique for modelling and understanding the multitudinous requirements of users in an agile project. These stories serve as building blocks to conceptualise and inform projects (Inayat *et al.*, 2015), utilising the stories to divide up the work to be completed by the team, working in consultation with customers or owners of a product (Agile Alliance, 2018a). User stories were first introduced as a part of the traditional XP practices (Beck, 1999) and later as a core component of the XP2 evolutionary practices, reflecting the importance of obtaining and understanding user requirements to make meaningful improvements to software or application quality (Beck and Andres, 2004).

A user story is a way to capture a technology application or software functionality. Each story should address: who (the type user or consumer of the piece of technology), what (the need to develop or use the technology), and why (the benefits and outcomes) (Patton, 2014). User stories are scheduled and extracted from backlogs that sort and rank according to their particular importance, relevance, category or scope within the cycle or iteration (Dimitrijević, Jovanovic and Devedžić, 2015).

2.5.2.2.2 Story Mapping

User Story Mapping is a usage-centric approach in which agile teams visualise the stories that they are working on (Rubin, 2012), using terms like 'activity' or 'epic' to reflect on the primary categories of the task, and 'task', 'theme' or 'subtask' to reflect on less important categories (Rubin, 2012; Patton, 2014). Using a Story Mapping approaches encourage user-centred design and decomposition of story requirements in the workflow context (Rubin, 2012; Taibi *et al.*, 2017). The effect of this is that agile teams can obtain a better overview and understanding of the stories at a low level, as well as a high level understanding of the entire project to ensure commonality of vision (Taibi *et al.*, 2017).

2.5.2.2.3 Agile Release Planning

The purpose of the release planning technique is organising, categorising, and prioritising user stories during the mapping process. Effectively, release planning is a roadmap of the

release plans for a particular project (Dimitrijević, Jovanovic and Devedžić, 2015). The release planning technique groups related user stories in terms of the area of functionality within the final application, product or service (Heldman, 2011). In Scrum (an Agile approach), this approach is referred as Scrum Resales Planning, which focuses on what will be delivered and how that aim will be achieved (Sutherland and Schwaber, 2017). All members of the Agile team should be involved in these types of planning activities, thereby ensuring consistency of approach to user stories (Alliance, 2018) and agreement on deadlines and the number of releases required (Cohn, 2004).

2.5.2.2.4 On-Demand Task – a Kanban Technique

Kanban is the method that Toyota devised to implement an Agile methodology (Ahmad, Markkula and Oivo, 2013; Saltz, Shamshurin and Crowston, 2017; Agile Alliance, 2018b). In this approach, tasks are scheduled on-demand or as a Work In Progress (WIP) (Ahmad, Markkula and Oivo, 2013). Kanban focuses on visualising the WIP and associated work iterations as fully transparent (Matharu *et al.*, 2015). In contrast to Scrum, Kanban allows iterations to be changed (Wang, Conboy and Cawley, 2012), with WIP enabling stories to be delivered in other ways, as each piece of work is updated as it is completed.

2.5.2.2.5 Sprint

Sprints is the most common short Agile technique (Baird and Riggins, 2012; Hobbs and Petit, 2017). It is mostly used in Scrum because iterations are time limited, with work being forbidden on additional User Stories or changes (Rubin, 2012). Every Sprint includes specific Sprint Backlog items that are prioritised based on the decision of the Scrum Product Owner and the Scrum Team decided which work to carry on and for how long (Adjei *et al.*, 2013; Sutherland and Schwaber, 2017).

2.5.2.2.6 Retrospective

The retrospective technique enabled teamwork to be improved by reviewing the feedback, advantages and disadvantages after every Sprint (Stellman and Greene, 2014). The process of sharing feedback with the team is a crucial final step of the Scrum or XP. This enables plans and improvements to be carried forward to the next spring, improving the output and performance, as well as increasing trust within the team. The development of intra-team trust is the main goal of this stage in the iteration (McHugh, Conboy and Lang, 2012). However, because retrospective enables the verification and validation of team

progress, if all members are not present at these meetings then the team may be adversely affected (Drury, Conboy and Power, 2012; McHugh, Conboy and Lang, 2014).

2.5.3 Agile Value Co-creation

Value co-creation is a client-centred strategy (Nambisan and Baron, 2009) that originated in marketing literature as a way to conceptualise the design of services through and based on the client (Prahalad and Ramaswamy, 2004; Auh et al., 2007). This concept was later utilised in the development of service-oriented architecture in IT research (Ordanini and Pasini, 2008). In terms of football literature, value co-creation has been employed in the design of strategic governance models to oversee the integration of local trusts and football clubs (Castro-Martinez and Jackson, 2015). However, value co-creation has not been specifically defined in terms of sport management. Therefore, a definition has been adapted from other disciplines to match the current context. Accordingly, this research defines value co-creation as the process of collaboration among the members of the coaching team (team manager, coaches, data analyst and sports scientist) to create new analytic models and use new technologies to discover knowledge in the datasets. This strategy seeks to blend different perspectives to obtain a mutually valued outcome (Prahalad and Ramaswamy, 2004). In this vision, all participants (i.e. data analyst, coaches, team manager, and other stakeholders) have their own views, needs, and interpretations of the data. For instance, while a coach manager might have a preference for visualised techniques, such as video analysis, the data analyst is likely to be more quantitative oriented. Combining these views may produce deeper, more insightful analysis of data through the development of new models (Mackenzie and Cushion, 2013). Presentation and interpretations may also be affected by social environment, the philosophy of individuals or their roles, and the particular qualities of recipients (Groom, Cushion and Nelson, 2011; Barbour, Treem and Kolar, 2017). By extension, any proposed intervention of video analysis from an analysts would entail careful consideration of the format, design of session and delivery, and specific targeted outcome (Groom, Cushion and Nelson, 2011). Hence, communication during the value co-creation process is vital to ensure understanding of the reports and associated technology outputs (Booroff, Nelson and Potrac, 2016), as misunderstandings can result in incorrect or unwanted consequences, lack of commitment, and limited or incorrect understanding of the various metrics (Wright, Atkins and Jones, 2012).

Much of the literature stresses the importance of trust and respect in the collaboration between the members of the team, including manager, coach, and data analyst, as well as including athletes, assistant coaches, and administrators (Potrac, Jones and Armour, 2002; Jones, Armour and Potrac, 2004; Cushion, Armour and Jones, 2006; Potrac and Jones, 2009; McGarry, T., O'Donoghue, P., & Sampaio, 2013, pp. 175–186). Trust can improve understanding and reduce conflict between stakeholders (Groom, 2012; Groom, Cushion and Nelson, 2012), ultimately improving the value co-creation process.

In summary, stakeholders collaborate to produce a fuzzy output (knowledge), based on fuzzy requirements (knowledge requirements for taking a proper decision), creating the need to combine concepts (the agile approach and the value co-creation process) to create the Agile value co-creation process. This new process facilitates understanding of the creation and discovery of knowledge, ultimately improving the co-creation development process.

2.6 KD Maturity Models (MMs)

Maturity Models (MMs) are developmental models that describe the ability of an organisation or system to produce a specific outcome or logical relationships between the development routes of certain dimensions, attributes, or processes (Kuznets, 1966). Early forms of maturity are evident in the work by Maslow (1970), Kuznets (1966), and Nolan (1973; 1979), who created a hierarchy of human needs, economic growth, and IT progression in organisations, respectively. Maturity models assume that organisational development is predictable and structured. For this reason these models are extensively used in Information Systems (IS) literature, especially in data analytics (Chen and Nath, 2018). The current research is primarily focused on these models in the organisational context.

MMs help organisations to meet performance objectives by providing valuable insights into current levels of capability, process, and resource development (Cosic, Shanks and Maynard, 2012), as these theories offer a systematic overview of organisational capacities along a particular projected path (van de Ven and Poole, 1995; Gottschalk, 2009). In other words, MMs are "*a set of characteristics, attributes, indicators, or patterns that represent progression and achievement in a particular domain or discipline*" (Caralli, Knight and Montgomery, 2012, p. 3). In IS literature, MMs demonstrate the process through which organisations attain certain

levels of technical competency and strategic alignment (Becker, Knackstedt and Pöppelbuß, 2009a).

MMs are utilised in the current research. In order to provide a coherent understanding of the rationale for this and a theoretical basis for the use of this model, the following section discusses the purpose for the development of maturity models. This is followed by an overview of the structure of MMs, including key mechanisms, components, and the main aspects in the design, after which the section presents the presenting different approaches and types of MM. This discussion concludes with a critical examination of the literature on MMs in the field of data analytics, which is believed to be the closest form of MM that can utilised within the context of KD in football.

2.6.1 Maturity Model Purposes

MMs have three possible purposes of use: descriptive, prescriptive, and comparative (De Bruin *et al.*, 2009). In other words, MMs are used to describe current situations, predicting possible futures based on identified areas of weaknesses, or comparing different cases based on a particular set of measures. Each purposes of use is described in more detail below.

A descriptive purpose for MMs uses diagnostic tools to investigate certain attributes and practices in order to identify areas for development and provide an informative assessment to stockholders (Becker, Knackstedt and Pöppelbuß, 2009b; Maier, Moultrie and Clarkson, 2009). This enables MMs to evaluate or control progress and inform developmental initiatives (Iversen, Nielsen and Norbjerg, 1999), utilising sets of characteristics, goals, indicators, practices and processes to better understand and assess the patterns of work in an of the organisation (Rosemann and Bruin, 2005; De Bruin *et al.*, 2009). In contrast, a prescriptive purpose evaluates the current maturity of an organisation to create viable strategies and paths towards desired levels of development (Becker, Knackstedt and Pöppelbuß, 2009b). Finally, a comparative purpose compares data from a current MM with historical data in order to benchmark internal and external performance and progress (De Bruin *et al.*, 2009; Maier, Moultrie and Clarkson, 2009).

2.6.2 Maturity Models Structure

In broad terms, MMs are measures that create a sequenced progression path from one state to a desired mature state (Becker, Knackstedt and Pöppelbuß, 2009a; Gottschalk, 2009). This involves two main components: Maturity Levels and Measured Components.

Maturity Levels, or stages, represent different levels of progress or improvement over time. These levels are characterised in terms of stages-of-growth models, stage models, or stage theories (Prananto, McKay and Marshall, 2003) and must be aligned with the culture, strategies and structure of the organisation.

Measured Components are the processes that deliver the needs of the proposed Maturity Levels (Pöppelbuß and Röglinger, 2011). These components are often organised hierarchically into multiple layers (Rosemann and Bruin, 2005) and include competence objects, maturity levels, criteria, and methods for data collection and analysis (M.C. Paulk *et al.*, 1993). Fraser *et al.* (2003) state that these components usually consider the following, as part of the activities undertaken at certain levels of maturity: levels, descriptors, descriptions for each level, capability areas (dimensions), activities for each capability area, and a description of each activity. Finally, general frameworks exist to inform the structure and function of MMs, creating a practical approach for the design and use of these models (Pöppelbuß and Röglinger, 2011).

2.6.3 MM Approaches

There are three main approaches taken by MMs: progression, capability, and hybrid (Mehravari, 2014). A progression model is a lifecycle approach that focuses on improving organisational maturity, with no relation to process or capability. A capability model focuses on improving performance in terms of process and goals from one maturity level to the next through the assessment of processes and capabilities (Wendler, 2012). Finally, a hybrid model mixes both models, with maturity being constructed to reflect on development in terms of improvements in performance (Eppler, 2014).

Maturity models can be either focus on a single process or the entire set of available processes (Proença and Borbinha, 2016). In the former scenario, the MM investigates the capability to complete a specific task, such as obtaining or designing a technology, whereas the latter approach examines the ability of the entire organisation to complete tasks efficiently (Rosemann and Bruin, 2005; Hammer, 2007).

2.6.4 Capability Maturity Model (CMM)

The Capability Maturity Model (CMM) was one of the most common MMs in literature and in professional contexts. This framework was designed to assess and develop software processes and performance in the software industry (Caralli, Knight and Montgomery, 2012). CMM examined three components: Key Process Areas (KPAs), Process Capability and Maturity Levels (Mark C. Paulk *et al.*, 1993). The CMM was adapted by companies and universities, and was extended to better focus on certain organisational requirements (Dawson, 2009). The CMM was replaced by the Compatibility Maturity Models Integration (CMMI) framework in 2010 (CMMI, 2010; CMMI Product Team, 2010, 2018).

The Capability Maturity Model Integration (CMMI) was created in 2000, with the creation of a single framework from three integrated MMs. This framework, the Capability Maturity Model for Software, System Engineering Maturity Model and the Integrated Product Development Capability Maturity Model, has subsequently informed the development of other MMs, such as the People Capability Maturity Model (P-CMM). Versions of the CMMI framework exist for acquisition, development, people and services, and a new version (V2.0) of the CMMI Development Framework was recently released(CMMI Institute, 2018).

The adoption of the CMMI framework results in improvements in key processes areas, increased organisational capability and higher levels of maturity (Software Engineering Institute, 2005, 2010; CMMI, 2010; CMMI Institute, 2018). CMMI assesses organisational change in terms of the match between business strategies and capability levels, which are specific developmental goals, or maturity levels. The function of CMMI versions is to provide information for adaptation or modifications based on a specific score to meet the organisational or team needs in the fulfilment of specific sets of strategies.

2.6.5 Data Analytics Maturity Models

MMs for data analytics systems measure the ability of organisations to integrate and utilise knowledge in organisational decision-making (Popovič, Coelho and Jaklič, 2009; Popovič *et al.*, 2012; Chen and Nath, 2018). For instance, Chen and Nath (2018) investigated management perceptions and attitudes. They discovered that the maturity of business analysis influences both process performance and organisation performance. As will be discussed in detail below, Business Analytics Maturity Models are generally classified into technical focused, organisation focused or mixed.

2.6.5.1 Technical Focused Maturity Models

Technical MMs focus on technical aspects of IT infrastructure in terms of connectivity, data warehousing, integrability and security, with earlier versions focuses on those

aspects most relevant to enterprise data architecture and Business Intelligence (BI) applications. Examples of technical MMs include Watson *et al.*'s (2001) Data Warehousing Maturity Model, which uses levels of growth to classify technical capabilities into three stages (initiation, growth and maturity), which are defined by nine characteristics of enterprise data warehouses. Another example is Eckerson's (2004, 2009) six-stage MM for data technology implementation, which was inspired by human growth. The final example is Tan *et al.*'s (2011) Enterprise BI Maturity Model, which measures four dimensions (analytics capabilities, data warehousing architecture, information quality, and master data management) to determine five maturity levels.

These technology focused maturity models are criticised for a failure to recognise the relationship between BA technology and the culture, powers, processes, strategy and structure of organisations (Lahrmann *et al.*, 2011).

2.6.5.2 Organisation Focused Maturity Models

Organisation MMs are the other category of BA, with an exclusively non-technical focus on people aspects of a company. The first example of this kind of MM is Gartner's Maturity Model for Business Intelligence and Performance Management, which uses a five-point scale to evaluate dimensions including availability of performance metrics, levels of business sponsorship, organisational structure support, and scope of the BA initiative (Rayner and Schlegel, 2008). The second example is Capgemini's (Capgemini, 2012) Predictive Analytics Maturity Framework Assessment which seeks to maximise business benefits by the evaluation and optimisation of the maturity of the predictive analytics environment of an organisation in terms of its competence, deployment, enablers, governance, strategy and vision. The third example is Davenport and Harris's (Davenport and Harris, 2007) DELTA model (Data, Enterprise orientation, analytics Leadership, strategic Targets and Analysts) which focuses on a number of analytical factors in an organisation: "Analytically Impaired, Localized Analytics, Analytical Aspirations, Analytical Companies and Analytical Competitors". Another organisation MM was developed by Harriott (2013) to assess the effectiveness of analytics strategy and then develop structured development paths to optimise value output from investment in business analytics. Harriott's MM uses seven dimensions, which are "business challenges, data foundation, analytics implementation, insight, execution and measurement, distributed knowledge, and innovation". The final example is The Data Warehousing Institute (TDWI) model, which looks at readiness for analytics, culture, data management, internal politics, and skill sets, in order to categorise an organisation in terms of five levels of analytics maturity. This enables identification of business drivers and cultivates a positive BA environment, in which analytics are properly integrated and strategically aligned (Halper and Stodder, 2014).

These kinds of BA MM are criticised for ignoring technological aspects of business. This led to the development of the third stream of BA MM, which focuses on the integration and alignment of technical and organisational dimensions.

2.6.5.3 Integrative Maturity Models

One of the most notable capability-focused MMs is the BA Capability Maturity Model (BACMM) (Cosic, Shanks and Maynard, 2012). The BACMM focuses on assessing capability in governance, culture, technology and people, evaluating a range of technical, organisational and strategic issues on a five-level scale of maturity from 'non-existent' to 'optimized'. This comprehensive approach has resulted in the BACMM being criticised for its generality, as the model is broadly applicable to most IS contexts and phenomena. There is also little evidence of empirical testing to validate the claim that the BACMM leads to a sustainable competitive advantage.

In general, impact-focused BA MMs focus on organisational performance and decision enablement. As an example of this, Teradata's BI and DW Maturity Model examines the impact of analytics on certain business processes and decision capabilities in terms of five maturity levels: reporting, which seeks to categorise events; analysing, which looks at the reason for events; predicting, which tries to determine likely future events; operationalising, which looks at current events; and activating, which describes the push to make events occur (Olszak, 2016).

The IDC's Big Data and Analytics Maturity scope examines organisational development towards the effective and efficient use of data in decision making (Vesset *et al.*, 2013). This model examines quality of data, strategic clarity, people (skills and culture), process improvement, and sophisticated of technology. These are graded in terms of five stages, from 'Ad Hoc' to 'Optimized', to predict maturation of organisational capabilities. IDC states that technological and data capacity develops faster than strategic intent and people. Similarly, the Business Intelligence Maturity Hierarchy (BIMH) illustrates the knowledge management capability of organisations in terms of data, information, knowledge and wisdom (Rajterič, 2010). Finally, Lahrmann *et al.* (2011) studied an impact-focused maturity model of BA with 103 business and IT practitioners. Their study validates the theory that a MM based on deployment of BA technology lead to greater internal process efficiency and overall performance in the adopting organisation. The assessment of BA maturity should involve the measurement of those areas most affected by analytics(De Bruin *et al.*, 2009; Souza and Gomes, 2015). However, the review of MMs suggests that many models focus on certain aspects of BA maturity, rather than comprehensively examining all related aspects and considerations.

2.6.6 Criteria for a "Good" Maturity Model

MMs face certain inherent challenges which should be overcome during the development of a model. Ahlemann et al. (2005) state that MMs should have a solid empirical base, software tool support, standardization, flexibility/adaptability, benchmarking applicability, certification, disclosure of potential for improvement, evidence of correlation between maturity model adoption and performance. Good capability assessment models should be cost-effective, reliable and valid, as well as using good academic evidence (Simonsson, Johnson and Wijkström, 2007), however many MMs are overgeneralised and do not have a sufficiently rigorous empirical base (Benbasat et al., 1984; King and Kraemer, 1984a; Rosemann and Bruin, 2005; McCormack et al., 2009). Additionally, MMs are not applicable for all industries or sectors (Mettler and Rohner, 2009). Even when they are applicable, the design must clarify the need for continuous improvement before prescribing a set sequence towards a specific "end state" (King and Kraemer, 1984b) leading some authors to design MMs as lifecycles or continuous processes (Becker, Knackstedt and Pöppelbuß, 2009b; De Bruin et al., 2009; Maier, Moultrie and Clarkson, 2009; Mettler, Rohner and Winter, 2010; Solli-Sæther and Gottschalk, 2010; van Steenbergen et al., 2010).

2.7 Synthesis of the Literature Review

The literature review for this research covers the following areas to aid in developing and understanding the width and depth of the research. Additionally, to aid in improving the current Football Data Analytics practices within the coaching team. Table 2-3 highlight the major research subjects that are investigated in this research to develop the models, framework and the maturity model.

Subjects	Related Paper Subjects	Sections	
Knowledge	To understand the definition of knowledge and the key differences		
	between Data, Information, Knowledge and wisdom.		
Knowledge	To look at the definitions and the descriptions of KD in the	2.3	
Discovery	computing, informatics, information technology, information systems		
	This is to review the related applications on KD in the domain of		
	football. Also, the objectives of using KD, Data Mining, Data		
KD in the	Analysis and related subject in football data analytics. Also, to		
KD in the Football Industry	reflects on the current needs of developing performance analysis	vsis	
	applications in football based on the variations of the KPIs (i.e.		
	psychological, physical, technical and tactical). Finally, to look at the	, to look at the the analytical	
	subjects of FDA used in football in order to understand the analytical		
	model used in this research area.		
Knowledge	Most of the studies focus on the what would football data analysis		
Discovery	would benefit researches with minimal focusing in what would be the	2.5	
Value	value of FDA. The review in this subject addresses these concerns.	2.5	
	That aided in the development of the initial framework.		
The initial	The scope of the area of the literature it to build the framework, the		
KDV	models and sub-models needed to address the needed of the research.		
Framework in	The subjects covered here are the Technological Resources in		
Football Data	football (i.e. data sources and data analysis technologies), The KD		
Analytics	Human Resources (i.e. Team management roles, competences for the		
	different roles (i.e. team managers, coaches, data analysts, and the	26	
	members of the coaching teams). Additionally, it covers the Agile	2.0	
	approaches, tools and artefacts that are best suitable to use in		
	addressing the research needs. Finally, it reviews the principles of		
	developing Maturity Models (MM) (i.e. purpose, structure,		
	approaches, the variant application of MM and the criteria of		
	developing good MMs)		

Table 2-3: Research Subjects used in this research

2.8 Summary

This chapter sought to review and synthesise the literature relevant to the knowledge discovery process and associated resources, with particular reference to football. Based on this review, the Melville framework has been adopted to structure our understanding. This paradigm argues that technological and complementary resources are important for the delivery of outcomes, with consideration of human resources, technological resources and value. A definition was provided of the concept of value in the context of KD to develop a new functional definition for the sports industry. As the main mechanism for the realisation of value through the KD system is performance analysis, a review was then conducted into analytic methods and KPIs. The second section models for use by football teams, the literature in this area was outlined, with particular reference to MMs and data analytics. The next chapter will introduce the methodology utilised as this research aims to operationalise the results into a maturity model for use by football teams, the literature in this area was outlined reference to MMs and data analytics. The next chapter will introduce the methodology utilised to extend, improve and validate the Melville framework, as outlined in the literature, to fit the needs of the football industry.

Chapter 3 Research Methodology

3.1 Introduction

This chapter aims to provide a detail examination and discussion of the research philosophical stances to answer this research question and fulfil the objectives of the current study. Based on the question and the philosophical stances embraced, the research strategy is developed to inform the adoption of the research methods and analytical approach. This chapter begins with a discussion of the research philosophy, following this with the resultant research strategy, presented in terms of research methods. This is followed by the research analytic approach proposed to provide insightful knowledge from the research activities. Finally, the chapter concludes by looking at the research ethics governing the data collection and analysis process, and the process by which the quality of the current research was maintained.

3.2 Research Paradigm

This research aims to develop a framework for understanding the value co-creation process in the football industry through the use of knowledge discovery systems. In order to develop the philosophical stances governing the research strategy by which to fulfil this aim, it is necessary to first identify the nature of knowledge (ontology) and how to obtain this knowledge (epistemology).

The two main research concepts are the interpretive paradigm and the positivist paradigm, each of which is built upon certain ontological, epistemological and axiological stances (Creswell and Miller, 2000). The ontology defined "what" is the knowledge and characteristics of the knowledge (Teddlie and Tashakkori, 2012). This can be a single global reality or numerous different contextual realities (Walsham, 2014). Ontological stances can characterise the definition of science in terms of relationships between concepts (Poli and Obrst, 2010) or the understanding of a given phenomenon (Walsham, 1997). The positivist paradigm defines knowledge as a set of relations between different concepts (Venkatesh, Thong and Xu, 2012), whereas the interpretivism paradigm believes that knowledge cannot be understood through relationships and requires in-depth investigation. Because of these beliefs, positivists define a framework as an abstraction of reality and models as sub-sets of that framework

(Tashakkori and Creswell, 2008). The model is defined as a set of tested relations for the positivists while interpretive defines the model as a sub-set of the framework (Gregor and Hevner, 2013). The aim of the current research is to develop a framework for understanding the value creation process, rather than testing it. Therefore, the knowledge shall be framed based on the interpretive ontological stance, in which reality is abstracted in terms of value creation in the knowledge discovery process, rather than testing relationships between concepts.

Ontology discusses whether reality is external and unknown, or internal and known. In other words, ontology governs the question of whether respondents are assumed to 'know' the knowledge, which the researcher must grasp, or the respondents are assumed to not 'know' the knowledge, meaning that the researcher should discover the relationship between concepts. Positivism primarily believes that knowledge is unknown and that correlations should be derived to test propositions; interpretivism mainly believes that knowledge is known and so researchers should capture the insights of participants. Because the KD process happens intentionally, the respondents shall be informed of the intentions and rationales of their activities, meaning that the interpretivist ontological stance is most appropriate for adoption in this research.

Epistemology is concerned with the question of 'how' to gain knowledge (Lincoln, Lynham and Guba, 2011). Here, positivists believe that the best way to gain knowledge is to take an active position in setting propositions, from the literature, and testing it. In contrast, interpretivists believe that researchers should understand the knowledge, enabling them to draw the abstraction of reality in a framework, in its context (Walsham and Waema, 1994). The interpretivist believes that the researcher shall understand the knowledge, to draw the abstraction of reality in a framework, in its context, which is called social construction of reality. The current research follows the interpretivist paradigm to socially construct reality in the context through face-to-face discussions with each participant, supplemented by examination of their practices and behaviours. This is important because the different financial support and resources available to coaches in each team creates differences based on context. Furthermore, each manager has an individual rationale for the various practices involved in their role, making reality different in each context. This is important because the contingency theory stipulates that there is no correct prescription for all managerial and decision making issues, because the

best mechanism should be used to fulfil specific aims in specific contexts (Drazin and de Ven, 1985). Accordingly, it could be important to understand each coach practices, rationales, and intentions alone then consolidating different views in different contexts to develop a comprehensive framework.

Axiological stance defines the level of subjectivity or objectivity in a given piece of research. Positivists believe that researchers are fairly objective and that they use objective tools to test the propositions of their studies (Mertens, 2007). In contrast, interpretivists believe that a researcher will inevitably have a subjective understanding of contextual factors and, by extension, reality. From this perspective, the background and education of the researcher influences how the perception and interpretation of interviews and facts surrounding them. Accordingly, the researcher has taken many courses in coaching and data analytics to be able understanding the participants thoroughly. Although this can provide insights to a problem, too much subjectivity can destroy the credibility of the research, which is a risk inherent to interpretive research, such as the current study. For this reason, this research will adopt a conceptual framework derived from the literature to guide the data collection, data analysis, and reporting process (Aliseda, 2006). This framework is not intended to limit or propose certain relations, but rather to highlight the concepts that need to be investigated to ensure less subjectivity in perception and understanding. This is supplemented by use of validation techniques, to ensure that respondents have similar understanding of the researcher, and the use of a verification tool to construct new arguments for those with less knowledge or awareness.

3.3 Research Strategy

This research is a multi-phased (three) study based on a design science research methodology (March and Smith, 1995), meaning that it seeks to address and develop a solution for a certain problem (Peffers *et al.*, 2007). The aim of the current research is to answer the following question:

"Why are some teams better able to get value from investing in knowledge discovery technologies than others in the football industry?"

A maturity model was required to assess the abilities of teams to realise the value gained from investment in KD technologies. In the specific context of this study, design science differs from traditional research approaches in the sense that it not only attempts to understand the reasons for differences in realising the value, but also seeks to develop tools for teams to benchmark their capabilities. The development of such a tool requires that research concepts be identified, taxonomised, and operationalised into scales for assessment (Gregor and Hevner, 2013). In the current research, this process involved operationalising the expected value of KD to the coaching team, identifying and categorising KD resources, and depicting a model of the role of football technologies in improving coach performance. This was carried out to enable the research to identify and frame the role of different knowledge, skills and competences required from producer (i.e. data analyst) and consumer of the knowledge (i.e. coach), thereby enabling the expected value from KD to be realised.

As recommended by design research (Hevner, 2007), this research is organised into three phases. The first phase sought to understand the AS-IS to develop a framework that explains variations in understanding to develop the initial framework, models, and tools based on an assessment of coaching awareness and use of KD in developing football related strategies. The second phase involved the development of the maturity model. Finally, the aim of the third phase was validating the framework and verifying the maturity model, in order to ensure the quality of the research outputs. The interviewees listed in Table 3-1, Table 3-2, and Table 3-3 will be discussed, analysed, referred, and discussed in more details in Chapter 4 Chapter 5 Chapter 6.

3.3.1 First wave of interviews: Developing the framework

Development of the framework started with in-depth interviews with experts, coaches, data analysts and coaches. The following sections outline and discuss the sampling method, interview guide and analytic approach adopted in the development of the research framework utilised in the current research, as well its subset models and tools.

3.3.1.1 Sampling

The sampling approach adopted in this research is purposeful and selective (Suri, 2011). This means that the selection of respondents is based on strict criteria, in order to inform the development of a constructive framework to aid teams for improving the value realisation from investing in KD technologies. Because of this aim, well-informed opinions were required, which means that participants needed to be experienced. Therefore, the first criteria were three years of experience in football. The second was that experts had senior positions in their teams (only team managers, coaches, and data analysts). The third criteria were that participants had to be based in the chosen

geographical scope of this study: i.e. Saudi Arabia. Interviews with experts from the British League were used to advise on the current status of Saudi practices, so that a constructive framework can be developed. A pilot study was conducted and then interviews were held to discover current practices in utilising the KD technologies in Saudi Arabia. Participants were coaches, data analysts and team managers from the premier league, youth league, second league, and Olympic league in Saudi Arabia, as well as certain high-level decision makers in the Saudi Arabia Football Association. Since 2018, there are sixteen teams in the Saudi Professional League (SPL) (GSA, 2018), as well as one Olympic team, a youth team, and a national team (SAFF, 2018). These teams constitute the population of this research. Invitations were sent to all eligible coaches, team managers, and data analysts in Saudi Arabia. Interviews were scheduled for a duration of 1-2 hours, in an attempt to understand their perspectives on optimal usage of KD resources.

Code	Team / Bodies	Role
W1TD	Football National Teams	Technical Director & Expert (Coach, Player in Different Leagues)
W1FC2	Football Club - 1 st Team	Coach
W1DA3	Football Club - 1 st Team	Analysts
W1DA4	Football Club 1 st Team	Analysts
W1FC5	Football Club - Olympic Team	Head Coach
W1DA6	Football Club - Olympic Team	Analyst
W1FC7	National Olympic Team	Coach
W1FC8	Football Club - 1 st Team	Coach
W1FC9	Football National Teams	Assistant Coach
W1DA10	Football Club 1st Team UK	Analysts
W1BM11	Football Organisation - UK	Performance Analysis Team Member
W1BM12	Football Organisation - UK	Performance Analysis Team Member
W1BM13	Football Organisation - KSA	Technical Committee
W1RS14	Rugby Club	Director of Performance Analysis
W1PSC15	Sports Data Consultancy	Data/Video Analysts
W1BM16	Football Organisation - KSA	Technical Committee
W1PSC17	Sports Data Specialists - UK	Representative - Sports Data Specialist
W1PSC18	Sports Data Specialists - International	Representative - Sports Data Specialist
W1PSC19	Sports Consultancy - KSA	Manager – Football Data Specialist
W1PSC20	Sports Consultancy - KSA	Representative - Football Data Specialist
W1PSC21	Sports Data Specialists	Live Scouting Administration

Table 3-1: Wave 1 - Initial Participants – Framework Development

3.3.1.2 Interview Guide

The invitation was accepted by five teams from the premier league, the Olympic team, the youth team and the national team. The research utilised semi-structured interviews, examining components of the conceptual framework. As a brief summary, these components are value (i.e. benefits), outcomes (enhanced use of data analytic models and key performance indicators in planning and coaching), the technological resources, human resources (required skills, competences and knowledge), and value co-creation process and tools. The questions focused on the current and required knowledge of coaches and data analysts regarding the effective utilisation of KD technological resources. Since KPIs are the main outputs of KD, questions also discussed the main KPIs used by the team manager and data analyst, how they were developed, and how they benefit from KD resources. Finally, interview questions focused on the value co-creation process across the stakeholders, looking at issues like communication, tools, and mechanisms used in listening and understanding the needs or expectations of stakeholders, and the types of knowledge required by the team.

3.3.1.3 Analytic Approach adopted

As the sample size in the current research was too small for thematic analysis, other analytic approaches were evaluated for adoption. The closed coding was used to guide the data collection and analysis process (Charmaz, 2006). The theoretical lens of this research was provided through a combination of the Melville framework, benefits mapping, value co-creations, and Agile frameworks, which are discussed in depth in the final chapter. Closed coding does not mean rejecting other interesting ideas (Urquhart, Lehmann and Myers, 2010), instead providing a flexible, open starting point for the analysis. The significance of the rationale is weighted in terms of the plausibility and feasibility of a statement, as perceived by the researcher, rather than number of respondents. Although this approach has an obvious inherent bias, the results of the research were contrasted with the literature to ensure the validity of the research findings, as well as being subjected to the aforementioned validation and verification process in the final phase.

3.3.2 Developing the Maturity Model

Maturity Model (MM) were developed using the results from the previous phase. The MM was adopted for this development process and customised to meet with the specific
needs of this research, as they, e.g. CMMI and DELTA (see section 2.6.5) was originally intended for use in software development, therefore requires adaption for new analytic contexts (Kim and Grant, 2010). However, this research aims to developing new analytic models to develop new technologies are similar process. This research borrowed the CMMI from the software engineering literature to apply its approach to KD literature. Questions were prepared on the Likert scale (5 items, from strongly disagree to strongly agree). The maturity model was designed immediately prior to the framework being completed. The face validity of the model was ensured by having three interviewees answer the questionnaire, to ensure that any misunderstanding or miscommunications were eliminated, which improved question clarity and focus for the respondents.

3.3.3 The Second Wave of Interviews: Validating the Framework

The second wave of the interviews were intended to validate the results of the first wave, as well as to validate the maturity model. This wave focused on obtaining data from experienced individuals in senior positions, due to the belief that they would have greater insight into the applicability and usefulness of such technologies. Interviews were conducted with those in the Saudi Arabian Football Federation positions, such as the executive manager of the technical committee, technical committee members, and the director of the Academy of Football. Senior figures from local teams were also interviewed to gain their perspectives on results.

Code	Team / Bodies	Role
W2TD1	Football National Teams	Team Director – Former Player – Former Coach
W2EM2	Football Federation	Executive Manager of the Technical Committee
W2TCM3	Football Federation	Technical Committee Member – Professional coach
W2FC4	Football Federation - Football Club Academy	Professional Coach – Academy Director – Former National Team Coach
W2FC5	Football Federation - Football Club Academy	Professional Coach – Academy Director – National Youth Team Coach – Scouting & Talent Identification
W2FC6	Football Federation - Football Club Academy	Professional Coach – Professional Player Mentor
W2FC7	The UK Football Association	Professional Coach
W2FTM8	The UK Football Association	UK Team Manager
W2SC9	University	Principal Lecturer in Sports Coaching Science
W2TD10	Football Federation	Ethics Discipline Committee

Table 3-2.	Wave 2 - 2	2nd Particinants	(Framework	Validation)
10010 5 2.		ind i di ticipants	(IT UTILE WORK	vanaationj

3.3.4 Third Wave of Interviews: Verifying the framework

The last wave sought to assess the weaknesses and strengths of each team regarding the technological and human resources required to use KD effectively. Data analysts and coaches from five teams completed the assessment and were sent the analysis, after which their feedback was received and analysed. The five teams represent different leagues in Saudi Arabia, in order to achieve a representative view of teams in different categories. Thus, two teams were selected from the professional league, Olympic national team, under 19s national team, and one team from the secondary league.

Case	Code	Team	Role	Team / Bodies
Case 1	C1P1	CT1	Video Analyst	Football Club - 1 st Team
Case 2	C2P1	CT2	Assistant Coach	U19 National Team
	C2P2		Data Analyst	
Case 3	C3P1	CT3	Data Analyst	National Olympic Team
Case 4	C4P1	CT4	Assistant Coach	Football Club - 1 st Team
Case 5	C5P1	CT5	Football Coach	Football Academy Club

Table 3-3: Wave 3- 3rd Participants (Verification and MM - Application)

3.4 Research Quality

The research quality process serves to ensure that a study produces outcomes that reflect reality and that the data obtained are suitable to address the research question, especially given the subjectivity way in which human understand and process ideas, which is integral to interpretivist research. During this process, five main indicators are considered in order to determine quality: reliability, validity, credibility, reflexivity and transferability. These criteria will be briefly outlined below in general terms, and then discussed in detail in the following sections.

Reliability describes the truthfulness of participants in a study (Rossman and Rallis, 2003), which is important because people can become inclined to misrepresent their perspectives or to give intentionally mislead with their answers for a variety of reasons, including a dislike for being observed, a feeling of shyness, or even because of a perceived lack of security (Kirk and Miller, 1986). In recognition of this tendency, a number of specific measures were adopted to ensure the integrity of the interview and to guarantee the ability, and willingness, of participants to speak freely on the topics under investigation.

Validity describes the accuracy with which a researcher has understood the statements and utterances of the participants in their study (Arksey and Knight, 1999; King and Horrocks, 2010). Therefore, I took extensive notes and paraphrased the answers of respondents, then checked these interpretations with the interviewees in order to assess whether my understanding of their meaning was a fair representation of the meaning that they had intended to convey.

Credibility describes the correctness of any records made of participant contributions, in terms of accuracy in the `observations and transcripts (Lincoln and Guba, 1985). Throughout the three waves of interviews, there were prolonged engagements with experts in this domain in order the understand the context and dialects of the coaches, team managers, analysts and relevant stakeholders. Additionally, I took professional analysis courses, official coaching certificates, attended workshops, conferences and professional training to gain as much as possible of up to date experts' knowledge within the field. Moreover, I met with football journalists, football associations and football federations members to gain knowledge of current trends and current practices and issues of this area of the research (Lincoln and Guba, 1985). In order to ensure that a high degree of credibility was achieved, all interviews were recorded, transcribed and translated. Exceptions were made in those situations where participants explicitly forbade recordings, such as for reasons of security. In these cases, a degree of credibility was sacrificed in return for access to valuable data.

Reflexivity describes the relationship and the reflection of the researcher on his research. It is a process of which the researcher engages with his research, communicate with interviewees, develop and improve knowledge in the research area, observe and analyse interviews and questioners. This should lead to maturing the knowledge gained from aliening that with literature readings to consolidate concepts and understanding. As well to which the framework utilised in the study is trustworthy and the rationale for the interpretation of data is suitable for its intended purpose. In consideration of this issue, the literature was evaluated to ensure that the researcher was cognizant of the salient facts. Finally, transferability, which concerns the applicability of the research practices and finding outside the research context and how this research could aid in other research areas.

Table 3-4: Summary of research qualities

Research Quality	Definition	Measure used
Reliability	Participants are telling the truth and their discourse is constant over time.	Asking the same question in different ways at different times. Showing a consent form to confirm that all data gathered are confidential and will not be shared with anyone.
Validity	Ensure that the understanding of the researcher matches that of participants.	Paraphrasing the answers and ensuring similar results. Emails were sent to give confirmation of the interviews.
Credibility	Ensuring that everything written is correct, truthful and trustworthy.	Consistency of the report with few contradictions between findings by using a recorder and recording interviews, then checking transcripts.
Reflexivity	Ensuring that the researcher's understanding is correct, and her interpretations are the most suitable in the context.	Evaluating different possible understandings of the facts using literature review and different rationales.

3.5 Research Ethics

This research study targets experts in the domain of football coaching because the study involved human participants, it was necessary to consider ethics and gain ethical approval from the researcher's university. Ethical approval was granted based on that. Towards this informed consent was conducted whereby participants were informed about the purpose of the study and their consent was gained. The participants were also informed about their right to withdraw from the study at any time.

3.6 Chapter Summary

This chapter aimed to discuss and set the methodological foundations for this research. This is an interpretive research adopting design science strategy to develop a framework for understanding the value co-creation process of the Knowledge discovery resources in Football so that a maturity model can be used to assist the team manager evaluating their teams' abilities to realise value from investing in the KD. The literature is used to set the theoretical lens for this study and set the methodological basis for defining the sample, interview guide and analytic approach of this research. This research follows three phases: developing the framework, developing the maturity model, and validating and verifying

the maturity model and the framework. The next chapter will be the first of three in depicting the findings.

Chapter 4 Knowledge Discovery Resources

4.1 Introduction

The analysis is presented over two chapters while the validation and verification are devoted in a separate chapter. This chapter aims to investigate the required resources for gaining value from the knowledge discovery activities while the following chapter focuses on the value of the knowledge discovery activities. These couple of chapters are ended by value co-creation knowledge discovery maturity model to operationalise these research findings into an assessment tool that can help team managers for gauging their capabilities to realise value from knowledge discovery.

This chapter adopts Melville et al. (2004) framework in identifying and classifying the required resources. According to Melville et al. (2004), the value from investment in technology comes from IT resources and complementary resources (i.e. people/human resources). The same framework is adopted in this research to understand the required resources.

This chapter structures as follow. After spotlighting the required IT resources as an infrastructure for the knowledge discovery activities, Human resources and their competences are identified. Based on the common challenges faced in the HR communications, proposed Agile methods for improving the knowledge discovery process.

4.2 The KD Technological Resources Model

The first model of this research is to understand the role of IT resources in the knowledge discovery. In this model, it shows that there are five types of technologies are perceived to be critical for the knowledge discovery. They are tracking technologies, body sensors, annotation software, database interfaces, and knowledge discovery analytics. They have been classified into hardware and software categorise as visualised in the Figure 4-1. Data capturing technologies in terms of tracking technologies and body sensor aim to capture live data in the match time and in the training time. The annotation software is to give meanings to different data taxonomies and ontologies. The historical data are combined in the analysis through having an analytic interface to retrieve and use historical data. Finally, data analytics systems consolidate these data, after cleaning them, to develop new analytic models or KPIs for discovering new knowledge.



Figure 4-1: IT Resources model for enabling knowledge discovery - © by the researcher

4.2.1 Hardware required to collect the data

The KD process in the football industry requires technologies to record and track the players' movements and to measure the external environmental factors. The technologies can be classified into body sensors and tracking technologies. Body sensors are those wearable devices that track the body physical performance. The body sensors outputs include the heart rate, VO_2 max, and B2mx (Oxygen in blood).

"Yes, maybe I know where you are going. I have many positions in the field, but also I have many kind of measuring VO₂ max, I can check if the player breath under masses, have all kind of information. I want to show you know, let's see what happen" W1FC2

In contrast, tracking technologies measure and report the positions, movements and actions of players, in addition to weather indicators such as temperature and humidity. Tracking technology generates massive data that can motivate the coach to explore and discover the implications and uses of these data. This argument is conditioned by certain capabilities required from the coach, as will be discussed in team manager and coach competences sections.

Not all teams in the sample were found to have access to similar technologies: two teams out of eight have all technologies, with seven having tracking technologies, and one team (T3) having no access to any form of technology at the time of interview. Two months after the interview, the researcher was informed by the team analyst that they had bought tracking technology. Limited or no access to tracking and annotation technologies can be perceived as a disadvantage that constrains the KD process, however manual data collection is a viable substitute to other technologies.

"We do that manually. We do very primitive analysis due to limited access to first hand data. Through my assistants, we can calculate number of successful and unsuccessful passes and number of shooting. We cannot do more due to the technological constraints." W1FC8

The managers or the coaches of teams without KD technologies must often rely on secondary data or analytic reports published by specialised companies. However, these commercial secondary data have certain inherent limitations, because the data are not updated, not in-depth, and not customisable. Outsourced analysis from external sources is limited to camera-based data, which still lacks the use of sensors in tracking physical performance. The outsourced analysis can feed in tactical and technical performance and basic physical performance.

"Just the one I show you before, Stats. With Olympic team we use video and trying to get GPS data" W1DA4.

"We use InStat before with national teams but now I only use visual observations." W1FC5

4.2.2 Software: Knowledge Discovery System

Software applications required for KD are three: data base interfaces, annotation, and analytics. The three technologies are perceived to be important. The first software required, which is usually web-based service outsourced from external vendor, is the interface for accessing the vendor data of other teams and other matches in the league. This is the most accessible and prevail across the teams that are affordable to subscribe.

"I provide the data in different format to reflect in his view. We use Opta reports and InStat to access secondary databases for getting information about players and teams in the league" WIDA4.

"Yes, that is the problem I had, when I first got Opta data about last year, and when I got it last time, I was amazed about it, what is this. It was 90 minutes of events-based dataset of everything happening in the Match. What will I do with this, I have not seen dataset like this before. I said ok, that is a good thing, it is a good structure and good quality." WITD. Using this data, a coach or data analyst can model and understand the performance and behaviours of opponent teams in a variety of different situations. The second software application is for annotation, based upon code data received from the sensors and tracking technologies. These technologies generate huge quantities data, which can be coded and annotated for comparison with external data, or which can be modelled and analysed by analytic applications.

"With Olympic team video analysis and observations. Stats, ProZone when I was the analyst of the first team." W1DA6

"Regarding the methodology, I was thinking about linking the data between the 1st team and the Olympic team, not only using one tool or application, but using any suitable, useful technologies that could provide a future benefit. All the data we have starts from the 1st team, though. There is nothing from the teams before." W1DA4.

The main benefits of annotating the primary data is the availability and its depth, which are not available in the secondary data. However, it needs more technical human resources, as will be discussed later. The reason is that it implies more coding and customisation abilities than the standard reporting system such as Opta and InStat. Thus, fewer teams are adopting this type of applications.

"We have software to annotate movements. It gives us the information about the real time and game we can see that the midfield covered more distance then the forward player, the midfield they covered the medium speed in the game 5km per hours and the forward covered less distance that speed in the game use 10 or 8 km in the average so that the different." W1FC2

Finally, the data analytic applications can facilitate the development of sophisticated analytic models to enable the creation of effective, targeted knowledge (e.g. predictions, estimations, probabilities, scenarios, confirming and verifying certain believes or plans) to meet the specific needs of each team.

"We use Top Sports Lab to predict performance and injuries" W1TD

"We use fitness data, we work also with InStat football (InStat Scout) very nice algorithms and key performance indicators for players and teams' performance (technical and tactical data)." W1TD In summary, all of the teams utilise software applications that provide access to secondary data. Four of the eight are using the annotation technologies. The reason for this choice seems to be less concerned with the technology and more with the perceptions of the team managers and the specific capabilities of the data analysts, as will be discussed in the human resources sections below.



Figure 4-2: A model to show the relationship between different software applications and hardware in discovering knowledge - \bigcirc by the researcher

4.3 The KD Human Resources Model

The main actors in the value creation process are the consumer of the knowledge (i.e. team manager) and the producer of the knowledge (i.e. data analyst). The various capabilities and the relationships between these parties in the KD process are outlined in diagrammatic form below in Figure 4-3.



Figure 4-3: A model to explain the role of different competences for discovering knowledge for football teams - © by the researcher

4.3.1 Consumer of the Knowledge: Coach Competences

The head coach is expected to have complementary skills that enable him to communicate with the data analyst, to understand the data and, to recommend new exploratory questions in such a way that new, pertinent, useful knowledge can be discovered and utilised in the fulfilment of team aims. Without such competencies, communication is likely to be influenced negatively, resulting in the KD process suffering.

"Honestly the board members they did not have sport technical background, so in this case you can only offer basic details. On the other hand, I was developing some reports, and I email it. But there was no feedback and there was no specific methodology, there were no time for this, and the time was limited and only for one season." W1DA4.

The main competences required here are statistical competences are to read the reports, which refers to the ability to interpret the reports and to translate their strategies/ideas into hypotheses/questions, and technological competences, which involve knowing the limitations and capabilities of the systems. Although data analysts have a responsibility for analysis, mutual understanding is instrumental in improving the KD process. It can be difficult to establish mutual understanding when the team manager does not possess the requisite competences.

"[There is a] lack of knowledge among head coaches. This is the main reason for no engagement in the knowledge discovery process from the systems." WITD

The summary of key findings is summarised in Table 4-1.

Table 4-1: Summary of key findings regarding the role of the coach competences on the effectiveness of knowledge discovery process

	Findings
1	The team coach perception of the usefulness of the knowledge discovery and positive
	attitude towards the knowledge discovery technologies play a role in utilising benefits.
2	The problem formulation is based on the team coach's ability to comprehend the
	problem
3	The team coach is bounded in his ability to comprehend the problem by his perception
	and his understandings
4	Team coach who has more experience and awareness of the knowledge discovery tools
	and knows more about the data science, are able to articulate the problem in a more
	detailed and data driven way
5	Team coach who are aware of relevant data and have analytical understanding of
	available data are able to address their team performance needs precisely.

- 6 Team coach who has more closed relationship and more frequent meetings with the data analysts are more able to define their needs in a more specific and clear way. Also the reports used are richer than others
- 7 The problems are structured over the time. Each coach by the time has certain needs and certain problems. This can hinder the team manager to be able to discover and explore new insights from the data.

4.3.1.1 Statistical Competences

KD axioms are based on the discovery of insights from data that enable the discovery of new knowledge. Research states that the attitude of team managers regarding statistics can determine the intention to use KD practices. According to the theory of reasoned action, when users believe that a system could improve performance and productivity, their intention to use the system improves (Ajzen and Fishbein, 1980).

"So, what I believe is that we are not ready yet, with data science and only few coaches use it or either afraid of it, they do not use it because they are afraid of it but I believe that its will be one of the most important aspect of football" WITD.

"Couches are too afraid, there are no secrets" W1TD.

The knowledge and awareness of the statistics and acknowledging the importance of it plays a significant role in shaping the attitude toward the use of statistics in discovering knowledge.

"Were to board members were not aware of the benefits these tools or this information could help and supports them, were thy naïve about it, don't know what the outcomes of it, what they could achieve of this, what opportunities of could lead to, the technical staff were worried about what data could leak or exposed or the players could not understand it and cannot digested it, many things really, and my responsibilities were very limited as an analyst." W1DA4.

On the other side, some coaches have a belief in and understanding of statistics, enabling them to do more through the KD process.

"Statistics are all information that we know about the players. We can rely on it. We process and clean data to be used by data analyst and myself. I am repeating that statistics are very powerful and can change the game rules, only if you understand, read and able to interpret them. In this way, it can transform the team manager thoughts, expectations, and strategies" W1FC9 If the coach knows statistics, he will be able to propose questions and ideas that can be answered through the data.

"Depends on the coach, for example, other coaches may ask more about players data because he is into analysis and technologies" W1DA4.

"It is honestly depending on the coach and what information he wants to use to his advantage, and fully know how to use it and utilise it." W1DA4.

"When I was working in Marseille with "Marcelo Bielsa" and he is crazy about statistics and performances, we collect all of the data and based in these data we talk about fitness. Marcelo knows very well what he wants from the data and how to deploy them in his match strategy development" WITD.

The coach can recommend how analysis should be done, in order to ensure that he can access the required outputs.

"It depends what I need from the data. I direct the data analyst how to analyses to get what I want. The more the classifications of the data, the more I will be able to get what I want" W1FC7.

4.3.1.2 Technology Competences

More useful and relevant knowledge is created when team managers are better equipped to use and understand the functions and capabilities of data analytic systems.

"[We are] Happy about the system used in the Saudi Team. I think in general that data mining can be much more successful, and I think the issue is a lack of knowledge among the head coaches can be a bit of a problem." W1TD.

In other words, coaches must believe that the system constitutes an effective mechanism with which to improve their job as coaches.

"I take all this information. Then I take decision depends on how my player need sprint session or strength session, fortunate. I have all this information to help me. So, when I make my planning the main aims is the game." W1FC2

4.3.2 Producer of the Knowledge: Data Analysts Competences

The modern game in football has been influenced by the emergence of numerous technologies over the last decade, which have had far-reaching repercussions for the

reading of results and prediction, through complex analysis across various levels of interpretation. These data gathering and collections practices, collaboration and processes are generally accepted as enriching the data available to coaching staff. However, the demanding, high-pressure nature of football competition requires practices to fulfil the required tasks meaningfully, efficiently and effectively while still meeting ongoing yet critical weekly deadlines. The producer of such knowledge is required to understand the needs of the knowledge consumers, which requires familiarity with football knowledge and the ability to construct reliable and effective communication channels with the consumers. In addition, analysts are expected, by default, to have strong statistical and numerical competences that enable them to effectively exploit the data and translate them into knowledge that meets the needs of consumers. The summary of competences are listed in Table 4-2.

Table 4-2: List	of competences	required by	the dat	a analyst to	have an	effective	knowledge	discovery
process								

Competence	Rationale					
Communication	For collaboration					
Competences	Identifying/understanding the requirements and					
	needs To identify which information shall be shared with					
	different stakeholder (Motivational, envy)					
Football Planning	1) Communication/collaboration 2) Proposing					
Competences	new ideas to the coaching team 3) Interpreting the					
	reports 4) Improving Active Listing 5) Expediting					
	understanding required task.					
Statistical Competences	Data mining / modelling for converting data into					
	insightful reports, deriving insightful interpretation					
	and reports from data (i.e. translating requirements					
	into data and statistical algorithms)					
Technology	To identify/use the best hardware and software with					
Competences	lowest cost and highest effectiveness. Also improves					
	the ability of use technologies more effectively and					
	efficiently					

4.3.2.1 Football Planning Competences

In order to have effective communication between the data analyst and team coach, it is essential for the data analyst to have good knowledge of football, as this can help in the communication/collaboration process, as well as with interpreting reports and proposing new ideas to the coaching team.

"For me, before anything else, he shall have good knowledge and passion of the football technical, tactical, and physical aspects" W1FC7

The consequence of this is that coaches sometimes set certain requirements in terms of education and background, while others give specialised training to their data analysts to get the right knowledge about football

"Having knowledge of football is critical. Also, having a technical and tactical perspective is important for me, because this helps me to communicate with him and can help him to give me interesting ideas through the data. Sometimes, I prefer if the data analyst has experience in coaching as well" W1FC8

"It is vital to have a profound knowledge of the football. The analysts are working with team managers and others; he shall be able to see things differently. It is not only about data" W1FC7

Communication not only comes from talking the same language, but also denotes the ability of the data analyst to translate and explain data in more useful ways (i.e. wisdom).

"That is why this needs to go step by step and as a sport scientist. The most important thing is when you talk about data mining/science is to show people understandable data." WITD.

Theme	Sub theme
Knowledge in football	Communication (improve value creation process)
	Proposing new ideas
	Interpretations

4.3.2.2 Statistical Competences

Data analysts also require sophisticated numeric ability. Although this seems natural, in the context of the Saudi teams, some of the data analysts demonstrated a low level of familiarity with statistics, which resulted in the need to outsource reports from professional firms. However, it is generally argued that data analysts should be able to model data in appropriate time.

"He shall know statistics and familiar with SPL statistics to identify different useful KPIs" W1FC9 This statement is interesting because the understanding of the concept statistic is limited to the KPIs without modelling. In other words, the concept of modelling the data is not prevailing, with only two team demonstrating knowledge about the modelling process.

"When you have millions of data, you have to put them in a system, and you have to start looking for relations and correlation between the data. That's your work." WITD.

Other teams demonstrated a limited understanding of the statistics, with their apparent perspective being confined only to a perception of the role of the data analyst being to generate descriptive data.

"Faster analysis, qualitative insight than the stats we collect." W1DA3

Across all teams, a serious gap was identified between the analytic models discussed in the literature and the actual knowledge or practice of the data analysts in Saudi Arabia. No team used simulation models or sophisticated analytic models. This can be perceived as a weakness in their data analytic knowledge regarding quantitative analysis models, techniques, and software applications. None of the data analysts interviewed were aware of any data analytic models, as they relied exclusively on Excel software to perform basic mathematical calculations. Ironically, no data analytic staff came from statistical or mathematical backgrounds, with all noted analysts coming from sports, business, or physical background, with only a few having had training in statistics and numbers. This may be a factor undermining the value of KD resources in this context.

4.3.2.3 Communication Competences

Communication competences are another important area for data analysts.

"Communication is the key, it will be easier to work with him (coach) and do my job, you understand him. You know his goals, objectives, he is clear, you understand how he work, what he wants it will be easier for me to do my job. It is better than working with someone who does not have clear idea of what is he doing, or not organised it will be hard to be to do my job." W1DA4.

This research defines communication competences as the ability to listen actively and understand question in such a way that problems can be translated into sets of knowledge requirements, in addition to the ability to explain findings to the users of the data in meaningful terms and circulate them to the right people with a level of trust. Trust and mutual understanding are complementing each other.

"This is a compulsory competence. He shall be trustfulness. Trust shall be in communication and circulating the knowledge, and in the quality of information presented. Without such trust, communication will never be established. Also, data analyst shall be able to understand the team managers' information needs" W1FC9.

Communication is critical to the knowledge co-creation in the coaching team, including the data analyst.

"You as a data analyst, sometimes you can see things differently. Also, could be against the beliefs and assumptions of the team manager. If there is a mutual trust between you and the team manager, he will listen to you. Communication is all about mutual understanding and trust between you and the team manager. W1FC9

There are two main competences that data analysts should develop in order to construct effective communication with their team managers: developing proper communication channels and determining the appropriate level of data interpretation.

• Developing communication channels

Data analysts need to construct proper communication channels to enable them to identify which, when, and to whom information should be released. Communication also involves circulating and distributing reports to the right persons at the right time.

"The best data analyst for me is the one who understands what I want and knows very well which information can and shall be shared with the players and which are no" W1FC8

"These data have high level of privacy. Data analysts shall not disclose or share any information in formal or informal meetings with anybody" W1FC7

Certain information is also believed to have the potential to have negative psychological impacts on some players. For instance, in scenarios when the performance of the players on opposing teams are significantly better than the team, some coaches feel that arbitrarily sharing this information about their weakness could have a negative influence on performance and morale.

"The reason why I hate the data analyst share information to players is that players are comparing each other's too much. This has negative influences on their morale" W1FC8

According to the perspective of this team manager, telling players about their weaknesses in a more convenient way influences their morals.

By having proper communication channels, the right person is expected to receive the right information and to inquire the useful information. This can lead to the delivery of the right knowledge to the right person, potentially improving the use and value of the knowledge. In addition, the interaction between required knowledge and what is available could stimulate the data analysts to think differently, leading to the discovery of new knowledge.

• Interpretation level

The reports created by data analysts are the ultimate knowledge with which team managers can improve performance. Even the presentation of this data is important, in order to maximise its usefulness.

"No, the collections are good but the way of representing the data is not that good." WITD.

"And the same for data scientist, you can come up with the best data ever but if you cannot convince the head coach to use those data to change his tactics what are you going to do with it. Then you are just analysing data for the sake of analysing data, which is not wrong at all - Because it is also a scientist". WITD.

In order to be useful, reports must present findings in a comprehensible format that is easy to read and process. The findings illustrate that the overreliance on specialised jargon adversely affected the ability of some participating data analysts to effectively communicate their knowledge.

"So you see interpretation of data is very successful but yet again I may think people like you, data scientist, will be probably one of the most important people in a team but the biggest problem now is that people like you they are presenting their KPI's to coaches while they should ask coaches, listen to the coach, what do you want to know, which KPI do you want to receive and then you need to make these KPI's understandable and that is your challenge" WITD.

This indicates the importance of data analysts giving useful interpretations of the results in order to help the head coach to make the right decisions. At the same time, it must be remembered that excessive subjectivity of interpretation can also be misleading. "Sometime as an analyst there are boundaries, you don't push too much, the head coach does not want too much interference with him. So, know it's not providing information but interference in his work. You will find yourself ignored or even kicked out if you push it too much. There is thin line, provide information, back off, give him space and let him work with the data." W1DA4

These findings are an effective summary of the relative pros and cons of the interpretations that data analysts perform of the data. The pros are primarily related to the ability to help the coaching team understand data and obtain insightful ideas from reports. The cons pertain to the interpretations that are bounded by the knowledge that data analyst has in their specific field (here, football), which is closely related to the ability to inform or mislead the coach in terms of reaching effective decisions.

- Help the Coaching team to benefit from the data

- Gives insightful and new ideas to the coaching team

Advantage of Data Analyst Interpretation

- The intrepretation is bounded by the technical abilities

- This could mislead the Coaching team if subjectivity interfers the implications of interpretations Disadvantage of data analyst interpretation

Figure 4-4: Pros and Cons of data interpretation by the data analyst

4.4 Proposed Value Co-creation Process Model

The last section clearly advised that the key in the discovery of knowledge is communication between data analyst, coach, and other stakeholders. In other words, value is created through discursive and constructive communications: this is called the value co-creation process. In the classic approach, data analysts produce reports, regardless of the real needs of the head coach. This is a challenge perceived by the head coach. "But most sport scientist they go to the raw data and develop some KPI's and they say to the head coach. It should work the other way around" W1TD.

In other words, the KD process traditionally begins with the production of the reports without filtering the knowledge to different actors, or even identifying their specific needs and requirements.

"Analysts record a game and then send the analysis to the locker room. All the information about the tactics and what happens in every match half. Then we try to fix our play or change our tactics." W1FC2

"Mostly Reports for the technical staff as well for the players." W1DA4

This approach limited the KD practices and abilities of the participating teams, since communication was one way and was not perceived as an effective co-creation process in discovering knowledge, resulting in harmful rigidity in the formulation of team strategy.

"There is no flexibility to change the team positions / game model. The change is for limited positions. The determination of the communications between the coach and data analyst is fixed and rigid." W1TD.

This even restricts the data analyst from discovering new knowledge, as there is no reason for that event to occur.

"Also, I was very limited in scope by what the needs and requirements of the team manager and I cannot do as I wish. If I had more responsibilities from the board member then it would be a different scenario." WIDA4.

However, this can be addressed by proper co-creation of the knowledge among the stakeholders. There are three proposed methods for improving the value co-creation process between data analysts and coaches: user stories, sprints, and backlog. These strategies are all borrowed from Agile methodologies literature and are discussed in more detail below. The process is visualised in Figure 4-5, and detailed in the following sections.



Figure 4-5: The proposed Value Co-creation Process Model - © by the researcher

4.4.1 User Stories / User Question

In an Agile process, user stores are a technique for the development of the features required to meet the specific goals of users. User stories capture what needs to be discovered, based on their specifically articulated acceptance criteria (requirements, questions, and visions), in order to enable new insights to be achieved in the necessary domains.

"I tell the data analyst different scenarios of the match. Then I explain to him what my preferred scenarios is. The data analyst validates what I am saying using his data" W1FC5

"You as a team manager, articulate your point of view in stories to simplify the identification of the required knowledge from the data analysts. As data analyst, you have to understand the knowledge required, the head coach objectives of the analysis, and understand the reason for asking of this thing in depth so that you can do your analysis" WITD.

In this way, user stories help to create a clear written list of what needs to be achieved, by whom and why, then allows that to be tracked and progressed further. There are many stakeholders involved in this process, including: players, head coach, assistant coaches, fitness coaches, data analysts and others (e.g. team manager). This process is based on training or match activities, scenarios, and practices that reflects the performances of players or the team.

"You need to understand the head coach philosophy and approach in training and setting match strategies. How can you know that? It is by meetings and listening to his stories" W1DA3. This Agile collaboration technique helps the stakeholders achieve the following (Cohn, 2004; Drury, Conboy and Power, 2012; Patton, 2014; Orłowski, Ziółkowski and Paciorkiewicz, 2017):

- **Identifying** the promised scope for reaching the intended result by writing related questions to reach new insights.
- Acceptance criteria and understanding about certain features during specific sprints to reach the intended insights of the analysis within the coaching team.
- An approach of **documenting**, **recordings** the analysis process to **maintain** and **preserve** the **processes** and **approaches** taken, as well as the knowledge learned by the different sprints (approaches and processes).
- **Clarifying** the problems to be analysed, breaking down larger questions or request into smaller manageable clearer questions/stories/requests, which can then be further divided into smaller analytical questions (stories dealing with a highly specific purpose).
- **Insights** are then **studied**, investigated, evaluated and validated against questions, stories, and backlogs for confirmation of insights.
- The insights reached are then used in the **learning** process, as well as for **improving** the development of new stories, questions or requirements.

They are the simulated moments required by the head coach from the data analysts.

Table 4-3: Generic Agile Story Template

As <user>

I want to <do something>

So that I can <get benefits>

Table 4-4: Proposed Agile Story Template - © by the researcher

As a <PLAYER> In this role <ROLE>, In this location <POSITION>, When the team is <Moment of the Game>, I want to know <expected performance KPI> , So that I can <UNDERSTAND THIS SCENARIO>

4.4.2 Stories mapping / Questions board (mapping)

Story mapping can also be an effective way to address limited understanding or knowledge about KD tools on the part of coaching team. Through story mapping, the data analyst can translate requirements into hypotheses to be tested or elaborated using statistics and data.

Story mapping starting by a certain objective, then investigates possible scenarios that could influence whether or not these objectives are met. An example of this approach is provided below:

Coach: I believe our team players can perform longer than the competitors. Thus, I can bet over the last 10 minutes of the match. My players can cover more distance faster. I believe that we can play long ball in the last 10 minutes to score goals.

Data analysts can operationalise this story into a set of arguments to be tested: Do our players perform well in the last ten minutes? Can our players perform competent long passes effectively? Do our attackers fall in off-side traps often? Are the defenders of our competitors always coming in front? By breaking down the story into quantitative questions (or cards), an analyst can gain a better understanding and translate the requirements of the coach into required data led information.

4.4.3 The Story/Question/ Analysis Card

As noted immediately above, stories can be structured using cards. This is an Agile artefact to capture, documents and organise conversations about stories to address questions and concerns about the analysis. The proposed story cards contents from a recent study by Patton (2014) which has been tailored to fit with the research scope:

- Author: Stakeholder name for reference and also to clarify any additional information about this card.
- Dates and time: When this card was born.
- Short title: The story analysis/question card should include a short title reflecting that story/question. It should be meaningful and reflective, focusing on the task or question under analysis, enabling the conversation to run smoothly. It should reflect on the analysis task the stakeholders want to learn about.
- **Description**: One or two sentences should briefly describe the action or subject required to examine. It should highlight who is this card about, as well as the action and the purpose for that action (benefits, values and importance).

- **Purpose**: The purpose of this card, the intended way that it will help, and its link to the analysis.
- Story tracking number: A number for easy tracking and future referencing.
- Estimate size or budget: The estimated time or duration of take, with weight and budget if needed.
- **System used:** The tools or systems proposed for use in the discovery process, as well as applications, software or hardware requirements that assist in this process.
- Value: The maximum time and effort accepted for getting the requirement. It is of the story/question in this sprint and how it is reflected in the goals of the intended sprint. Also, to reflect on if this value shared with another card.
- **Importance:** The criticality of the card and its priority in this sprint, or other sprints, using a scaling system (numerical or alphabetical) or coloured indicator.
- **Dependencies**: The link between the card and other cards.
- Status: A record of the progress of this card.
- **Requester:** The stakeholders from the coaching team who requested the analysis.
- Related Cards / Stories: Any relationship between this card and another one.
- Acceptance Criteria: A number of criteria, usually 3-10, to enable a clear scope to be defined, as well as to ensure no ambiguity and uniform understanding across the team.

The proposed examples here are to aid in showing how the FDA process can be done to ease in gathering the needs by coaches, data analysts and the coaching team. They are listed in the next page to ease of use.

These are three proposed example cases for the use of story cards to identify knowledge requirements. These examples story cards are provided below:

Table 4-5: Example 1 - Story/Question Card

Author	Analyst 1	Story	S12	Importance	2	
		tracking				
		number				
Title	Defensive Third	Dates and	18-02-2018	Value		
	ability for player 5	time				
Dependencies	Story 1	Estimate size	Light	Data used	Statistical	
		or budget	_		data	
Purpose	Analysing the defen	sive thirds play	ers and their	Requester	Head	
	abilities for next mate	ch			Coach	
	Metrics					
Description:	As a <player #5=""></player>					
-	In this <centre back<="" th=""><th>>,</th><th></th><th></th><th></th></centre>	>,				
	At this <central defe<="" th=""><th>nding Zone >,</th><th></th><th></th><th></th></central>	nding Zone >,				
	When the team is <lo< th=""><th>osing the ball>,</th><th></th><th></th><th></th></lo<>	osing the ball>,				
	We want to know <% of tackles>,					
	So that we can <know ability="" defensive="" his="" in="" strength="" team="" the="" third=""></know>					
Acceptance	The zones/location where the tackles occurred					
Criteria	• The frequencies in which these tackles occurred in the relevant locations					
	• The players engage	ged in the tackle	S			
	• The time stamps	these tackles occ	curred with resp	pect to match res	sult	

Table 4-6:	Example .	2 - Story/	'Question	Card
------------	-----------	------------	-----------	------

Author	Analysts 2	Story	S13	Importance	1		
		tracking					
		number					
Title	Next match midfield	Dates and	18-04-	Value			
	attacking strategy	time	2018				
Dependencies	Story 2	Estimate	Light	Data used	Statistical		
_		size or	-		data and		
		budget			technical		
					data		
Purpose	Analysing the best strategie	es for creating	g attacking	Requester	Head		
	opportunities for next matc	h.			Coach		
Description:	As a <midfield players=""></midfield>						
	In Role <midfield line="">,</midfield>						
	At this Zone <central th="" zone<=""><th>>,</th><th></th><th></th><th></th></central>	>,					
	When the team is <in ball="" p<="" th=""><th>possession>,</th><th></th><th></th><th></th></in>	possession>,					
	We want to know <attacking opportunities="">,</attacking>						
	So that we can <possible attacking="" scenarios=""></possible>						
Acceptance	• significance level						
Criteria	• Tested and verified usi	Tested and verified using empirical data					
	• Weakness of the opport	ents defendir	ng line				
	• Strength of next match	midfielder an	nd attacking	line			

Table 4-7: Example 3 - Story/Question Card

Author	Analyst 1	Story tracking number	S01	Importance	2	
Title	Midfielders #8 attacking build up passes	Dates and time	20-03-2018	Value		
Dependencies	Story 1 and Story 2	Estimate size or budget	Medium	Data used	Tactical and technical data	
Purpose	Analysing the midfield puilding up attack	haviours when	Requester	Assistant Coach		
Description:	As a <player #8=""> In Role <central midfield="">, At this Zone < Central Midfield Zone >, When the team is <in ball="" possession="">, We want to know <distance 1<sup="" covered="" during="" the="">st and 2nd half >, So that we can < the effective distance covered when the player is building up attacks></distance></in></central></player>					
Acceptance Criteria	 significance level level of accepted error Tested and verified u Best collaborating plate Pros and cons of previous of previous constructions constructions of previous constructions constructions	or sing empirical dat ayers vious matches	a			

4.4.4 Sprint

The sprint is a method used to improve communication and gathering requirements. A sprint is a time boxed Agile technique, with each time box determined in advance. The idea of the sprint is to ensure continuous, iterative data gathering and analysis. Matches are played weekly. Most teams asked the data analyst to develop one report after the match and 1 report before the match (answering queries).

"I do not give all reports one time. This is useless, if I gave all of the information one time, all of us will be confused. I will not be able to address his issues, and he will not be able to fit these analyses in his work. Different reports are produced in different times to ensure the mutual understandings and to address the real needs. The timing of reports is identified at the beginning of the week" W1FC7

"I do give daily reports for the match and the training, based on the needs and requirements. For the match for example, about the fitness, there is another person who gives report an out the tactics etc. I only give about fitness" W1FC2

"One of my tasks is, after every match I make an analysis of what the team made, what we did well and what we need to improve to the next match. I always do this analysis after the match and we always have a plan for next match. So, we work on that and we try to achieve these objectives for the next match" W1DA3.

The main assumptions for these teams were that the data required should be structured and fixed through matches. However, the findings suggest that this did not always, or often, occur. Besides blocking KD practices, the matches and contexts are different, which by nature pushes team managers to look at new information for new matches.

"Also, it depends on the match from game to game, depends which categories of the match (A, B, C) which depends on the opponents, some matches you need more details and information, other ones not that much. With time and experience you really will know what you are looking for. "W1DA4.

The proposed method is the sprint. In the sprint method, it could be expected to have 5 sprints. Starts by a meeting with the team coach to identify the requirements using stories. The following day, 4 or 5 hours as a time box, the data analyst is working alone for working on the questions developed from the stories. Once the questions are identified, the next day meeting is to verify the questions and the potential use of the answers of the questions.

"We have several meetings to elaborate the understandings. It is very important to ensure both of you are on the same page. It is important to define appropriately what is the required information and how will be used" W1FC8

The second time box is intended to derive answers to the questions. The function of the third day meeting is to discuss these answers and to produce any new questions. This case was only seen on one occasion in the current research, but not structured in the typical sprint structure outlined here.

"There are two separate processes here. There is a time for data analysts to do their analysis and other time for discussing the results". W1FC7

"I used to work alone and sometimes I have meetings with the head manager, sport science, assistance coach. We usually have technical meetings before each game (a day or two days before) we explain the performance simple." W1DA4.

"What if Scenarios?" I.e. "if we have this information, can you resolve the problem" W1DA4.

Small releases in the sprint can also be a way to avoid misunderstandings or poor interpretation of the data.

"P2" for example is a player who always finishes the game with almost 100% passing rate, he never ever loses a ball, but he never passes the ball upfront. He recovers the ball and then he passes the ball to "P1", it's, done. So, you can say based on passing rate he is our best player, no of course he is not our best player. He does not lose the ball and that's important because he is the link player." W1TD.



Figure 4-6: Sprint Model for KD value Co-creation framework - © by the researcher

The developed sprint model shown in Figure 4-6 was inspired by the SCRUM Agile methodology. A diagram of the SCRUM framework shown in Figure 4-7 to show where the sprit model should fit within an Agile methodology.

SCRUM FRAMEWORK

Scrum.org

Figure 4-7: SCRUM Framework

Sprint is organised by a backlog that identifies the objective of the analysis, breaking it into small releases with certain acceptance tests. This backlog is revisited at the end of each release to set directions. The following section explains the application of the backlog, small releases, acceptance tests, and retrospective approaches in the specific context of football analysis.

4.4.4.1 Backlog

Backlog is a file utilised to document the objectives of the analysis. This includes a number of factors: the knowledge required (needs), the required software/applications, the required type of analysis, and the overall intended function of the data. The backlog is expected to contain all the thoughts and ideas that the team manager needs to be further explored. The team manager is responsible for formulating such a document with the assistance of the data analyst.

"The organisation of the information exchanges is the key. The documentation also improves communication between data analysis and team managers. In each document, there are

fields for the objectives of the analysis, the knowledge required and other things we need" W1FC8

Despite the theoretical application of this document, the backlog is not always complete and comprehensive. Instead, it is a dynamic document that changes over time, as new stories or questions arise to be addressed, as reflects the analysis needs. On many occasions, team managers do not know the exact problem or knowledge required, so exploration of knowledge from data can potentially be a relatively unstructured process. An example of a proposed backlog is provided below from T3. This backlog concentrates on time, as this was the information that was useful to that team, at that point.

"This process can help to identify all the required information for each scenario, and for each match." W1FC8

	Identify the weaknesses in the team X					
Objective of the analysis	Day 1	Day 2	Day 3	Day 4	Day 5	Sprint Review
Story 1: Defence gaps						
Task 1: Comparison in speeds of defence of the						Day1
opponent with attacker of our team						
Task 2: Comparison in tactics abilities of						
defence of the opponent with attacker of our team						
Task 3 : Identify best and weakest player in the defence						Day 2
Task 4: Identify the weaknesses in potential						
passing in the defence						
Task 5: Review the plan and find out further						Day 4
investigative						
Story 2: Midfields passes						
Task 1: Identify the potential weaknesses in						Day 2
passing between defence and midfields						
Task 2: Comparison in speeds of midfields of the						Day 3
opponent with attacker of our team						
Task 3: Comparison in tactics abilities of						
midfields of the opponent with attacker of our						
Teals 4. Identific notential concertanities and						Dari 5
threats in the expected opponent strategy						Day 5
Story 3: Attacks go through						
Task 1: Identify the weaknesses in passing						Day 4
between midfields and attackers						Day 4
Task 2 : compare attackers speed of the opponents						
with the defenders						
Task 3: set the possibilities of the match win						Day 5
under different scenarios						-

Table 4-8: Example of Backlog file - © by the researcher

2.5.1.1 Small Releases with simple designs

Instead of finishing the whole match analysis in one report, the data analyst breaks the task into several, iterative reports to ensure the continual progression towards the required knowledge/understanding of the problem. As discussed in earlier section on user stories, working on small releases uses a narrow, effective scope for the analytic approach. In other words, rather than focusing on a large, complex task, the analysis is conducted on small tasks in short sprints, which maintains the work-and-confirmation basis. This helps to engage the whole coaching team and stakeholders, especially given the 'story-telling' session after the sprint enables the current state of the analysis to be clarified, ensuring that the visions and requirements of stakeholders are met.

"Our analysis is conducted through layers. I do not do all the analysis and knowledge discovery once. I break down the questions into 5 main questions covering different aspects such as the space, attacking style, corners, fixed balls, and develop probability-based reactions and consequences for each proposed event. I trained on this way when I was in Belgium" W1TD.

"When we do the analysis of the team, we don't take it separately we have to consider these actions with the team. When I start making the analysis after the game, I try to see the most information possible and focus on specific actions. Also look at the dynamics of the team as see each player as well. We need to consider several aspects of the match." W1DA3

Because knowledge required sometime vague and not well specified, the team manager can address the problem and data analyst can propose simple designs - initial information to address the problem

"Similar as we are doing now we meet with the head coach, one to on, effective communications, and I get his request and provide it to him later once it's done." WIDA4.

"Head coach, assistance coach, fitness coach, sport science, goal keeper coach by the way. And I provide it the info for the players. Sometimes they ask about more info and I provide it to them based on their requests." WIDA4.

4.4.4.2 Acceptance Test – Story Acceptance Criteria

After each release of data reports, the team manager validates and verifies the results to ensure that they contribute to the objectives. By defining acceptance criteria, the questions that are similar in scope can be grouped to ensure that issues are worked on effectively in sprints. This also helps to ensure that sufficient understanding of the outcomes of the KD activities are maintained throughout the process.

"Another time for me to replicate and verify the processed knowledge through watching by myself the moment. Usually, analysis is supported by recorded events or simulations to help me verifying the analysis". W1FC9.

"With some data you can have some understanding, but the data is somehow is a good way to start but the not give you the full picture. The statistical data won't give you the full picture but would be the starting point for your research. To make a good decision or good analysis you need to look at some videos to deliver to get the full picture." W1DA3.

4.4.5 Retrospective

The backlog enables to have effective retrospective meetings. Retrospective technique will help in maintaining the iterative need between the questions, research and result so that level or collaboration is achieved as will leading enhancing maturity level in the

collaboration process over time. It occurs after each sprint to collect any improvements by the team. Receiving the team feedback. Linking the reached insights with the discovery process. Revisit the acceptance criteria and work on any enhancement or feedback about them to improve next sprints discovery process.

"So, we know we have the log book and TM goal and objectives is clear (but people have different interpretations than the team manager)" W1FC2.

Retrospective sprints provide an opportunity for the team to reflect on previous sprints, and then suggest plans to improve future sprints. In addition, this stage can help the coaching team find ways to increase the quality of analysis by improving work processes. Improvements and enhancements may be implemented at any time, so the sprint retrospective provides a formal opportunity to focus on inspection and adaptation to perform the next sprints with improved quality of analysis and understanding of the KD process.

4.5 Summary

This research has found that the value of an Agile system can be summarised into the use of new KPIs, which enable the coaching team to develop well-informed recruitment strategies, training strategies, and match strategies. These improved strategies then influence the main objectives of coaching teams, which are the financial return on the player and the match results. The beginning of the value story is the existence and usage of these new KPIs. In order to deliver these KPIs, two conditional capabilities (Agile and KD) should be considered. Agile capability refers to the ability of stakeholders to collaborate in such a way that the development process is lean and flexible, based on a high level of mutual understandings in the team. The Agile capability is developed through effective data governance systems and collaboration requirements, which include mutual understanding between stakeholders, frequent meetings, and the mutual acceptance of who shall interpret the data and to what level. The second conditional capability, KD, is a blended approach using technical and human resources. Technical resources include software and hardware, as well as the required technical abilities to use these technologies. The technical resources serve to collect, analyse and report the required KPIs at an affordable level of effort and cost, in a convenient time. The human resources in KD identify, develop, interpret and integrate KPIs into new practices, based upon specialist skills, knowledge, and abilities (i.e. statistical, technological, and technical football competencies).

Having outlined KD resources in detail, the following chapter seeks to critically discuss the value that these resources and approaches offer for the football teams who integrate them into their systems.

Chapter 5 Knowledge Discovery Value

5.1 Introduction

The aim of this chapter is to operationalise the value of the knowledge discovery. If the knowledge discovery uses and benefits are not aligned with the mission, objectives and tasks of the team manager or the head coach, there is no value of it. As can be deduced from an extensive review of IT business value literature, the capabilities involved in using technology are intimately linked with the ability to derive value from that technology. Value is defined as the sum of the tangible and intangible benefits realised from the change in the current routine due to the existence of a new technology. Related to this, capability denotes the ability to realise the expected value from a certain resource (i.e. the technology). Thus, first, this section covers performance metrics leading his knowledge discovery direction. Based on this, as will be detailed in the second section, the main team manager or head coach strategies will be mapped. Since strategies are formulated based on data (i.e. raw facts), information (processed data) and knowledge (i.e. processed information), the next following sections are to identify and estimate the key performance indicators (as information) of the players and to identify the analytic models that could process these KPIs to deliver knowledge. At the end of this chapter, findings are summarised into the knowledge discovery framework and operationalised into a maturity model.

Figure 5-1:KD Value - © by the researcher
5.2 Performance metrics of the Team manager

There are two main objectives for the managers of football teams: maximising the results of that team, measured in terms of winning ratios in championships and leagues; and optimising the return on investment with players. While significant, this latter objective is less commonly discussed, despite the fact that some team managers are searching for cheap, high performing players, often from South American or African countries, who can be recruited and then sold after development at vastly higher prices. These objectives are commonly achieved through three main strategies: transferring, training, and match strategy. These are discussed below.



Figure 5-2: Team Manager Strategies leading to metrics - © by the researcher

5.3 The KD Outputs: The Role of KD in Developing Effective Strategies

The three aforementioned strategies that team managers (transfer strategy, training strategy and match strategy) use to achieve their objectives are not developed and operationalised at the same time. Knowledge Discovery outputs in terms of new KPIs and new data analytic models could help the coach capture more of the dimensions on the SWOT of his team and the opponent team which enables him to develop a better, knowledge based, and more insightful transfer, training, and matching strategies.

Data analysis in professional sport occurs pre-match, during the match, and post-match. Pre-match analysis aims to set match strategy based on the current information and knowledge of the team and their opponents. During the match analysis aims to monitor and assess that match strategy. Finally, post-match analysis closes the loop by determining the feedback that informs training and transfer strategies.

"Different knowledge required different times. Some of them before the competitions, within the competitions, or even after the competition. The coach could not start his work without having some information about players' KPIs." W1FC7



Figure 5-3: The role of the Knowledge Discovery in formulating team manager strategies Model - \bigcirc by the researcher

5.3.1 Transfer Strategy

In this research, transfer strategy is defined

"I use statistics to search for players. They allow me to value players in terms of number of matches played, number of scores, and overall performance records." W1FC8

According to this strategy, the optimal player is defined in terms on the specific objective, with variations in desired outcome informing the choice of the most appropriate methodology. There are two main objectives in player recruitment, namely to complete a Transfer for Investment or to Transfer for Match Results.

5.3.1.1 Value-based transfer process: Transfer for Investment (Money, Exchange)

In terms of Transfer for Investment, the optimal player should be chosen based on the differences between the actual market prices of players and the precise value of that particular player. The difference is the profit from buying players cheaply from a remote location and exposing him to the market through matches and games. Synthesis of interview data indicates that the steps were: determining the most appreciated KPIs in the market, locating a player who matches/exceeds these requirements, and identifying the highest possible price for that player.

"We already know that data driven approaches in sports can let you do a lot of important things such as finding undervalued talent to hire for the team or assessing player performance with things like GPS locators and heart rate monitors, and so on." W1FC9

"It also allows you to analyse gameplay by looking at the predictors of scoring or potential weaknesses in defence and while the first of these, finding undervalued talent, is typically

at the domain of the team manager, the other two along with many others are of direct concern to coaches." W2FM8.

In summary, the value-based process to transfer players is illustrated below (see Table 5-1). KD is an effective way to identify the most valuable KPIs in the market. Data analytics can correlate between the market prices and KPIs per position, helping the team to locate and purchase undervalued players. KD can then help the players offering the highest potential expected ROI to be identified, by identifying those players with the highest required KPIs at the lowest price in the market.

Table 5-1: Summary o	f value-based _l	process to	transfer pl	layers
----------------------	----------------------------	------------	-------------	--------

Step	Step	Evidence
1	Defining the most appreciated	"Each position has its main requirements and KPIs. Some of them
	KPIs in the market	are more appreciated than others depend on the position". W1FC9
2	Identifying the players who possess these KPIs	"So, you look at those data, you look at the technique of the players, how is their passing rate, do they lose the ball very quickly". <i>WITD</i>
3	Identifying the actual market prices	"Financial wise, full data about the player, you know the value of the players, salaries, and bounces performance wise, all of these measurements. So, these data help not to be subjective, not a point of view, data, facts." <i>W1DA4</i>
4	Identify the fair market price	"When you renew players contracts, know how much they worth in the market, plan B, ultimate players." <i>W1DA4</i> "All of that can be used to assess his value" <i>W1FC8</i>
5	Identify the gap	"determine the difference between the price and what should be" W1FC8
6	Buy the player	
7	Expose him to the market	

5.3.1.2 Value-based transfer process: Transfer for Match Results

In terms of Transfer for Match Results, the optimal player requires an in-depth understanding of the preferred strategy of the team manager, the required KPIs in the identified positions, the KPIs of the players currently holding positions, and the gap between the required and current KPIs. These metrics can help team managers to identify the best players to fulfil these requirements.

"The roles determine the rates, not the other way around. When you put "Coach 8" as a midfield defender, his passing rate will be 99% when you put his as 10 then it will be 60% and you don't say because "Coach 8" is 60% then we put him in 10! He had some skills which makes him for the 10 position, such as creativity, agility – these kinds of things. And based on his position, the rate starts to change. When you have a player at number 10, you don't look at his passing rate, but you look at how many assists, goals, etc. These are all that really counts. When you speak about a player at 6, you talk about passing rate being very important, so if he loses the ball many times, then every time counts. This is why I believe that the role determines the rates." W1TD

The process for transferring a player based on match strategy is as follows: the team manager identifies his preferred strategy, then he defines the targeted KPIs for each position, based on his preferred strategy. Once this has been completed, the current KPIs of the current players are assessed and insignificant differences dealt with by training and strategy. Otherwise, the player will be replaced in that position by a more suitable alternative.

The KD system can also help managers to identify the targeted KPIs relative to the indicators of other team players. For instance, defenders should run at a faster speed than the average speed of the attackers in the league. Midfielders should have a higher ability to run for long periods of time than the average of competitors. In this way, KD systems can identify the main KPIs required for different strategies, based on the KPIs in the current league. The benchmarking process can even be multi-dimensional, enabling coaching staff to optimise the players or their positional performance, thereby improving the overall performance of the team.

KD systems can also define 'significance' based on historical performance and patterns of improvement. Furthermore, KD can correlate body measures with the different KPIs in an attempt to identify the maximum potential of each player, even after they have received training.

The step	Evidence
Identifying the team manager	"So, it depends again in the game model that you are playing but to get something in general it is very difficult of course, Central Defenders are taller than your full backs. So these things are normal but in general no, it depends on your game model." WITD
preferred strategy	"Also depends what the coach goals and objectives from the players of the team and team needs. Coaches some time like fast players, strong player and so on. Depends on the playing style of the team manager.

Table 5-2: Value based transfer process: Transfer for match results

	Also, Defensive aggressiveness and offensive aggressiveness which can
	be analysed from videos and matches. Also, and this has to reflect each
	positions and role. WIDA3.
	<i>"Attacking player: (Assist and goals) - attacker - offensive = productivity</i>
	score. Winger; how much he can pass, one-against-one, to create
	overlaps so that's important. Midfield: there are two kinds of midfield
	player - defensive midfield player (6) and offensive midfield player (8).
	But it is difficult, everything depends on your playing model!" W1TD
	"On average wing back covers distance more than others. But even when
	you talk about high look at Barcelona FC, they are all small, look at
	Bayern FC they are all tall" W1TD
Identify the	"It depends if you are a wingon or contral striker look to "Andy Carol"
KPIs	he is 2 meters, but he cannot play as winger. So when you ask about it
required	again you cannot determine like this because you also you need to have
for a start	this, it's the size goal keeper need to be tall and fast CD: tall not fast
for each	side defence not tall you need to run striker should be tall winger
position	should be small but fast and based on this you can start to work "WITD
	"Attacking player: (Assist and goals) - attacker - offensive = productivity
	score" W1TD
Assess the	"I see some details about some player and later we tell them what they
current	did well or not. Also, I analyse in team level and we show the team what
players	they did good and wrong. All we provide feedback for the players and
KPIs	the team." WIDA3.
Identify the	
significance	"You look at fitness data (meters run over 24km/h this is very important
significanc	factor) if you don't have this kind of players because it is genetically it
e/insignific	is not something that you can train on you have physically fit player, so
ance of the	you do not have them. In this case you may need to substitute him"
gap	WITD
Substitute	"If you have two different players. One of them is faster than the other.
the current	The slower one has bigger body than the faster one. It is critical to
players	allocate players on the positions to fit the match strategy. If the ball is

with	expected to be most of the time in centre line, who shall be the defender
significant	and who shall be in the midfield? W1FC7
gaps by	
others	

5.3.2 Training Strategy

Training can be based match feedback or based on a trend in performance. This research argues for training to improve performance based on trends, rather than on feedback from a single match. One reason for this is that psychological factors can be an element in suddenly reducing the performance, so feedback from a single game may provide useful insights into psychological situations or for showing the level of progression in performance.

"Also, I see some details about some player and later we tell them what they did well or not. Also, I analyse in team level and we show the team what they did good and wrong. All we provide feedback for the players and the team." WIDA3.

Therefore, this research seeks to utilise the advice of coaches to create training programmes based on a clear profile created for each player that ensures that they meet the minimum standards required and that they are the best fit for the team.

"yes, for the forward player, because this kind of player make a lot of sprint all the time, and this one move like this (tic) so the player will make longer sprint than this one" W1FC2

- "Also, the data we have is from match to match, A, B, C nothing in between there should be something between these matches link the training sessions, between A to B there should be a link between them, B to C there should be a line as well and so forth. You could compensate these limited by other technologies in order to have a consistence flow of data." W1DA4.
- "All the information I get for the games for session training if I want improve, I would improve sprint – improve his sprints with respect to his positions" W1FC2

"I take all this information and make session I want train another day strength then another day the speed. So that what I see the fitness in football then another sport" W1FC2

If players demonstrate significant gaps between their own performance and the desired KPIs, then the team manager will buy new players. Otherwise, the head coach will train

these players to better fit their roles, with training primarily focusing on fitness and technical skills, like dribbling and interventions. However, there are no clear guidelines for when differences should be classified as significant. Although KD can support this decision-making process, ultimately the decision is dependent on the judgement of the managers.

"Software gives us different kind of information first they cover distance is different and the sprint are different also, so the session trainer has to specific for them I can't make a session training the same for the forward and midfield or the back player, so I need to make deference with the distance covering the sprint for every kind of information the software gives me". W1DA3.

- "The capacity to make a lot sprint and recover faster, when I say recovers faster is to make another fast sprint. For example, I make 20m high intensity 30km per hour that's kind of movement maybe about 20 sec to recover to make another one. That's important. Strength, to be strong all the time because in the forward he has to support, hold the defence". W1FC2
- "When we came here we take the software we analysis the game and the training session what I saw the player doesn't cover 76m per min they cover 52 or 56 less then hour in their area in specific game, I take my excurses in each session they cover 52m per m so I need to improve that I need my player cover over 70m/m over ok, now how many sprint they make per session training maybe 24 I need to improve it 30 up over 30 so that can of meaning is reality that kind of information I take I make new excurses to try make the stress for them like this what it means I need to make one exercise to make the player make over 30 sprint so with the staff a lot sprint with the ball definition cover cross, finishes, I don't know like real movement I say Ramon I need that kind of excurses to improve the sprint in the game but for improve we have to train it not just say you need to do it." W1FC2

5.3.3 Match Strategy

In this research, match strategy is the process of using a match model to understand the strength, weaknesses, opportunities and threats (SWOT) of opponents in the match environment, in order to help the team winning the match. Based on this definition, the first step in devising the strategy is to do a SWOT analysis and the second is to use that analysis to devise the match model.

"If I can understand my team KPIs very well, my training path will be right and fit with the match strategy. My tactics are developed based on KPIs and capacity and limitations of

the players. By setting the weaknesses and strength of each player for each position, I can set the plan and define tasks for each players". W1FC7

SWOT analysis is to identify the strengths, weaknesses, opportunities, and threats for the team to win the match. All of these aspects are relevant and not fixed. They are relevant to the opponent situations. Thus, it is important to identify, compare and analyse these dimensions for both teams, which can be developed through knowledge discovery systems. Indeed, it has been observed by most of the team manager starting point to formulate the match strategy is identifying the relative power and performance metrics of the competing team before anything else.

"The first step I start with is to identify the relative strength and weaknesses of the opponent team relative to my team. Measures I use are number of successful passing, speed of covering, empty spaces, number of winning matches, how it plays in its home, abroad, when it plays with big team, with small team, formulating its attack, strength and weakness points, and sources of strength and weaknesses points" W1FC8

"I start by identifying the main strength and weakness points of the opponent team relative to my team. I take care of detailed and general information about sources of strength point in the tactics of the opponent teams. I rely on the videos, statistics and other reports from my team. Believe me, if you missed any of these details you can lose the match easily". W1FC9

From the interviews, the most important dimensions can be classified into internal factors for identifying strength and weaknesses, and external factors with which to identify opportunities and threats. The internal factors can be classified into the performances of players in different positions in the team, as well as the overall level of team cohesion. This is important because different positions need different skills and abilities, which enable each position to fulfil certain functions in the match. These resources enable the team manager to select an optimal mix, based on the level of talent in the opposite team in each position.

"For example, you ask a Right Back (RB) that he cannot attack and only cover the Right back zone and he gave two assistances and score a goal maybe I'm not happy. This is taking in my mind, as I know the capacity of the opponent positions of the right back". W1FC2 This gives a unique importance of the use of KPIs to measure the relative endowment of the skills required in each position with taking into consideration the opponent facing positions to each of them. The second internal aspect is the team cohesions and the predicted consistency of the players' performance in the match. This is also not a fixed measure, this relevant measure as it requires to assess the passing rate with taking into consideration the opponent team ability to intervene the ball. Likewise, the ability to do forward play and possess the ball for period of time can be relied on the dribbling ability and speed of the players, in comparison to the opponent team position players' abilities. This gives another uniqueness of the importance to measure the team cohesion with taking into consideration the other team's ability to intervene and to stop the team possession of the ball in the match. Currently, there is no unified approach to do that. It is based on the experience and the coach team knowledge and views. This gives a clear importance to use KD capabilities to build certain heat dynamic maps for simulating different scenarios in different conditions, as will be discussed later.

"To know his team performance, only." W1DA4

"So, it depends on the personality of the Head Coach. For example, if we have similar to this information with Fabio we could perform much better. It is honestly depending on the coach and what information he wants to use to his advantage, and fully know how to use it and utilise it." WIDA4

"For Forward Players, I think is to know really the stress the player has. I mean maybe is important for me to know how many stress has the player after the game or after the session training for to know how many time for them to recover, I can give know chart for the player how he is stress to that, the software give a lot of information but maybe I need another support about it always there is room for improvement" W1FC2.

The heavy use of the data analytics makes the team as mechanics with high harmony and coherent performance. Because, as it is perceived by of the team managers, European leagues use the knowledge discovery and data analytic models extensively, there is a perception that it looks like "Chess" in how they formulate match strategy and how the match is played.

"I prefer watch a game between Brazil and Argentina in opposite of Europe match because they play like chess". W1FC9



Figure 5-4: The role of the Knowledge Discovery Indicators on the match analysis model - © by the researcher

5.4 The KD Outputs: Balanced Key Performance Indicators Roadmap

This section seeks to develop a road map to help team managers and data analysts to effectively process historical data to improve the effectiveness of team strategies and related policies.

"I believe anything can be measured. If it is measured, it can be theorised, valued and assessed. If you understand and measure how things work, you can control it. If you can control it, absolutely, you can win" W1FC7

There is an insightful argument that the best way to understand any KPI or any analysis is to understand the context, taking into consideration other KPIs that can contribute to, and enrich understanding of, the variation in match performance.

"For examples, you may have a player that have high accuracy of passes, for example %90 pass accuracy. Then when you look at the details of the players passes you see most of them to the side or to the back, it is different than a player who passes forward most of the time. You really need to have a lot of things when you one to consider some aspect of the analysis." W1DA3

The balanced scorecard of KPIs (Norton and Kaplan, 1998) is based on reading, analysing, and interpreting historical data and information based on a structured approach. Structured analysis can make KD activities more valuable by facilitating the use of Agile methods. In other words, structured analysis can help each step to be clearly formulated in each sprint, in order to improve match results by improving the

opportunities of scoring more goals and conceding fewer goals than the opposition teams, ultimately improving match performance as measured by the criterion of winning each game. These match performance indicators are constrained by other KPIs. Therefore, by building a map of indicators, the root reasons for key metrics can be investigated. In this study, the main factor cited by interviewees was tactical KPIs. These are influenced by the technical and fitness KPIs. Fitness KPIs influence technical KPIs. However, fitness KPIs are not sufficient to improve match performance without the addition of technical and tactical KPIs. Psychological KPIs influence all of these dimensions. The following sections provide an in-depth explanation of these relationships, using supporting evidence from the interviews.

"The most important KPIs are speed, tactics, technical, fitness. Sometimes you look at passing rate, the amount of recovery, distance run over 24 Km/h these kinds of thingsm, but it depends" W1TD.

"All of the KPIs are correlated and interdependent. Speed is required to create successful passing. Successful passing is required for penetrating opponents team lines. This, if utilised successfully, leads to scores. Thus, all KPIs are required and complement each other's" W1FC7



Figure 5-5: KPIs BSC - © by the researcher

5.4.1 Tactical KPIs

Tactical KPIs are metrics that are intended to measure the ability of players to position themselves effectively in such a way the probability of passing, possessing, scoring and intercepting are improved. Tactical KPIs are measured in terms of players, units of play (set of players), tactical lines (e.g. attacking, defending, or midfield line), or by the team. These KPIs are then classified into passes, possession, and playing style.

"Number of passes making significant attacks, number of constructive passes, number of goal keepers touches, making significant attacks, number of crosses and overs, all of these skills make much difference in the match results". W1FC8

"The most important KPIs for winning the match is the players' movements. How the player intervenes, receive and shoot the ball, which come from positioning in the right time and the right place during the match. All of that depends on the players' abilities in being able to see himself in the pitch" W1FC9

"Tactical skills are all about movements with ball and without the ball, and utilising spaces" WIFC7

Tactical skills are perhaps the most important skills for modern footballers, with most valuable activities in the match being connected to these skills, even leading to 'man of the match' decisions.

"Who is the best player? If you want to know who the best player is, it is not the one runs most, or the fastest. He is the one who has highest successful passes with least missing. The one who can receive the ball and send it back successfully. To identify the best player, you need to have KPIs and analytic system that can find him. You cannot find him by your eyes, since this can be illusion. Many things are taken into considerations such as number of constructive attacks, successful long balls, and initiating successful counter attacks" W1FC8.

"The player ability to play within a team is the key here. His ability to create new space, preparation for the successful attack, good possession of the ball, ability to intervene the ball and return it back to the attacker in short time. All of these skills make the team plays better and the team scoring abilities improve" W1FC7

It is difficult to measure the quality of movement or positioning performance in the match through quantitative indicators. Thus, other metrics key factors are used as proximities to this ability.

"I think that's most important, and the you look at their tactical skills, and that something you can measure (how many times they lose the ball, how they use their right/left foot, etc..)

and then you look at their tactical qualities. This how you determine a role in a team" WITD.

These key factors are the passing rate, scoring from assists, scoring from faults made by the opposition, interceptions, and stopping counter attacks quickly.

"So, these are the things I were looking at. Then there are of course some underlying data like how many passes that we gave, what was the success rate we knew that with "T10" if the passing rate was less than 80% then we are not doing good. But as I already said before if P3 successful passes rate was less than 95% then he played a very bad game because he was a link player" W1TD.

Tactical skills are important for all positions, but especially crucial for defenders to create successful counter attack, for the link player (e.g. midfielders) to build a successful attack, and for attackers to score.

- "The link player should not have less than 95% successful rate. And then you have "P6" (Number 10) he is the one who has to create chances, for example if he has more than 80% passes rate then for sure he is playing a bad game which means he is always looking for easy solutions, he should take risks, he is the genius, he has the creativity so he should take risk, he should look for possibilities." WITD.
- "Maybe the location the player has, when I saw in the game to P5 playing, P5 have different movement then P4 because P5 can play back to goalie but P4 need to play forward. That kind of movement take you give to take good decision. But I can imagine I need to P4 playing like P5 also. P5 playing like P4, but P5 has more tactic to play with the ball. W1FC2
- "So that means if you have 5 occasions and you score 2 or 3 goals and those are the most successful team. When you look at Chelsea FC or Leicester City FC this year they don't have many chances, but when they have a chance they score and that's very important." WITD.

Head coach job is to understand and to predict the players' capacities and position them of his team and the opponent teams.

"Tactical performance is based on head coach ability to absorb different players KPIs in his team and the opponent team to fit players in the right places" W1TD.

Effectively, this consideration describes the ability of the head coach to comprehend different KPIs and to situate the players in different circles or domains on the pitch, thereby improving the controllability of the match. In this way, following the directions of the head coach can enhance the tactical abilities and performance of the players, resulting in better overall results for the team as a whole.

"Player commitment is the key in tactical performance. I set circles to play for each player in such a way my tactics can win. Any leakage could have negative consequences. I wish if we can invent a technology to control players' minds to ensure that" WITD.

The reason for this position is that the tactical performance of players is arguably more a function of the team manager than anything else and should therefore be assessed more in terms of the team than in terms of individual players.

"Tactical performance is solely determined by the team manager capabilities to understand the match parameters and the players commitment to the training managers' directions." WITD

"the best player who understand and follow the train managers instruction with taking into consideration the match dynamics. this is very important to be able to see the opponent and his team movements. it is for me not a fixed calculated thing. it is a chemistry" W1FC9

However, others who believe that tactical performance can be assessed per player as well as per team face this argument

"It is important Information about the individual quality about the players of the team. It covers Tactical information about the team." W1DA4

"It is usually when he plays a good game it means that his successful passes rate was about 50%. Again, you have to look at player by player and that depends on the objective the head coach given" WITD.

Tactical performance also encapsulates the ability of a player to read the dynamic, fastmoving situation of a match, including the movements and intentions of players and their positions. Without predicting the opponents and the performance of their players, it is difficult for a player to demonstrate tactical performance in terms of constructive long passes or crosses or the creation of a successful attack utilising a certain pass at the right time to the right player. All of these KPIs do not only come from the tactical capabilities of individual players, but also from their fitness and technical capabilities, as the next quote highlights.

"Finally, to measure the player performance, we looked at occasions, occasions like "clear cut chances or clear chances" and approaches to the goal. Then you calculate them and of course the performance is related to how many occasions you can create but according to me the most successful team in the world is specially how many occasions do you convert in a goal and the most successful team are the ones with more than 40% conversion rate" WITD.

Sources of tactical performance information are the players' commitments to the team manager directions, in addition to their specific technical and tactical skills, as will be detailed in the following sections.

Table 5-3: Sample of noted Tactical KPIs from the interviews

Sample of Tactical KPIs			
Passes Indicators			
Overall all passes performance	% of successful passes per match (e.g. spontaneous passes, 1 to 1 passes, unit		
index	passes, constructive passes, and long passes)		
% of successful spontaneous	SP is defined as the passes without having a clear intention to build a		
passes	constructive attack (i.e. due to pressure from opponents)		
% of successful 1 to 1 pass	1 to 1 pass is the several passes between two players only aiming to construct an		
	attack, penetrate defensive line or shift the direction of the play		
% of successful unit passes	Unit passes are the several repeated passes between more than two players.		
% of successful constructive	Constructive passes are more than 2 passes with more than 2 players aiming to		
passes:	construct an attack or shift the play direction. (e.g. second ball and third ball)		
% of successful long passes	Long passes is a movement of the ball from a zone to another zone or from		
	tactical lines (e.g. from back to front, from the left side to the right side or from		
	defending to attacking line). E.g. successful crosses/ counter attack		
% of successful interceptions	tions from short passes or long passes (crosses or counter attacks)		
Team Playing Style			
Time player with/without ball	minutes played with or without the ball		
Time played per position	in offence, defence, and midfield		
Ball recovery time	Average time required to regain the ball		
Total Possession	% of the time the team hold the ball		
Distances between attackers and	<i>d</i> The average distance between the attacking and defending lines.		
defenders			
Maintain distance between	The % of the time that distance between players within the ball range is lower		
players (close down space):	than the coach defined space in each zone		
Offside (Tolerance)	(such as % of the successful deliberate offside (avoidance) and % of the		
Management	successful avoiding opponent deliberate offside)		
Player Positioning Performance			
Role rotations	% of successful (constructive change) changes of players' positions in the pitch		
	during the match.		
The duration of a player being in	The duration of a player being in specific zone.		
specific zone.			
Player density: % of time played	The position where the player(s) playing in during the match.		
in the player specified zone			
Marking (man to man marking –	The ability of a player to mark an opponent's players – or zonal area of the pitch		
Zonal marking) :			

5.4.2 Technical KPIs

In this research, technical skills are the different physical competences required by different positions in the game. It is the ability to control the ball for the sake of accomplishing the required tasks effectively and efficiently. It is defined in this research as different individual football physical competencies required to control or to regain the control, to direct the ball, and to build constructive movements during the match. They are classified into off the ball competences (ability to regain the control) and on the ball competences (ability to direct the ball towards a constructive movement). Examples of technical KPIs could be the ability to save the ball for a goalkeeper, interceptions for the defence players, or aerial skills for midfielders.

"It depends a positions by positions, you have goal keepers, defenders and so on." WITD.

The one who cover the most important spaces in the ground is the one who has the fitness and technical skills. They are most important to be able to intervene opponent passes.

"It is also a technical skill to be able to stop the ball in the right time in the right way" W1FC5.

- "You need to consider the position of the player, different aspect different characteristics and to see which player is best in which position. Also depends what the coach goals and objectives from the players of the team and team needs" WIDA3.
- "For each position, there are certain essential technical skills. For instance, flanks should be able to control the ball and able to dribble. Also, defenders shall be able to play with head, aerial skills, and tackle the ball. "W1FC7

Technical skills are essential compliments for tactical skills. Without good technical skills (e.g. dribbling, intervening), there will be less possibility to control the ball, giving fewer opportunities to score or contribute to goals.

"We can separate between the fitness and the tactics." W1FC2

However, these skills are not sufficient for tactical performance and match performance (scoring and guarding the goal).

"Two players are different in their body size with different KPI indicators. one of them is big but heavy and slow, while the other is thin and fast. Fitness indicators can be misleading here. The heavy person can be positioned in the right place so that he could score better as his shoot is strong. But he has weakness in the speed. The other person who is thin, he can run fast, but he does not have such ability to shoot as strong as the other but can score from penetrating in the opponent lines" W1FC7

Indicator		
Ball Regain	% Success rate of the attempts to regain control on the ball (e.g. tackles or	
	interceptions)	
Dribbling	% Success rate to take on (e.g. dribbling)	
Aerial Interaction	% the players ability to win aerial interactions	
Ball control with	Significant change in the speed with the ability to build a constructive pass	
speed	or goal. i.e. 80% change in speed within 1 min leading to successful attack.	
Free Kicks	% of successful shots towards the goal in different situations (in plenty area,	
	outside the plenty area, when marked by 1 person, by more than 1 person).	
Free Kicks Goals	% of successful free kicks towards the goal from different zones (e.g. right,	
	left, middle zones).	
Innovation index	Number of innovative movements in the match (new dribbling, tackles,	
	passes or movement)	

Table 5-4: Sample of noted technical indicators from the interviews

5.4.3 Physical KPIs

In this research, Physical KPIs are those physiological and fitness measures for the players' abilities. Some of them are traits that cannot be changed, such as the height and ambidexterity while others can be improved by training such as speed, high/moderate intensity running and recovery rate. These factors can be preliminary indicators for predicting match results

"As an example, normal team they run 800/900 meter in more than 24 km/h and the more successful teams they run 4/5 Km. That's already a very important parameter. "W1TD.

The most noted and observed by all participant as they are published online and relatively stable, although they can not be changed from match to match. W1DA3`

"I take all the information about my player how many km they covered in the session how many sprints made in the game how many crosses how many tools. So, I take all the information then I make my decision then I know because I read many papers in football strength sprint speed condition" W1FC2.

Some experts claim that these factors are relatively fixed for most teams, except the number of sprints and high intensity movement over distance, which means that these factors can help a team outperform their opponents.

"and when you speak about fitness it is specially about the distances they run more than 24 km/h so the total volume they run is not important at all because if you look at the last 20 years the total distances has not increased the only thing that was increased was the number of sprints and the distance in high intensity." W1TD.

Physiological measures can provide the basis for technical and tactical skills, as these skills are insufficient without them to get match results.

- Without the fitness aspects, talking about the technical and tactical aspect would make no sense. The ability to follow the ball depends on the ability to run easily and speedy, with taking the ability to observe others". W1FC9
- "The technical and tactical skills are based on body movements. If the player does not have the ability to do these movements fast with high level of concentration, these skills can be demonstrated in the match. To do good passes and receive balls, you have to run and move faster than your opponent. Once the fuel finishes, the team will be defeated easily. "W1FC2

Several factors were noted as being likely to improve the probability of better match results, in terms of scoring goals, or preventing opponents from scoring goals.

"but for me I knew that when we ran more than the opponents (24 km/h) the likelihood of us being more successful is higher, when we recover more balls in the final third, more occasions; specially the conversion rate." W1TD.

"The critical KPIs for me are Meters in high intensity, 24Km/h, amount of occasions, the conversion rate; how many occasions you score, %of occasions you score, recoveries in the final third of the field." W1TD.

The interviewees suggest that fitness is only a contributor to tactical performance but is insufficient on its own to score in the match.

"There is a confusion here. Player X can have good sprints but number of unsuccessful passes. is it a tactical issue or physical issue? I think relying only on distance run per match could be misleading as a proximity for the tactical skills." W1FC9

There is some evidence to suggest that physical KPIs and tactical KPIs are sometimes mixed, because there is a high level of association between the two factors, with fast and fit players having more likelihood to possess the ball and make successful shots on goal. Additionally, fitness metrics are published and accessible, while tactical are not easily quantified (relying on video analysis rather than statistics) and therefore much less frequently published.

"Here is the issue. There is confusion between tactical and fitness aspects. for example, if you find a certain player has not run much in the match, you can translate that as fitness

issue; but indeed, it can be tactical issue, and vice versa. To avoid such confusion, I prefer to use the computer for measuring the KM run in the match, but I prefer it based on the videos recorded. Statistics is difficult to help here". W1FC7

Physical Indicators	Descriptive
Player Speed Index	Average player speed per match in different modes (low, moderate, high speed and sprint) with and without the ball
Player speed in different running categories with the ball	In low, moderate, high speed, and sprint with the ball
Player speed in different running categories without the ball	In low, moderate, high speed, and sprint without the ball
Distance covered	The total distance run during the match
Distance covered with the ball	The total distance run during the match with Ball
Distance covered without the ball	The total distance run during the match without the Ball
Distance covered in different speed categories	In low, medium, high speed, and sprint
The maximum speed of shooting the ball	
The maximum distance of a throw in	
The maximum height of aerial action (i.e. jumping for header)	

5.4.4 Psychological KPIs

These indicators measure the volatility in other KPIs in different contexts. Psychological KPIs referees to the ability to play in the standard performance under different psychological pressures, which can be called "resilience indicator". As mentioned before, the physical performance of players is fixed and stable over time except some situational fluctuations. These fluctuations are reasoned to psychological issues.

"I can give you an example. One player has 99% success of the intervening opponent passes as long as within 5 meters. I can rely on him. But, from my experience, any sequential missing the balls in this area is an important flag for psychological issues. Once they are resolved, the player performance return to his normal rate" W1FC7

In this way, mental factors can affect all other measures in certain contexts.

"All of us has a fear of the psychological pressures. We have a serious fear of failure, which can influence our performance in a positive or negative way. But fear can increase the possibility of mistakes which can be translated into goals. Team manager shall be able to manage the level of psychological parameters such as stress and fear of players to overcome them" W1FC7

Environmental contexts can be influential in psychological factors and some individuals are psychologically affected by other players, which can affect their performance.

"It is crucial to understand the psychological factors affecting the players. Not even individual psychological issues but also psychological issues as a group. If the chemistry between players is not as should be, the performance will go seriously down" W2EM2

Personality is an important factor here, with players having different attitudes to risks and opportunities: some are risk averse and have a defensive style under pressure, while others are risk takers who adopt more aggressive response in pressure. These personal characteristics can influence players in different positions and at different times.

"Ibrahimović is egocentric but every striker in the world is egocentric. You cannot but an egocentric player as central defender because he will take risk just to show off, so you cannot do this." WITD.

Indicator	Description		
Resilience	% of change in the performance indicator in different contexts		
	(e.g. opponents, home/away, fan support, get paid well)		
Ethical indicators	number of cards or injuring other players		
Discipline indicators	% Body language/facial expressions of anger against the coach/referee decisions		
Manipulative indicators	Number of free kicks against opponent		

Table 5-6: Sample of noted psychological KPIs from the interviews

5.5 The KD Outcomes: Analytic Models

Analytic models are used to combine different indicators for revealing and discovering some of new knowledge. There are different and integrative analytic models are underlined in the interviews. They are predictive, simulations, context based, comparative and synergetic analysis. Predictive models are to predict the probability of scoring in a match for different models. Predictive models are shortcuts of the simulation models which aim to identify weaknesses and strengths of each match model before the match. Simulation models can use sophisticated applications or based on scenarios and comparative analysis. Context based analysis aim to understand the players' performance under different cases. Synergetic analysis is to find out the incremental performance of certain player when plays with a certain player, team or in a certain position. Indeed, all of these models complement each other, and it is argued to work together at the same time.

"The data analytic systems enable me to see the situation better. I can set certain scenarios, and hypotheses; then testing them through the data. This helps a lot in setting the match model to improve the possibilities of winning" W1DA3.

"Data modelling made the teams can read each other's before the match. it seems like a machine." W1FC9

5.5.1 Predictive and Simulation Models

Predictive models are equations or estimations used to estimate the probability of scoring in a variety of different situations. In effect, these models estimate the probability of delivering the objectives of the coach, in terms of scoring goals and obtaining a desired match result.

"If you play next game make sure that you play better because otherwise, I will be taking the decision next time to take you out of the field. So, he was using all of these data very well. His system was impressive. He uses this equation: occasions = goal attempt approaches to the goal. Let's take a look at this last one in gameplay. Now data science and gameplay are when you start trying to model what's happening on the field to find the predictors of desired outcomes" WITD.

There are two main outcomes for simulation analysis: predicting the outcome of games in different scenarios (i.e. match model), and identifying the potential strengths, opportunities, threats, and weaknesses for each scenario. This analysis can enable a team manager to train the players or change his match model to improve the possibility of obtaining the desired results.

Interviewees noted that simulation analysis can be achieved in a number of different ways. In fact, some did not even use any type of information technologies, using KPIs to mimic the match two days before it took place, which enabled the team manager to watch the simulated match, identify weaknesses and pressure points, and then ensure they are dealt with in the real match.

"In the last technical training, I ask the team to have a simulated match 48 hour before the real match. I ask the players to simulate the planned scenarios in fixed balls, building up

attacks, counter attacks, and corners. This can help the team and myself to identify the weaknesses in the expected match." W1FC7

This approach has many positive and weakness sides. The main advantage over the virtual one is it can be seen as training players on the expected match model. It can be expensive, costly, the simulated match may not fit with the real one, training coach may find this match model is not the best and needs to change it again. Thus, it is believed that the use of virtual simulation models can help to identify and anticipate the weaknesses, strength, opportunities, and threats before the match, or before the on-field-simulation. The main weaknesses of the virtual simulation that teams are not affordable or accessible to different team managers due to financial and people resources.

"In pioneering teams, like in Europe, they use virtual simulations. This is very beneficial for them to identify the weaknesses and strengths based on the opponent team. It is used in Real Madrid, Barcelona, and others. These tools can help them but needs people who can use it and extract the required knowledge from it. Also, they are expensive" W1FC9

5.5.2 Context based Modelling

The context-based modelling is defined in this research as the identification and measurement of the players' KPIs in different training and match context. Context here can be game location, quality of opposition, match status, and match timing (e.g. Match Half or extra time). The reason for contextual analysis is the performance of players can be sensitive to certain contexts which shall be considered in setting different scenarios or in simulations.

"We do different of series of comparisons through the application. Indeed, from my experience, the player performance changes from context to context. The KPIs are relevant to the context. By understanding these contextual factors, I can optimise the team composition KPIs to get the best. Also, all measures are identified with upper limit and lower limit to set for other unknown factors. This is the work of both train manager and data analyst." W1FC7

This modelling is not a one-time analysis, but rather it is based on a series of analyses to ensure that the measures used are reliable and valid.

"And then I'm interested to see if P3 is playing a good game or not. Then I go layer by layer. and see how many meters, where did he run his meters, was it the first half or the second half, what was his average position, what was his opponents' positions and go deeper and deeper and deeper. But when you ask me what should I do when I make my first analysis, which data I look at, these are the ones" WITD

According, to do that, the categorisation process is required to categorise the different states that players' performance can be sensitive to.

"He will study it and meeting with the staff, assistance coach, fitness coach, based on each game, match to match, categories of the match, so in Cat A match P3 performance in this way, in Cat B match he perform in this way and so on. Home or away, team performance will be low because of the "pressure" crowd, audience. He defines the categories based on previous data, experience and so on." W1DA4

5.5.3 Comparative Modelling

Comparative modelling is a statistical method for the comparison between players or teams utilising different KPIs. This form of analysis covers playing positions and competitive level to incorporate data from measures such as game result, tactical tisposition, or influence of fatigue.

"Also, specific data about each player performance. Also, opponent's analysis. Also, information about scouting new players." WIDA4

"Comparisons is a strategic tool to identify the weaknesses in the opponent team. This opens opportunity to explore other different models to get the best of the match." W1FC7

5.5.4 Synergetic Modelling

Synergetic modelling is a technique used to identify the correlation in a player performance with others in the team. It is called synergetic because synergy is defined as one plus one is more than two. I.e. collaboration of different actors could improve the outcomes if each of them is working alone. Football as a sport focuses on the collaboration of the players in the team. If there is not such collaboration between team members, the performance is expected to go down. The team members, if they work collaboratively, will be able to strength each other and improve their performance as one cohesive team.

"Players are not machine. The psychological and fitness are inseparable. Sometimes even you can see psychological relations between players. Some players' performance is highly correlated with others' performance. These factors are important, but I am not sure how knowledge discovery or data analysis can help here". W1FC7

"Fitness, technical skills and other abilities are important, but working as a team is a different story. As a team they can strengthen each other." W1P7

Several participants noted that player performance is affected by other players, rising or falling in response to the 'chemistry' between them.

"When X, Y, and Z players play together in any team, the performance of the match changes radically. All of the KPIs of the players such as running, passes rates, and scoring, change to better if they play together." W1FC7

It has not been noted any systematic approach to find such synergetic impacts. What is suggested in this research is to use correlational analysis. But what is noted by the interviewees is using the context-based analysis. I.e. the impacts on performance if some players missing and some players in the match.

"I tell you about one player called N, last year he was playing back to player L. If player L absent in any match, you can see player N performance goes down significantly. There is a chemistry between these players. But the same player does not have such chemistry with others in the team. This analysis if found by the context analysis when we noticed the player performance goes down without a clear reason." W1FC8

The drawback to this approach is the influence of chance, because it can be difficult to identify such factors on a certain player given that 10 scenarios for other 10 players will also be taken into consideration, which can make the analysis overwhelming. Thus, correlational analysis is recommended as a way to examine the different KPIs of players. Significant correlations can be signs of synergetic impact, based on the assumption that chemistry between the players with synergetic impact will have a consistent, measurable correlation as they influence each other.

"The Chemistry depends, on adding the weakness of the other" W1TD

The synergetic impact can be on tactical and physical KPIs, but do not typically support arguments for technical KPIs. Regarding the synergetic impact on tactical abilities, one interviewee said:

"Z" and "X", one is offensive, and one is defensive. IF you have both of them defensive then you will never have a goal chance, because the two players are too similar" W1TD

Whereas, the same participated said the following about synergetic impact on the technical abilities of players:

So "Z" defensively is not but "X" is good, and it depends on their foot as well, "P1" is left footed "P2" is right footed, it brings balance in the team. One is good in the air one is good in the ground." WITD

5.6 Knowledge Discovery Value Co-Creation Framework for Football Data Analytics (KDVCCFFDA)

This underpinning framework to understand the value creation from the adoption of the KD technologies is the resource based view, adopted from Melville and Kraemer (2004). Besides, the benefits map as a structural framework, adopted from Ward and Daniel (2012), to break down the value creation process into outputs, capabilities, outcomes, and benefits (i.e. value). Moreover, the value creation process is not waterfall due to the fuzziness nature of knowledge. Accordingly the Agile methodology is adopted (Beck et al., 2001; APM, 2015) to ensure the process is Agile and both parties (head coach and data analysts) are taking active role in discovering knowledge.

First, according to Melville framework, the resources are broken down into technological and complementary (or human resources). This research contributes to the knowledge by defining and taxomising the KD technological resources in the football industry. The Information Technology resources that aid in improving the understanding of the data available to the coaches and data analysts. These are Data capturing technologies, software applications for annotation and coding and lastly, databases, datasets and their interface to process and query the data available.

Also, this research contributes by detailing the human resources required for creating value from the KD technologies in the football industry. The Human Resources in this research are as followed. The Data Analyst Competences found in this research are

planning football, statistical, technological and communication. They are to clarify and improve the competences in which knowledge can be built, processed and used. The knowledge in football competence of the data analyst are there to aid in identifying the suitable and best technologies in capturing and analysing match events. Additionally, that as well help in improving the communication and collaboration between data analysts and coaches and improving in processing the tasks required. Thus, improving the analytical models that address the several issues that are required by the data analysts. The competences of the various technological resources available in football will aid the data analysts to investigate suitable data capturing technologies, look for specific and databases or datasets that support and optimise the data processing and analysis needs. Thus, improves the KD analytical models developed.

Next coaches' competences are found to be the statistical and software application competences. The coaching statistical competence in this research is to reflect on the trust, motivation and courage of the coach in using statistical data within the coaching team. It is very important for assisting in the data planning, proposing new statistical analysis and develop custom fit coaching models for the coaching team. Alongside of that is the coach capability of using different software application in his coaching activates and how that is reflect in his positive and negative coaching practices due to the awareness of limitation and allowances of software technologies. The outputs of using these resources effectively are discovering new KPIs and developing their analytical models (i.e. discovering relationships between their own sets of KPIs (e.g. relationships between physical and technical KPIs).

Because the knowledge is fuzzy in its nature, waterfall approach to discover knowledge is not a valid option. This research contributes to the Agile project management literature by underlining the role of the co-creation process in discovering new knowledge as project. The co-creation approach using Agile tools are found successful with the teams using them. Based on that, Value Co-creation tools are proposed. They are User Stories, Story Cards, Story Mapping Sprints, and Retrospectives.

The use of benefits map structural framework is found to be useful here in breaking down the value creation process into set of connected boxes as illustrated in Figure 5-6. The capabilities would be in the coaching team abilities in determining and understanding players' performance, identity the opponent's strength and weaknesses as well the opportunities and threats of next match. That as well with aid in understanding more about the opponent's teams' players' performance. The main outcomes that area highlighted by this research are the strategies in which the coaching team can reflect upon their practices. To elaborate more, developing effective training strategy are one result of the resources outputs and the coaching team capabilities. Based on the discovered KPIs and custom fit analytical models, the coaching team would be able to understand more about their training needs, area of weakness or strength, strength to utilise and optimise, weaknesses to avoid, work on and prepare for. By developing effective training strategy, match strategies can be improved. Having improved understanding about training needs and players capabilities (e.g. speed, sprints and passes and synergy between players) should aid in developing matches strategies. Also, reflecting in match training scenarios should reflect on improved match strategies. As a result of improved training and match strategies, understanding players and teams needs would be improved. Hence, assisting in developing effective recruitment and transfer strategies based on better understanding and reading of training and match strategies.



Figure 5-6: KD Value Co-creation Framework for football Data Analytics - © by the researcher

5.7 Knowledge Discovery Maturity Model for Football Data Analytics (KDMMFDA)

Knowledge Discovery Maturity Model for Football Analytics (**KDMMFA**) is developed to summarise the research findings and provide a useful tool for teams to assess their maturity in adopting knowledge discovery in their policies. The maturity model consists of four levels: Ad hoc, defined, managed, and optimised as will be detailed next.

5.7.1 Ad hoc

The first level, the ad-hoc, is unawareness or weak awareness of the KPIs, how to use them and how to put them into meaningful meaning. This level shows the coach is not interested and he does not care to consider KPIs in his training or match strategies. I.e. he might use videos and relying on qualitative analysis and experience to formulate strategies. This level assumes there is no existence of a data analyst. If existed, he has a minimal role in developing training and match strategies.

5.7.1.1 IT Resources

This level assumes that the team do not use or minimal use of technologies to capture the players' KPIs and so no use of any software for discovering correlations or classification (e.g. clustering analysis) analysis. I.e. to operationalise the Ad Hoc IT resources levels, the criteria are

- No existence of any technology that could code the players performance
- No existence of any technology that could track the players' performance (e.g. cameras)
- No existence of any technology that could do analysis or correlations
- No existence of database for the players performance

5.7.1.2 Human Resources

Human resources at the ad hoc level are minimal in terms of the required knowledge discovery competences for both

- No data analyst employed
- If a data analyst is employed, then is not aware of statistical tools
- Data analyst is not aware of the software applications that could help in analysis
- Data analyst is not aware of the football terms and concepts
- Data analyst communication skills are limited or weak

- Coach not aware of the value of knowledge discovery in the planning process
- Coach does not have specific attitude towards the use of data analytics in improving his strategies
- Coach is not aware how KPIs and use of KPIs could improve the planning process.
- Coach is not aware of software applications that could help in discovering knowledge in football.

5.7.1.3 Agile Based Practices Resources

Agile based practices resources are assumed to be minimal here with few use of them and with weak evidence of collaboration, communication, and/or trust between knowledge discovery actors (i.e. data analysts and coach). The items to be recorded as weakness at this level are:

- i. No use of stories in articulating problems
- ii. No clear vision of the full insight needed in the analysis
- iii. Lack of ability to of comprehensive analysis (partial task not the whole Inconsistency in the analysis outcomes
- iv. The tasks are not clearly identified for data analyst
- v. No clear timing of the meetings and production of reports
- vi. No lessons learned from the analysis process is documented
- vii. Lack of communication between the teams and/or stakeholders.
- viii. Limited contributions/collaboration by the stakeholders for developing the questions

5.7.2 Defined

In the "Defined" level, the team manager is aware of the different KPIs, analytics models, and how to use their results in the strategy development. This case could happen for new coach coming from abroad, but the club does not have sufficient resources to fund for the data analyst or technologies to have knowledge discover IT infrastructure. This is common in Olympic league and League 2.

5.7.2.1 IT Resources

In the defined level, the minimum IT resources required for getting players' performance are identified and used. At this level, the secondary data is only available from the third party but without access to any technology to track player performance.

- The use of secondary database provided from third party.
- No existence of any technology that could code the players performance.
- No existence of any devices that could track the players' performance (e.g. cameras/GPS).
- No existence of any technology that could do analysis or correlations.

5.7.2.2 Human Resources

In the defined level of human resources, the data analyst is available but with minimum knowledge of the statistics in terms of measuring the KPIs but not necessary being able to find correlational or clustering analysis. Also, at the defined level human resource manager knowledge in football is basic with basic communication skills. At this level of coach is aware of the importance of using statistics without having in-depth knowledge of statistics or use of the statistics in planning.

- There is a data analyst but with minimal understanding of statistic terms (i.e. limited to comparison in figures).
- The role of data analyst is limited to importing data from websites to write a report for the team manager.
- Knowledge of football is limited to reports produced.
- Communication skills are adequate. The communication is one way.
- Coach is aware of the availability of different tools and applications that could help in the knowledge discovery.
- Coach is aware that there are statistical models that can be used to improve the planning.

5.7.2.3 Agile Based Practices Resources

Coach and data analyst are using basic agile methods to improve the constructive communication between members. The basic methods here is operationalised in terms having scheduled meetings with defined acceptance criteria for the reports and outputs. The defined level is operationalised into

- The coach and stakeholders having well established Insights and vision regarding the analysis needs
- Meetings are done according to the sprints scheduling
- Defined acceptance criteria for each sprint regarding the discovery
- Encourage the knowledge discovery process by having set of stories/questions to work on in order to reach insights and vision of each sprints.

5.7.3 Managed

In the managed level, the team manager knows and utilises the basic features of data analytics. There are some coding applications that could quantify and measure the players performance such as SportsCode. Basic analyses are done such as comparisons between players. The focus is mainly on the physical KPIs because technical KPIs are more difficult to be captured through basic tools. Correlational, multivariate or other knowledge discovery analytic models are missing. The knowledge discovery practices are limited to what can be known from comparison of the physical KPIs.

5.7.3.1 IT Resources

At the managed level, the team has access to different tracking technologies besides the availability of the secondary data. Besides the use of coding applications to track and monitor different events in the match. But at this level there is access to advanced analytic applications

- Data analyst measures and uses coding application to get primary data of the players' performance
- GPS and other tracking devices are used
- No advanced analytic software application

5.7.3.2 Human Resources

The data analyst at the managed level are able to use tracking technologies and use of excel in sorting, coding, saving, processing, and retrieving basic KPIs. The level of statistics use is average, mode, and median with no use of variations analysis. Also, the data analyst could have a constructive communication with the coach since the knowledge in the football is profound to absorb the match strategies. Also, the communication is improved because the coach have accepted level of knowledge in understanding the

capabilities of coding application and awareness of the possibilities of using KPIs statistics in creating new match and training strategies.

- Team of data analysts are coding the match performance and proposed opponent teams
- Use Excel in comparing players and teams
- Ability to use statistical modelling is limited
- communication is two-way
- There is awareness of the use of advanced statistical models.
- Coach knows and uses the possible features of the coding applications
- Coach is interested in different tools and applications that could help in the knowledge discovery.
- Coach is interested in statistical models that can be used to improve the planning.

5.7.3.3 Agile Based Practices Resources

The communication is more structured so that the lessons learned can be documented constructively. The use of story cards, sprints and other agile techniques but the knowledge management in documenting and using the communications are not optimised and not used effectively.

- Set up a tool for manging the KD process in terms of *problem identification*, *user stories/questions generation*,
- Having and working on clear *research* framework to reach the vison and insights of each sprints.
- Enforce learning from the reached results
- Enforce modelling practices addressing the similarities in the analytical questions/stories in order for enhancement
- Set up a clear KD analysis process based on the above and reflecting on them at the end of each sprints.

5.7.4 Optimised

The coach and data analyst are using advanced statistical models to do data mining and exploring new patterns in the data. This case can happen if the data analyst is well trained on advanced analytic models that can do contextual, predictive and simulations. These requires existence of advanced specialised analytic software applications. Also, it requires strong communications between the data analyst and beneficiaries of the knowledge.

5.7.4.1 IT Resources

IT resources at the optimized level refers to the using the full potentials of the technological resources available for the knowledge discovery. This does not only include data capturing technologies but also data analytics technologies. I.e. the IT at the optimised level is operationalised into the following statements.

- Besides the data coding system, there is a software that can correlate and differentiate concepts under investigation
- Besides the tracking technologies, body sensors are used
- Database is well structured, and data is clean
- database includes the performance of the other teams in the league with the same level of details of the team

5.7.4.2 Human Resources

Data analyst and coach competences in the knowledge discovery at the optimised level is the highest. At the optimised level, the communication is highly structured and constructive. The competences of data analyst to inform new data lead strategies is profound. Also, the coach is able to translate the problems into clear stories can be translated into set of clear hypothesis or data inquiries. The data analyst has high level of competences on using analytic applications and artificial intelligence applications. Also, the coach has high data curiosity to understand planning and training problems better.

- Data analyst is familiar and able to use the data analytic applications (Such as R and MATLAB)
- Data analyst have good knowledge of different analytic models (such as Artificial Intelligence Algorithms, Multivariate analysis, and structured modelling)
- Data Analysis has profound experience in football strategies (Training, transferring, and match strategies)
- Communication is solid and constructive between team manager and data analyst
- Coach has good knowledge and understanding of the statistical concepts and models (e.g. artificial intelligence algorithms)
- Coach has good understanding of how different statistical applications and models can help in improving his strategy
- Coach is a fan of data in planning

5.7.4.3 Agile Based Practices Resources

Agile based practices at the optimised level aim to fully adopt Agile methods and techniques to improve the quality of the constructive communications that can lead to discovering new useful and insightful knowledge. The use of retrospective and lessons learned is competent enough to allow effective continuous improvement in the value co-creation process through constructive communications.

- Monitor and measure the KD analysis process quality (against, user questions, stories, stories/question mapping, acceptance criteria)
- Enforce documentations, recording, and tracking process, model development in order to develop and tell informative and relative analysis stories
- Enforce retrospective meetings to improve and issues in current sprint and improve future sprints.

The next section will present the knowledge discovery maturity model for football analytics in a table format for the ease of visualisation and use.
5.8 KDMMFDA – Table Overview

This section represents the KDMM for Football Analytics in a table formats for improving the visualisation of it. Please see table *Table 5-7: KDMMFDA - © by the researcher*

Ad hoc		Defined	Managed	Optimised			
<i>HR</i> <i>Resources</i>	 No data analyst employed Data analyst is not aware of statistical tools Data analyst is not aware of the software applications that could help in analysis Data analyst is not aware of the football terms and concepts Data analyst communication skills are weak coach Manager not aware of the value of knowledge discovery in the planning processs coach manager does not have specific attitude towards the use of data analytics in improving his strategies 	 Team of data analysts are coding the match performance and proposed opponent teams Use Excel in comparing players and teams Ability to use statistical modelling is limited communication is two-way team manager knows and uses the possible features of the coding application There is awareness of the use of advanced statistical models. 	 Team of data analysts are coding the match performance and proposed opponent teams Use Excel in comparing players and teams Ability to use statistical modelling is limited communication is two-way team manager knows and uses the possible features of the coding application there is awareness of the use of advanced statistical models. 	 Data analyst is familiar and able to use the data analytic applications (Such as R and MATLAB) Data analyst have good knowledge of different analytic models (such as Artificial Intelligence Algorithms, Multivariate analysis, and structured modelling) Data Analysis has profound experience in football strategies (Training, transferring, and match strategies) Communication is solid and 			

•	Coach manager is not
	aware how KPIs and use
	of KPIs could improve
	the planning process.

• Coach manager is not aware of software applications that could help in discovering knowledge in the football.

IT Resources

- No existence of any technology that could code the players performance
- No existence of any technology that could track the players' performance (e.g. cameras)
- No existence of any technology that could do analysis or correlations

- The use of secondary database provided from third party
- No existence of any technology that could code the players performance
- No existence of any devices that could track the players' performance (e.g. cameras/GPS)
- Data analyst measures and uses coding application to get primary data of the players' performance
- GPS and other tracking devices are used
- No advanced analytic software application

constructive between team manager and data analyst

- Team manager has good knowledge and understanding of the statistical concepts and models
- Team manager has good understanding of how different statistical applications and models can help in improving his strategy
- Besides the data coding system, there is a software that can correlates and differentiate concepts under investigation
- Besides the tracking technologies, body sensors are used
- Database is well structured, and data is clean

- Value Co-Creation Activities
- No existence of database for the players performance
- No use of stories in articulating problems
- No clear vision of the full insight needed in the analysis
- Lack of ability to of comprehensive analysis (partial task not the whole Inconsistency in the analysis outcomes
- The tasks are not clearly identified for data analyst
- No clear timing of the meetings and production of reports
- No lessons learned from the analysis process is documented
- Lack of communication between the teams and/or stakeholders.
- Limited contributions/collaborati on by the stakeholders for developing the questions

- No existence of any technology that could do analysis or correlations.
- The team and stakeholders having well established Insights and vision regarding the analysis needs
- Meetings are done according to the sprints scheduling
- Defined acceptance criteria for each sprint regarding the discovery
- Encourage the knowledge discovery process by having set of stories/questions to work on in order to reach insights and vision of each sprints.

- Set up a tool for manging the KD process in terms of *problem identification, user stories/questions generation,*
- Having and working on clear research framework to reach the vison and insights of each sprints.
- Enforce learning from the reached *results* from the previous steps i and ii and reflect of that based on the acceptance criteria and the intended vision and insights.
- Enforce modelling practices addressing the

- database includes the performance of the other teams in the league with the same
- Monitor and measure the KD analysis process quality (against, user questions, stories, stories/question mapping, acceptance criteria)
- Enforce documentations, recording, and tracking process, model development in order to develop and tell informative and relative analysis stories
- Enforce retrospective meetings to improve and issues in current sprint and improve future sprints.

Analytics Practices	 Team Manager does not worry much about the KPIs Strategies are developed based on videos Sloley Team manager relies on his coaching team opinions in developing strategies 	 Team manager is aware of the KPIs, but he cannot utilise them Team manager is aware of different analytic models, but he is limited in the ability to utilise them Team manager uses secondary data reports for planning 	 similarities in the analytical questions/stories in order for enhancement Set up a clear KD analysis process based on the above and reflecting on them at the end of each sprints. The comparative models (such as t test, u test) are used to compare teams, and players The use of physical and technical KPIs are less there is no use of advanced analytic models 	 Use of contextual analysis use of predictive analysis use simulations models use most of KPIs use advanced structured KPIs relations such as Structured Equation Modelling
------------------------	--	--	---	--

5.9 Summary

The aim of this chapter was to investigate the potential value from adopting knowledge discovery technologies in teams and to develop a maturity model for realising this potential value. The knowledge discovery framework is developed and operationalised into maturity model in this chapter. To realise the knowledge discovery potential value, it is required to have the technological and human resources. When technological and human resources are orchestrated effectively using agile approach, new KPIs and analytic models can be developed which should inform useful and insightful planning and training practices so that the match results are improved. The maturity model assesses the existence of technological, competences of human resources and the adoption of agile based practices so that new KPIs and new analytic models can be developed to create useful knowledge for the coach.

Chapter 6 Knowledge Discovery Maturity Model – Analysis, Evaluation and Verification

6.1 Introduction

The aim of this chapter is to verify and validate the maturity model developed in the previous chapter. This validation and verification will also improve the validity of the framework as a whole. The chapter begins by describing the methodology used for the evaluation process, after which a background of the cases is introduced. The validation and verification process first focus on the maturity model, and then reflects on the use of the KPIs and analytic models in the development of strategies. Finally, this chapter presents the landmark feedback obtained from the participants in the current study.



Figure 6-1: Chapter Structure

6.2 Chapter Methodology

Validation and verification are considerations involved in ensuring the quality of the research. Validation serves to ensure that experts agree with the framework and tools

developed in this research; verification pertains to the use of the maturity model that operationalises the research findings and then a discussion of the results with cases. Validation involved face-to-face interviews conducted with nine experts, who gave opinions and insights on the framework, models, and tools. The details of these experts are in Chapter 3. Their insights into the research elements were integrated to improve the weaknesses identified. Verification was conducted through five case studies (five football teams in Saudi Arabia).

6.3 Cases Background

Five teams were selected for involvement in this phase of the research. These teams were different from cases used in developing the maturity model and had different resources available for KD processes. The selection process sought to ensure that different teams with different resources were selected, to ensure that the verification would be meaningful. Two national teams are selected and three teams from the league (two of which held top records in the professional league and the third is an average team). One or two interviews were held with members of each team, in order to gain a deeper understanding and to ensure the validity of the responses for the model questions.

The maturity model was operationalised into a set of questions with a score of five on a Likert scale (from strongly agree to strongly disagree). The questionnaire is in appendix A. The interviews were carried out face-to-face and took an average of 90 minutes. The results of the assessment were discussed back with the respondents, in order to obtain their feedback, as defined in the verification process.

All teams' data, backgrounds and records were obtained from the official Saudi Football Federation website: <u>www.saff.com.sa</u>, Saudi Professional League website: <u>www.spl.com.sa</u>, and from the official broadcasting sponsor: <u>www.dawriplus.com</u>. The main reason for the limited background introduction of the team is to maintain the ethical anonymity practice and policies of the research. It is just to highlight the major achievements of the teams and their contributions. The summary of the findings of the teams' assessments is in Table 6-1.

Case	Team	Main strength points	Main Weaknesses
Case1	CT1	Resources – Data analyst background is professional football coaching. Existence of secondary data sets and use of video.	Data analyst is called video analyst. There is no any role/position for analysing performance data.
Case 2	CT2	Resources Bachelor's Degree in Computer Science Training in football analytics and Coaching.	Use of advanced analytic models
Case 3	CT3	Resources Master's Degree in football performance analysis (Portugal) Coach Experience in the SPL	Communications/ use of advanced analytic models
Case 4	CT4	Serbian Coach – Passionate of data analyst and know about statistics, the use of analytic software Existence of Electronic Performance Tracking System (EPTS)	Agile – communication
Case 5	CT5	Team manager strong attitude to use statistical. Great coaching experience	Low resources. No dedicated full-time data analyst.

Table 6-1: Summary of Cases Strength vs Weaknesses points

6.3.1 Case 1: CT1 – Background

CT1 Saudi Football Club is a Saudi Arabian professional football club based in the western region. It competes in the Saudi Professional League (SPL) and is one of the top teams in Saudi Arabian football.

Domestically, CT1 have won many titles including the professional league, King Cups, 6 Crown Prince Cups, Prince Faisal bin Fahd Cups and Super Cup. In terms of international club football, CT1 were the winter of several GCC Champions League titles and an Arab Club Championship, as well as reaching two AFC Champions League finals. Along with top Saudi Arabian football teams, CT1 is one of the four founding members of the SPL and have never been relegated from the top flight.

Team manager / Head coach

Since 2014, CT1 have great coaching teams that aid and lead to significant performance improvement. Coaches from different countries such as Switzerland, Portugal, Ukraine and Argentina are the latest coaching expertise joined this team.

Data analyst

The club has one data analyst who does video analysis and develops reports from STATS (formally ProZone) data about the players, lines, team performance and evaluates competitors. The background of the analyst was professional football coaching and training courses in football performance analysis.

Technologies available

The technologies available were video clips, and primary and secondary datasets. There was no clear evidence that the analyst used training datasets collected using EPTS resources or used training data produced in the report.

Models used

Descriptive analysis (i.e. speed, distance, % of success)

6.3.2 Case 2: CT2 – Background

In 2016, this team qualified for the final against Japan, the winner of 2016 U19 Asian Federation Championship (AFC). Proceeding to the 2018 AFC U19, T2 have not lost any game in Qualifiers Group D. CT2 qualified to the Group Stage and will compete against, Malaysia, China P.R. and Tajikistan in October 2018.

Head coach / Team Manager

The head coach of CT2 is a Saudi Arabian coach.

Data Analyst

One full-time data analyst/video analyst (term used interchangeably). This expert has a bachelor's Degree in Computer Science. Based on the interview with him, he started his analysis in football as a fan and developed by taking football analytics courses and football coaching training. The coaching team feel that the data analyst understands the training and coaching strategies and that he is delivering the required reports. The analyst understands that there are differences in statistical interpretations of the data and tries to develop these in a meaningful way.

Technologies

The coaching team uses video clips, EPTS (polar belts and watch) and primary and secondary data.

Value co-creation aspects

There is a good culture within the coaching team, which was evident from the interviews, questions, and overall performance of the team. The communication, meetings and trust within the coaching team seems to have developed positive collaboration towards coaching and analysis queries. The team manager, coach and data analyst have worked together for more than a year.

Models used

Descriptive analysis

6.3.3 Case 3: CT3 – Background

CT3 has not qualified for the Olympics since 2000, however they were the Champions of the Arabian Gulf Cup in 2008. In 2017, the CT3 performed well in the qualifiers and qualified to the Group Stage. In November 2017, a new Argentinian coach was assigned

to the team, but the team did not qualify from the Group Stage. The current coach nationality is Saudi Arabian and is a successful head coach of a professional Saudi Arabian football team.

Head coach / Team Manager

The current coach of CT3, led the one his football team to the top four teams in the Saudi Arabian Professional League (SPL), which is a significant improvement from their poor performance from 2013. As a consequence, he was offered the role of head coach of the CT3 after last season. His coaching team are formed of 7 members: the team manager, 2 assistant coaches, a manager, a physical coach, a goal keeper coach, and one data/video analyst.

Data Analyst

The video analyst of the CT3 have master's qualifications in football analysis. The current resources used now are video clips, primary and secondary data.

Technologies

Video clips, and primary and secondary data. There is no use of purchased primary and secondary data.

Models used

Descriptive, and contextual analysis model is developed.

6.3.4 Case 4: CT4 – Background

After qualifying for the SPL, the management of CT4 invested heavily in developing and improving the coaching culture within the team. At an Olympic level, the CT4 Olympic team won the Saudi Arabian Olympic League in 2018.

Head coach / Team Manager

Great coaches have joined CT4 and it tends to develop good and strong team outcomes. One of their coaches was one of the great coaches that performed very well in the FIFA World Cup 2018. The CT4 team seems to seek specific coaching competences that fits their football team strategies from their approach of selecting coaches.

Data Analyst

The team data analyst in seems to perform the duties expected by the coaching team, including statistical interoperations of players, lines and overall team performance. Technologies The resources used for players team performance analysis are video analysis, primary and secondary data. Additional analysis using sophisticated tools were stated but not clarified. There is no clear evidence that there are use of various Electronic Performance Tracking System (EPTS) devices.

Models used

The mostly developed analytical models were comparative and descriptive models, with an unusual focus on contextual analysis.

6.3.5 Case 5: CT5– Background

CT5 is one of the top classic football teams Saudi Arabian. It has been one of the top 5 teams in the SPL during the last 10 years. The team competes across all different league competitions, maintaining a top ranking in the league, and are recognised for their excellent football academy.

Head coach / Team Manager

For the most of his professional career, the coach, was a former player of one of the top Saudi Arabian football clubs. He resigned as player in 2014 and started as a coach while undertaking his professional training. In 2016, he was recognised as one of top three young coaches in Saudi Arabia. He is one of few Saudi Arabian coaches to holds an Asian Federation Professional Coaching License, in recognition of the national shift towards the development of professional coaches to improve coaching and training standards.

Data Analyst

No dedicated performance analysts, but one is available part-time. That may have been influenced the collaboration between the coach and the analyst.

Technologies

Basic technological resources, such as video analysis from videos and observation.

Models used

The focus in the analysis model tends to be comparative, descriptive and simple statistics.

6.4 Maturity Model Validation and Verification

This section is structured according to the elements of the framework. The model consists of four main elements: data analyst competences, team manager competences, value cocreation process, and technological resources.

6.4.1 Data Analyst Competences

The key competences for data analysts are technology, statistics, football planning, and communications. The validation sought to ensure expert agreement with these elements. This followed by a verification through applying the tool on five cases.

6.4.1.1 Validating the Human Resources Required

There was acceptance of the model and results, with some comments supporting this position. All participants perceived football knowledge to be the most important factor, given its importance for discovering new knowledge from data, presenting the analysis in a useful and easy to understand way, and effectively understanding the needs of the coach.

"I faced a problem with the data analyst in the youth team. Frankly speaking, I failed to work with him. As you have shown me, the data analyst needs the sense, the feeling and knowledge of football. Some analysts have strong knowledge about statistics and analysis, but no idea about football. I can confirm what you have said. I sometimes think that analysts should have at least minimal experience in coaching or has played football for a while. This can help building ideas and giving a better understanding." W2FC4

"It is crucial that the data analyst be able to understand the coach approach and his style so that can give insightful recommendations based on the data. I struggled a lot with my previous data analyst who were lacking sufficient football knowledge" W2EM2

As mentioned by other experts, it is critical that mutual understanding and communication are constructed between stakeholders.

"There has to be clear understanding in how you see the game. This only comes with time and by working together for a long time. The analyst understands when an attack starts and when it finishes – does it start in a building attack phase – does it start in creating an attack phase – so you need to have a clear understanding. If you have that, you can go a long way." W2FTM8

One of the participants addressed the effort and time spent to convey message to those data analysts who does not have sufficient experience in coaching and sports.

"I can give you an example. I told my analyst to analyse a certain corner in different scenarios for the defenders. He did not give me the full picture. He did not give me what I expected. The communication was not great because of his lack of knowledge in football. Sometimes I spent more than 2 hours to convey my requirements to him. It was a waste of time." W2FC4 Football knowledge was a key factor in improving the ability of analysts to augment the work done by coaches.

"Our Analyst who have worked with in the world cup is a very intelligent football guy who knew "what is what", and what I was looking for - so when he came to say I asked I want to know how many times our central midfielders got into the right – he knew exactly what I meant by that." W2FTM8

"The analysts are doing a lot of statistics on, basically, how many transitions from attacking to defending and from defending to attacking but again there should be a clear understanding between the coach is when does it starts and when does it finish – it's just had a clear and good understanding between the two people." W2FTM8

Another perspective here is the diversity of the data analysts and coaches that are working together to collaborate in improving the KD process. W2FM8 stated that his team has two different coaches; one for in possession and one for out of possession. Both coaches work closely with the analyst to explore issues and formulate team strategy for different positions.

"The analyst needs to understand how the coaches work. If you got a head coach, an assistant coach, and whatever coach – we have three coaches, a head coach, in possession coach, out of possession coach and we have to be in one room together for all three of us to come up with the right sort of information, because my view as head coach is different than the in-possession and out of possession coach – so we got to have a joint thinking between everybody and it takes time." W2FM8

6.4.1.2 Verification of the Data Analyst Competences

According to the proposed framework, there are four competences required for effective KD processes: technology, statistics, football planning, and communications. All participants from the five cases agreed that these competences are the most required for the data analyst, evaluating them five out of five (with the average score for each competence/per team is plotted on a radar chart). Nevertheless, certain differences were identified, with the highest average being case 3 and the lowest case 5.

"It's important for the team analyst to know his role and boundaries like any other employee, and what data is required by the Team Manager/Head Coach. Knowing that it makes his/her job easier and more efficient, helping them to build a connection and trust with the Team Manager/Head Coach." CIDA1 "It's important for a team analysis to know what data he is required and what software/tools that can provide him/her with the information and not go wasting his/her time in looking in other data sources and is not relevant." C1DA1

"The data analyst in our team is useful and productive" C2AC1

Case 5 has a relatively low match performance. This case scored lower on all aspects than the other teams, perhaps because of a lack of resources and a part-time analyst with limited access to technologies. In contrast, the data analyst in case 3 has a master's degree in data analysis and access to specialised software applications. The data analyst is full dedicated to the team.

Case 1 and 2 are interesting because they demonstrate high technology competences, but relatively low average statistical competences. This may be because the data analysts are highly technology led, which gives access to descriptive statistics, but without training in sophisticated data analytics courses that have provided advanced statistical modelling techniques. Indeed, the data analyst in case 1 is called 'video analyst', which reflects that his duties involve analysing videos without focusing in data modelling for technologies. The budget of these two teams are much higher from others, enabling them to outperform others in buying new technologies. These teams have good results in using different KPIs, but not the highest, perhaps due to insufficient statistical competence. This performance could be due to the communication, football and technological competences of the data analyst. The participants agreed with this analysis.



Figure 6-2: Data Analysis Competences

6.4.2 Coach Competences

Two coach competences proposed in this research as requirements for knowledge discovery. They are awareness of benefits of using data in planning (i.e. how to use KPIs in planning) and attitude towards using these KPIs in planning.

6.4.2.1 Validation of Coach Competences

Experts agreed on these two factors. The main confirmed points are the awareness and knowledge of the use and the usefulness of knowledge discovery technologies.

- "Yes, I completely agree with you! I have seen many coaches who do not have any idea about the data analytics that could help them in managing and discovering knowledge from team performances. They got nothing from such technology because they do not know what they can get from them." W2EM2
- "Currently, what you need is awareness. Only 5 or 6 teams are using data analytics in the football. This is even very recently starting from two years ago only. In this year, they become 8 teams only using data analytics. Out of 170 teams, 162 team do not have data analysts and there is no use of any analytic tools. Do you want the reason? It is not because

of the budget; but there is lack of awareness of the usefulness of the power of data analytics. we need to spread such culture of using data analytics in football sports" W2TCM1

6.4.2.2 Verification of Coach Competences

In this research, two factors are proposed as being critical for team managers in order for them to realise the value from the knowledge discovery: attitude towards statistics and use of statistics in the analysis, and the knowledge and awareness of different analytic approaches. All cases agreed that these factors are vital, scoring five out of five, for translating the data analyst reports and analysis into strategies and performance. Regarding the current self-assessment, team managers for case 1 and case 3 have the most passion for the use of data in analysis, although their knowledge and awareness are less than case 4. Case 5 has the lowest knowledge and awareness of the data analysis models, but relatively high (4/5) awareness. The differences between awareness and attitude could be seen as a factor for learning and adopting new analytic models in the near future. This could be also explained in terms of the need for competent data analysis. Based on the average score of attitude and awareness, case 4 is the highest with a score of four and case 5 is the lowest (2.88). The differences here are reflected in the budget allocated to data analysis technologies and resources. Case 5 has the lowest attitude towards the use of numbers in planning and the lowest knowledge about data analytics, possibly because the team is still relatively new and does not have access to resources. The coach is local, with no access to formal training on the use of data analytics on planning. Additionally, the management of case 5 has not invested in such technologies because there is limited belief in these technologies, whereas case 4 has a significant budget in these technologies.



Figure 6-3: Coach Competences

6.4.3 Technological Resources

The technological resources proposed are videos, databases, sensors and software applications. The following sections validate and verify these factors.

6.4.3.1 Validating IT Resources Model

In validation, the technological model proposed in the previous chapter was accepted by all nine interviewees. A comment was raised about the database and about the sufficiency of the model. The first uses FIFA videos and analysis, which provides a database of players.

"We use also FIFA videos and its analysis. It issues CD contains analysis of the world cup matches. It provides insightful analysis. We use these CDs to create benchmarks for our team performance. We use some videos as to improve our player and coaches' competences" W2FC6

The limitations of using FIFA CDs is that data are limited to only world cup matches, which is less useful for teams in the premier league. It is interesting that videos are being used to set new international benchmarks for the players. The other comment was on the sufficiency of the model, with criticisms being targeted at the weakness of the model without support by experienced data analysts with different specialities.

"I lead a national team for 4 years as a first coach. In the last season, we used different devices to track speed, distance, sprints, recovery rate, breath, and maximum distance in sprints. We get best use of it once we hired physical specialist who take control of the analytic process. He led the analysis process and present the reports weekly and disseminate relevant information to players. Also, he helped us in identifying new targets for players and investigating potential reasons for any gaps in the performance" W2FC4

This point is addressed in greater depth in the human resources section, although the model framework clearly relies on both human and technical resources. Others supported the efficacy and appropriateness of the model.

"InStat, Camtasia, and Dartfish are used. Also, we use PowerPoint presentation to present the ideas and statistics to the coach. We present information about effective attacks, number of correct passes. We use videos to support the presentation of KPIs and some videos to clarify unique behaviours" W2FC5

6.4.3.2 Verifying the IT Resources Model

Teams have different evaluations of the importance of each technology. While the premium league teams (case 1, 4 and 5) evaluate the use of databases as the most important, with scores of 5 for all teams, the national teams (case 2 and 3) perceive these as less important, with scores of 3 for both teams. This may be due to the availability and reliability of the information for different teams. While the professional league teams can play more than 70 matches per year, the national team plays six or more friendly matches per year. Accordingly, national teams (case 2 and 3) are perceiving the use of videos and tracking sensors as more important during the training sessions, to give them more accurate, reliable measures than secondary reports produced for matches at relatively low intervals.

"For Information Technology (IT) and the tools to use, it's important to connect training sessions data with real life game data in order to get connected data of the player to improve his/her performance and to get the best outcome." C1DA1

Accordingly, the level of usage follows the same pattern: case 2 and 3 use more sensors than other teams in tracking player performance during training, whereas case 1, 4 and 5 rely more heavily on databases than national teams.

The use of software applications seems to be associated with the level of data analyst competences, as case 1 and 3 have higher competences than the average (see Data Analyst 144

Competences) and are also using more software applications than normal in the KD process.

When comparing national teams, case 3 is doing slightly (0.6) better than case 2, perhaps because they have a more qualified data analyst, who has a master's degree in football data analytics. This expertise might give the team an edge over case 2, whose data analyst bases insights on football experience, rather than formal education in quantitative analysis.

"Sometimes you want to explore or use a new tech, but you can't because of financial matters" C2DA2

Comparison of professional league teams averages, case 1 is the best while case 4 is the worst, with a difference of 1.2 in a scale of 5. This difference could be explained in terms of the greater resources available for KD in case 1. This variation can also be understood in terms of head coach/team manager passion and attitude towards the use of data in planning. Hence, case 1 is the highest in the head coach attitude, while 4 and 5 are the lowest on the same scale, as detailed in coach competences section.



Figure 6-4: Technologies Used in Knowledge Discovery in Football

6.4.4 Value Co-creation

Value co-creation activities are built upon three main factors: structured communication channels, use of storytelling, and sprints. The following sections seek to validate and verify these aspects.

6.4.4.1 Validating the Value co-creation model

In general, all participants in the validation process were engaged and interested in the tools proposed during the value co-creation model. Three had not adopted any of these tools, but they were interested and expressed a belief there is a need to apply them, while others already use some of these tools but not in a systematic way. The conclusion is the tools are validated as being useful for coaches and data analysts. The main comments are documented below serve as evidence for such validation.

"I completely agree with this framework. It makes sense in terms of the iterative and continuous work and meeting between the coach and data analyst. I believe the analyst is a member of the coaching team" W2FC6

Similar feedback was received from another expert. However, he added the routinisation and documentation of such process would provide valuable lessons for learning for helping new members.

"Yes, I agree with having a systematic and lean approach in discovering knowledge. Also, documenting requirements and analysis to be as lessons learned. it will be very helpful in supporting new member joining our analysts' team" W2FC5

Tools were also validated by experts. Some teams use such tools without giving them to similar terms, whereas others found they are innovative and useful ideas. Starting with an examination of user stories/user questions:

"In the analysis process, we (coaching team) have around 35 questions about our team and opponent team. Questions such as speed, crossovers, and maximum sprints. By answering these questions, the picture becomes clear enough and from this point we can decide the direction of the knowledge discovery process" W2FC4

Likewise, story cards were perceived as an innovative method that made sense to the participants. Some teams already use this artefact without calling it "story cards".

"We sometimes have things similar to those cards. In questions, we ask about "areas of intensity and pressure of the attack from the opponent, players performance, the performance in the wings, the main strength pitch areas of the opponents, recovery time of the players after sprints, distance between lines, the dispersion of players in the pitch, the timing of closing the gaps between the lines and time of increasing the gaps between the lines" W2TD1

Similarly, 'sprint' was perceived as a convincing and useful tool that some teams were already using.

"Some analysts give you the analysis on small batches. You as coach give them the highlights and directions and the role of the analyst is to go in-depth and go for details. Analysts are able to develop new insightful ideas, all what they need is the direction and guidance" W2TD1

"Each week the analysis is improved and updated. It is not the same thing. Each match needs new knowledge, new insights, and new strategies. It is not copy and paste. It is an iterative and evolving process" W2FC4

"Yes, this is similar to what we do! Each period of time, data analysts update us with new fresh insights. We (the coaching team) sets the directions and they feed us from time to time" W2FC6

6.4.4.2 Verifying value co-creation

All teams agreed on the importance of structured communication between stakeholders. They also unanimously perceived the ideas of storytelling and sprints as being critical to ensure that team managers and data analysts understood one another and were working towards the same goal. In terms of self-evaluation of the performance on each aspect, case 3 rated itself as 5 out of 5 in structured communication, while case 2 was the lowest with score of 3.8. The reason here in that these variations could be understood in terms of the time working as one team, while the team manager and data analyst of case 2 and 4 have only started working together recently (4 months and 2 months respectively), while case 3 has a relationship that was established more than a year ago, enabling creation of a structured communication channel. The use of sprints was roughly identical for all teams, with score of 4.5, except for case 3, which scored 4.75. This gives an indication that teams are consistent in their perception of the importance and usage of these tools for collaboration in the production of meaningful knowledge.

Finally, storytelling was used most often in case 5 (4) and least often in case 3 (3.4). There is no evident reason for this variation. It may simply be an expression of the preferences of the coach and data analyst, who prefer to rely on sprints than storytelling.



Figure 6-5: Use of Value Co-Creation Tools

6.5 The use of KPIs in the Cases

The validation of KPIs occurs in four levels: the definition, importance, usage, and relations (for KPIs BSC). The structure of this section examines these issues with respect to the four main KPIs: physical, technical, tactical and psychological KPIs.

6.5.1 Physical Indicators

In this research, physical KPIs denote physiological and fitness measures for players. Some are traits that cannot be changed, such as height, while others can be improved by training such as speed or recovery rate. Physical KPIs are classified into speed, movement and distance. All the cases besides the experts interviewed, agreed on this definition, with all criteria given scores of 4 or more. Only two comments are proposed. Case 1 suggested adding an additional KPI to measure the 'agility' of players saying

"I connect "distance", but endurance & stamina. And movement can be defined as change of direction & jump. And I would add "agility" as a physical." C1TD The second added the need for the use of intelligent technologies in measuring the tracking these physical KPIs.

"It is important to reflect the use of intelligent system in measuring and tracking physical KPIs" C3P1

All teams agreed on the importance of using physical indicators in planning, with all participants grading this as 5, apart from case 4 and case 5, who graded it 4. This lower score may reflect their perception that using numbers in planning is less important. This is reflected in usage levels for these indicators, with case 1, 2 and 3 scoring usage as five, in contrast to case 4 and 5, who scored usage as 3.5 and 3. The lowest was case, not only because of low interest in statistics, but also a lack of KD resources and limited access to a part-time data analyst, software applications and access to different databases.

I think I might need to code the cases team with other codding names just to avoid confusion and be clear when discussing results not to get mixed with other teams in the framework.



Figure 6-6: Use of Physical KPI

6.5.2 Technical KPIs

Technical KPIs are defined by this research as "different individual football physical competencies required to control or to regain the control, to direct the ball, and to build constructive movements during the match". They are classified into off the ball

competences (ability to regain the control) and on the ball competences (ability to direct the ball towards a constructive movement)". There is a strong agreement on this definition across the respondents.

"Technical define as the level of player with the ball in controlling, passing, shooting, tackles, dribbling, and header." Case 1

"You should also add accuracy of long passes" Case 2

Both comments are not contradictory with the definition or objective of this research. Accordingly, the definition is validated in this research. Case 1 adopts all indicators mentioned in this research while case 5 is the lowest. This finding is aligned with the research proposition that the factors mentioned in the maturity model can function as predictors for the level to which the indicators are used in planning.



Figure 6-7: Use of Technical KPIs

6.5.3 Tactical KPIs

Tactical KPIs are defined by this research as "as metrics to measure the players' ability to position himself in the pitch effectively and efficiently in such a way the probability of passing, possessing, scoring and intercepting are improved". The tactical KPIs are measured by player, unit of play (set of players), tactical lines (e.g. attacking, defending, or midfield line), or by the team. They are classified into passes, possession, and playing style." There is only comment received on this definition.

"Tactical is define as ball in possession "attacking", ball out of possession "defending", transition from attacking to defending, transition from defending to attacking. And may add individual player intelligent and insight." C1DA1

This comment supports the current definition rather than being contradictory. Accordingly, this definition is validated. The usage level is associated positively with the level of resources identified in the maturity framework. In other words, case 1 has the highest level of usage, whereas case 5 is the lowest. This finding verifies the arguments of the research that the factors identified enables teams to identify new knowledge by applying and innovating different KPIs.



Figure 6-8: Use of Tactical KPIs

6.5.4 Psychological KPIs

This research defines psychological KPIs as the ability to play in the standard performance under different psychological pressures. I.e. psychological resilience indicator. All participants agreed on this without adding comment except for case 1

"Psychological can be defined as playing under pressure, leadership, composure, anger, and motivation." C1P1

This definition is consistent with the definition in this research. Accordingly, this research definition of psychological KPIs is valid. The level of usage for this KPIs is the highest in case 1 (5) and lowest in case 5 (3). This resonates with to the coach management style and data analyst competences. These indicators are not normally incorporated into the

software packages available on the market, instead requiring special coding for resilience indicators and discipline indicators. Accordingly, its extensive KD resources enable case 1 to adopt these indicators more effectively than case 5, which has very little investment in KD resources.



Figure 6-9: Psychological Indicators

6.5.5 KPI Balanced Scorecard

The balanced scorecard metric was validated by asking the experts and cases under investigation about the level of their agreement with each of the statements. A presentation briefing was given on the scorecard. The respondents appraised the framework and unanimously agreed that it was accurate. The KPI balanced scorecard framework was revisited and positively reviewed by experts. The validation examined the importance of understanding the relationships between KPIs, the existence of such relations as proposed in the framework, and the acceptance on the categorisation of the KPIs.

First, the importance of understanding the relationship between KPIs as an approach for knowledge discovery has been verified with all interviewees. The feedback addressed the role of such a model in KD.

"Statistics in itself add so little to an experienced coach. It is useful only if it is used to test such relations. It provides a concise and persuasive evidence for relationships between KPIs. This is the way the knowledge can be discovered" W2TD1

"This is really good, you got really good information here" W2FM8

The other feedback addresses the importance of having knowledge discovery technological resources. This framework can work only if such technological resources are available.

"This is very insightful and a rigours model. It is critical for discovering the knowledge and empowering the coach to discover the right knowledge for the game. But the main thing stopping me is the data. How can I get all of these data without having strong IT platform and centralised database? If I have such detailed information about the player performance in different contexts, I would say this framework will be useful. Also, I would like to add something here, the physical performance is monitored by a specialist who gives a detailed report about the player fitness. But it makes sense to connect that with other performance areas as you show here. I would be happy if you write this framework in a detailed book to be reference for discovering knowledge in football. I would say if this framework supported by videos and pictures, this could help a lot" W2FC5

Second, the existence of these kinds of relationships has been verified and confirmed by participants.

"I agree with you. This makes sense to me. I see the most important things affect the match result and tactical skills are the physical and psychological performance. Most of big teams now have specialists in psychological and physical performance. The teams that can understand and manage these factors of the players, they are the winner" W2EM2

In addition, there was broad agreement that tactical performance is primarily dependent on the ability of a player to understand the tactics of the game and match that understanding with high performance.

"The good attacker who has a good vision. He makes the play maker. Also, the good defender who can understand the opponent attackers' tactics" W2TD1

Tactical skills are confirmed as being critical for match success.

"a 90 min match I might not need all the player successful passes or shots however I will need the player critical passes that leads to a goal, created a chance/opportunity to attack and score that what will benefit me as a coach – not only 80% successful or 20% unsuccessful passes this reading does not help me a lot. This is very simple from 3/5 and a maximum of 6 if the player is a super star - He can create 6 attacking opportunists." W1TD Also, the influence of the psychological indicators on the physical, technical and tactical performance of players was recognised as being extremely significant.

"Yes, psychological factors play an important role in performance. You can see a player feel tired after 20 minutes. Even if his normal physical performance is far beyond that, he can get tired because of the stress. Stress pushes cortisone into the blood, which makes the body dysfunctional. This can influence the player performance in all aspects." W2EM2

"Yes, psychological factors influence the other aspects of a performance. If the mood is negative, the physical and technical performance will generally be weaker than expected. Even the ability to pass and or understand movements will be weakened. You are right, you connected these different elements together in a good way." W2FC4

A different review of the proposed framework focused on the belief that tactical performance can influence technical and physical performance. When opponents have stronger tactical skills, this can have a negative impact on the value of physical and technical skills in a team.

"Yes, but there are other factors not noted here. Something out of the framework; it is not connected with physical or tactical skills. The opponent ability to read my tactics and ability to control the game though the superior tactical performance. In this case, technical and physical performance could do nothing against such team" W2TD1

In general terms, however, there was general acceptance of the categorisation, with unanimous acceptance of the definition and classification.

"If my defender is not able to catch the attacker, the attacker is far faster than the defender, this of course will be analysed by physical measures and forward to physical specialist. If the attacker of the opponent team is clever enough in dribbling any my defender is not able to do proper interception, this will be covered in the technical training. Technical and physical indicators are different, but they are linked" W2FC5

Physical and technical skills are mainly gifted skills which cannot be changed easily. In addition, they are most often used to compare the players in opponent positions (i.e. defenders and attackers) to explore area of weakness in the team and in the opponent one.

[&]quot;Technical skills and physical performance are something gifted to the player. Training can do few here" W2TD1

Experts agree on the importance of tactical skills and importance of training on them

"Yes, tactical skills are the one. For instance, penetrating the defence lines need tactical skills. As a coach, with help of data analyst, identify the weaknesses in the opponent team. Here you can train player on specific key passes or throw passes that could penetrate weak areas in the opponent team. Here we present videos for the team players visualizing the tactics of the match to penetrate such lines in a specific time" W2FC6

In addition, the framework was verified by asking respondents to evaluate the impact of each type of KPI (physical, technical, tactical, and psychological) using a continuous itemised (5-point Likert) scale to give more freedom for the evaluation. The summary of the verification process is tabulated below (see Table 6-2). According to the table, the ranking of the factors affecting the match performance is technical, tactical, physical and psychological factors. The framework rank order of the direct impacts is tactical, technical, physical and psychological factors. The difference between the verification result and the proposed framework is the order of technical and tactical. Three of respondents evaluate the tactical is more important while only two said the technical is more important than tactical. Furthermore, two participants evaluated the impacts as identical. Bearing in mind, the framework stating the tactical has only a direct effect while the technical has a direct effect and indirect effect (through tactical indicators), this can be a reason for the two respondents who evaluated the impact of technical KPIs is higher than the impact of tactical KPIs on the match performance. The role of psychological KPIs on technical, tactical, and physical performance is very close without significant differences among them. This also verifies the original framework. But the framework proposes psychological factors can influence the physical slightly more than technical KPIs however the verification showed something else. But because the difference in the score is only 1.1 out of 100, it cannot be argued the psychological impact on the technical is higher than the physical. But both the framework and verifications show the least impact of the psychological is on the tactical impact.

The proposed framework is similar to the verification results in terms of identifying the factors affecting tactical performance. In order, they are technical, physical, and psychological. This research aims to develop this KPI BSC in order to illustrate how the interactions between KPIs shall be investigated to discover new knowledge and serve as

a road map for the discovery of new knowledge. Indeed, the participants in this study were pleased to adopt this framework in their daily data analysis

"Very comprehensive and useful! I will be happy if you write a book about that. This can be a useful encyclopaedia for us in football knowledge discovery. This will help us in doing what we do" C2DA1

Table 6-2: Experts Evaluations regarding KPIs BSC

BSC KPI	Case 1	Case 2	Case 3	Case 4	Case 5	Case 6	Case 7	Average	<mark>Rank on</mark> Match	Rank on Tactical	Psycho Impacts
1- The role of physical/fitness indicators on the tactical indicators	100	100	100	60	60	100	63	83.3		2	
2- The role of physical/Fitness indicators on the technical indicators	100	82	100	70	50	100	66	81.1			
3- The role of physical/fitness indicators on match performance	100	100	100	65	50	100	76	84.43	3		
4- The role of technical indicators on tactical indicator	100	87	100	80	90	100	92	92.7		1	
5-The role of technical indicators on match performance	100	95	100	75	90	100	93	93.3	1		
6- The role of match tactical indicators on match result	100	100	80	76	80	100	84	88.571	2		
7- The role of psychological indicators on physical indicators	100	85	30	70	40	100	39	66.3			2
8- The role of psychological indicators on technical indicators	100	90	40	70	40	100	32	67.43			1
9- The role of psychological indicators on tactical indicators	100	64	20	64	40	100	68	65.14		3	3
10- The role of psychological indicators on match performance	100	100	50	64	50	100	61	75	4		
Average	100	90.3	72	69.4	59	100	67.4	79.73			

6.6 Use of Analytical Models

Validating the analytic models was challenging because interviewees were unfamiliar with statistical analysis, many using visual built-in tools. Many of the data analysts were not even familiar with custom made data solutions. However, they are involved in trials demonstrating the importance of such tools, even if they are not currently using them.

"this makes sense here. I can give you example to support your research. **P7** and Rivas are two players. we spent lots of time understanding their performance. It was not clear for us the reasons for having different performances for each. After time, we have discovered that their performance affects each other. They are working closely as twins, if any of them is not playing, the other performance influenced negatively" W2EM2

Participants acknowledged the importance of analytic tools, but with a relatively low level of usage or adoption. The highest level of adoption was team 4, while the lowest was case 5. This variation can be explained by looking at the level of attitude and awareness of the coaches. The coach of team 4 is passionate about statistics, while the coach of team 5 has the least interest in data usage. All cases were below average on other scales, showing that there may still be a large scope for improvement if these teams utilise the power of KD.



Figure 6-10: The Use of Analytic Tools

6.7 Validating the role of KD in creating effective strategies

The participants agreed with the role of KD in the strategic models of team managers, with explicit comments addressing and magnifying the value of this model.

6.7.1 Transferring Strategies

All participants agreed that KPIs can play a significant role in the selection of players. However, one issue raised is that the current use of KPIs is insufficient on its own to obtain access to the best players. Accordingly, observation is necessary before taking decision, because of the weaknesses in terms of current KD practices.

"Yes, we do that, when we search for new players, we look at indicators about physical and technical performance. These figures are fixed and updated every 10 years. From age of 6 to age of 16, the performance is standardised. After 16, the physical performance of players changes significantly, here you can know whose performance is the 'standard' or above 'standard'. Bu our current set of KPIs, to be honest, are not sufficient to taking decision of transferring. It is just a starting point. I need to see the player by myself to see his personal interaction with others. Also, my coach team comes to assess different aspects which we do not know how to quantify using technologies. May be knowledge discovery science could play a role here! Who knows?" W2TD1.

Similar perception perceived and noted by the youth team manager

"Some coaches do skills/technical test for young player, evaluate and record it, this renewed every 10 years (based on the age categories (under 12 under 18 over 19). We observe Players performance in the match besides using tracking devices and recording player performance." W2FC5

Also, that was highlighted by a Team Manager,

"But again, if you work as an analysis and you know what I like in a footballer, what roles I expect a central midfielder to play and if the analyst know that then you could do your recruitment around that" W2FTM8

6.7.2 Training Strategies

Participants agreed that KPIs can play a significant role in training players. However, one point is raised here which is the effectiveness of training in improving players KPIs. The current use of KPIs do not show the maximum player performance can be achieved by training.

"I agree with you! Not all skills can be trained. There are special skills that are gifted, such as playing by right and left legs. These are not trainable skills. Also, the ability to change performance is limited. You have to know the limits of your player. This can help you when deciding to train or transfer." W2TD1

"Some skills are not trainable at all. Each position needs some skills that some of them, especially technical and tactical skills, are not gained by training" W2EM2

6.7.3 Match Strategies

The main use for KPIs is the development of match strategy through comparison of dynamic team performance with opponent teams to discover areas for improvement and potential remedies for ongoing issues.

"This is what I want from the data analytics. I want to collect all useful and insightful information about the players in my team and the opponent team. Measuring everything whatever can be measured from technical and physical performance. Based on such comparisons, I define the game strategy and match tactics" W2FC5

6.8 Case Studies Feedback

The participants from the case studies stated that they were extremely happy with the analysis and the assessment. Overall, the feedback was extremely positive.

"This is very insightful, thorough, and comprehensive. I believe it covers all aspects of KD." C2DA1.

"I hope that you can put all of these things together in one book to be a reference for coaches and data analysts for improving our performance" C2DA

"Interesting research, useful and doable." C3DA1

"I agree with these findings! All good! I cannot agree more." C4AC1

6.9 Summary

In order to validate and verify the research findings, a Knowledge Discovery Maturity Model (KDMM) was proposed in the previous chapter. In this chapter the maturity model was validated by experts and validated by 5 case studies. This step aided and supported in operationalising the research findings. The aim was to target heterogenous sample that should reflect variations of competences. That variations should be reflected on the

outcomes of KDMM across the different participants. Additionally, that will aid in spotlighting any weakness in the KDMM and aid in improving it.

The KDMM considered the following competences. Human Resources Competences (i.e. data analyst competences, coach competences), Technological Resources Competences and Agile Value Co-creation Competencies. It inspected the use of different KPIs by the different cases. Furthermore, The KDMM investigated the participants understanding of the relationships between the different sets of KPIs to identify potential areas of improvements.

Revisiting the outcomes of the KDMM, it is noticeable that there are variations between the data analyst competences across the five cases. The key highlighted by this chapter is that the improved realisation of the team's available resources is derived from the human resources competences of the team. The key point is that even if some teams have exceptional technologies that does not mean they are exceptional in understanding their players' performance than their opponents. It is an integrated cycle that require orchestrated utilisation of resources, competences and collaboration (i.e. Agile value cocreation). That should aid in improving the KD process of the coaching team allowing them to better understand their team performance. For example, if the coach has low attitude in using data to build on the team strategies that wouldn't assist him in realising the value of it. Additionally, it is noticeable that the data analyst knowledge of football is influencing the level of communications within the coaching team as seen in Figure 6-2. The could be perceived as a key in improving the value co-creation between the coach(s) and the data analyst. Finally, proposing the KPIs BSC was received positively by the experts. The visualisation of the sets of KPIs delivered improved and informative understanding of the possible relationship between them. That led into encouraging valuable discussions during one of the interviews. As an example, most of the experts agreed that tactical KPIs have a direct effect to match strategies. However, some experts emphasised that technical KPIs sets may have stronger effects on match outcomes. That does not contradict the research proposition, in fact, it supports that since technical KPIs sets have direct effect on tactical KPIs set directly and indirectly (see Figure 5-5). Due to the participants availability I was able to obtain limited feedback of the case studies. The feedbacks were positively encouraging and supporting future research collaboration in this area. The research findings were able to address professional concerns regarding their
team performance. Additionally, that aid them conceptually to spot out area of weakness and strength in their team and match analysis practices and activities (see 6.8 Case Studies Feedback).

Chapter 7 Discussion and Conclusion

7.1 Introduction

Knowledge Discovery (KD) is a new science that emerged concurrent with the concept of data mining and big data. The main difference between KD and other concepts is the focus on creation of knowledge through data analysis. In other words, KD is primarily concerned with complementing data reports with the observation of practices, as well as using experiences to create knowledge to enable organisations to outperform their competitors. This area is rapidly growing in the world of football, with investment in such technologies in the football industry exceeding \$1 billion by 2015. However, as such investment only started in Saudi Arabia in 2016, this study has examined the question of why some teams can outperform others through these technologies.

The Saudi Arabian sport context has yet to adapt to modern data analytics, meaning that a dedicated study that created one algorithm for a specific case would be likely to be undervalued. Therefore, the aim of this thesis has been to create a platform and a road map to unlock value from KD technology in a functional context. Thus, this research has attempted to develop a framework for understanding the value co-creation process between data analysts and head coaches using KD technologies. The development of such a framework enabled a maturity assessment tool to be created that was able to map the strengths and areas of improvement for teams, enabling them to gain value from KD technologies. Therefore, this research developed a taxonomy of KD technologies in football, identified required competences to utilise such technologies, the required KPIs and analytical models that could be a roadmap for data analytics and coaches for discovering knowledge, and identify the potential values that can be realised from them. This chapter begins by revisiting the research question, aim and objectives, then highlights the contributions of this study to the literature in this area. The implications of the research are then discussed. Finally, after identifying the challenges, this chapter outlines the limitations of the current study.

7.2 Revisiting the Research Question, Aim and Objectives

7.2.1 Research Question

This research starts by this question "Why are some teams better able to gain value from investment in knowledge discovery technologies than others in the football industry?"

The reasons for this variance can be explained by the resource-based theory, with the core reason for different levels of value being the different availability of technical and human resources among teams. In other words, not all teams have the same access to technologies and not are equally in terms of their human resources competences. The value is only perceived if the KD technologies are used to improve the decision making process of coaches through improvising the abstraction and visualisation of the contexts in terms of using new KPIs and new analytic models. Through this better understanding of the contexts than competitors in terms of using these new KPIs and analytical models, effective changes can be made to the coaching and training practices. Otherwise, if they are not reflected in changes in coaching practices, the existence of these KD technologies could be valued as virtually negligible.

7.2.2 Research Aim

This study sought to develop a framework for understanding the variation in the value cocreation process from the knowledge discovery systems in the football industry.

This study has fulfilled its research aim. Value is only created when coaching practices are altered due to the existence of new knowledge. This knowledge can be operationalised into new sets of indicators, or alternatively into mathematical or visual models that can help coaches to see something that they could not have seen before. Additionally, the value is constructively co-created among stakeholders through dialectics and communications. These communications and interaction between different stakeholders then provide questions and inquiries that ultimately create the requisite knowledge. The ability to scope and address questions requires certain competences from the actors involved in the value creation process, however. Additionally, technological resources can enable and enhance the ability of actors to create new substantive inquiries. In terms of the findings of this study, technology is an enabler and a delivery mechanism, but not the sole reason for performance variation.

The framework in this study consists of four main models: the technological resources model, the human resources model, the value co-creation model, and key performance indicators model, which creates a Balance Scored Card. This framework is mapped over the benefits map to show the interactions between different elements and to prescribe the journey from having the resources to realising the value from investing in this resources. The resources come together to build outputs which are new KPIs and new analytic

models. These outputs create explorative and exploitative capabilities in terms of ability to understand the team players' and opponent team players' performance so that the SWOT analysis for the matches can be created effectively. Through such understandings new outcomes come up in terms of the new insightful and knowledge-based training, transferring, and coaching strategies so that the performance results can be improved. All of these models are integral components of the framework, enabling comprehensive understanding of the value co-creation process for KD systems.

7.2.3 Research Objectives

1- To operationalise the expected value of the KD to the coaching team

The value of the KD process comes from the changes in coaching, training, and player transfer practices. This comes from the access to well-informed, evidence-based, verifiable knowledge, augments the decision-making process. However, KD refers to not only the use of indicators or visual analysis, but also the creation, association, correlation, and integration in analytic models of these indicators or visuals. Each new indicator can be seen as a potentially valuable perspective with which to better understand the performance of team players and their opponents, which creates new opportunities for the coach. Each new relationship between indicators can also be understood as new knowledge that could empower the coach to create robust plans relative to those opponents who do not have access to such knowledge.

Value depends on the level of change made to practices and improvements made to performance. The level of change in practices is assumed to be reflected in terms of levels of new knowledge generated through KD activities.

2- To identify and taxonomies the KD resources and depicts a model to understand the role of each class of football technologies in improving the coach performance.

The term technological resources refer to hardware and software devices or systems. Technologies include data capturing technologies (tracking, body sensors), database interfaces, annotations software, and KD analytics. Data capturing technologies allow the coaching team to record data for further processing. Database interfaces support the coaching team with statistical datasets concerning players and teams. Some also come with visualisations and team specific reports (not customised based on the coaching team preferences). Annotation software helps to process, and code pre-set, or specified, events or football actions. Lastly, KD analytics applications help coaching teams to analyse data

for deeper insights through the development of customised models based on specific interests. The capabilities in using these technologies, adapting them to serve specific needs and coaching vison, and developing related analytical models are based on the coaching team needs.

3- To identify and frame the role of different knowledge, skills and competences required from producer (i.e. data analyst) and consumer of the knowledge (i.e. coach) so that the expected value from the KD can be realised.

The roles that need to be understood in terms of gaining value are consumer and producer. A producer needs to be competent in using statistical analysis, software applications, communications, and football planning competences. The more competences that a producer has, the more he will be able to contribute to the KD process and help the consumer digest and absorb knowledge, thereby enabling it to be reflected in practice. A consumer needs to have a strong attitude for quantitative analysis and reporting to be able to contribute to this process.

4- To frame the value co-creation process and augmenting it with different tools to improve its value.

As an approach for the improvement and encouragement of effective, efficient communications, this study proposes the use of agile approach tools and techniques. These should enable the formulation of superior communication processes and practices between the producer and consumer of knowledge, potentially helping the optimal benefits to be created and accepted. The key tools proposed for understanding analysis requirements and delivering solutions based on accepted criteria are Use Stories, Agile Release Planning and Story Mapping. Iterative validation and evaluation techniques for these deliverables (i.e. analysis, tasks, reports and stakeholders' agreements) should be undertaken using Sprint and Retrospective approaches.

5- To develop resource-based maturity model to identify the weaknesses and strengths in the augmentation of the resources to get value from knowledge discovery activities.

The KD Maturity Model in Football Performance Analysis was developed to verify and apply the research findings. Because the research is proposing that clear criteria for KD processes and practices results in more effective and efficient KD outcomes, the levels of processes and practices need to differentiate between applicant maturity levels. This research proposes the use of four levels: Ad-hoc, Defined, Managed and Optimised. These levels differentiate the capabilities to analyse football performance data with respect to people, namely the coaches, the data/video analyst(s), the coaching team and any other related stakeholder(s). Additionally, the levels are gradated with respect to the following Key Process Areas (KPA); Human Resources Competences, IT Resources Competences, Knowledge Co-Creating Practices, Use of KPI Competences and Analytical Modelling Competences.

7.3 Contribution to Knowledge

This study provides six main contributions to knowledge, as follows:

- A novel adoption and customisation of the framework proposed by Melville *et al.* (2004). It is the first research to use it for developing a framework for understanding the value co-creation process from the KD technological resources.
- 2- A novel proposal and verification of Agile methods, as borrowed from software engineering literature (Beck *et al.*, 2001; APM, 2015). This provides valuable ways in which to improve the value co-creation process between coaches and data analysts for effective use of KD technologies.
- 3- The novel development of a taxonomy of KD technologies in the football industry.
- 4- The novel use of a KD maturity model utilising the resources approach.
- 5- Clearly articulating the required competences from data analysts and team managers for the value co-creation process in KD in the football industry.
- 6- The development of a novel KPI balanced scorecard, borrowed from organisational strategies literature (Kaplan and Norton, 1996) and its adaptation as a road map for the KD process in the football industry.

7.4 Academic Implications

This research in novel in integrating two complementary frameworks. The first framework was originally developed by Ward and Daniel (2012) and subsequently extended in other studies (e.g. Serra and Kunc, (2015); Badewi, (2016)). This framework demonstrates how project outputs lead to outcomes that can be translated into capabilities and changes in performance (benefits). This research is the first to borrow these concepts from IT project management literature. The current study has approached the data

analysis process as a value co-creation project that leads to benefits when the team strategy development is improved by superior understanding of the environment (through using new KPIs and data analytic models) through the ability to use KD technical resources.

This research borrowed the second frameworks from IT business literature, which is resource based view in IT value creation (Melville and Kraemer, 2004). This is not the first research to utilise this approach for improving sports performance. The resource-based view has been used to understand the resources required for team managers and coaches to improve their training and match strategies (Smart and Wolfe, 2003; Lechner and Gudmundsson, 2012; Costa *et al.*, 2018). But this research is novel in using it for investigating the relationship between IT resources and organisational complementary resources in the creation of expected values in the knowledge discovery and in football contexts. In other words, exclusive reliance on IT resources is less effective than when the KD process is supported by complementary resources.

Neither frameworks have been applied in previous studies of computing and football. The adaptation of these frameworks could provide deeper understanding of the practices and processes that exist between the coach and data analyst in analysing football data. This research is the first to combine two frameworks in KD literature to understand the value co-creation process. This process has been developed from IT resources and complementary resources to obtain new outcomes (using data analytics to derive new KPIs and data analytics models), new football team capabilities (better understanding of external and internal environment), and higher performance for the teams involved (ROI on players and match results).

The analysis has informed the development of a new KD maturity model to assess the ability of teams to utilise KD in value creation. The corners of the maturity models are IT resources (hardware and software), competences of data analysts and competences of team managers, agile practices, and sophisticated use of KPIs and data analytics models. The levels of maturity are ad-hoc, defined, managed, and optimised, with ad-hoc reflecting a lack of resources required to realise value from KD technical resources to the optimised level, which describes the highest utilisation of the resources to gain value.

7.5 Recommendations

1- Recommendations for Data Analysts

Data analysts need to be competent in using numerical analysis, software applications, communications, and football knowledge. The more that an analyst (producer) has these competences, the better he will be able to contribute to the KD process and help the consumer digest and absorb knowledge, enabling it to be reflected in practices. In the modern context, KD tools are now too complex to be learned from online learning. Instead, they should be studied under formal education. However, there are no colleges or academic institutes in Saudi Arabia delivering training on KD tools, football, numerical analysis, and communications to data analysts. There is no certification or any awarding body for accreditation as a football data analyst professional. None of the respondents in this study had received formal training on football data analysis, instead coming from a range of unrelated backgrounds. In Saudi Arabia, no academic institution covers this field, which no research being conducted into the improvement of data analyst competences. This weakness was observed in interviews with local data analysts, who had limited knowledge regarding basic analytic approaches such as regression, correlation, and data mining techniques. Although some where aware of the technologies, only one demonstrated awareness of annotation or other technologies. Rapid changes in technologies necessitates the involvement of research institutes, such as the production of magazines to ensure that data analysts are kept aware of recent trends. However, no such specialised news source exists for the Arab audience.

2- Recommendations for the Coach

As the consumer of knowledge outputs, the coach should have a positive attitude towards the use of these quantitative analysis and reports in their practices, informed by profound knowledge of quantitative analysis to enable them to contribute to this process. Currently, most 'big' teams are led by international coaches who have developed knowledge and competences. However, local coaches are less capable with numbers and have poorer attitudes towards statistics, perhaps because current coach accreditation does not require study of KPIs or technologies in sports. All courses reflect coaching and techniques, but none include use of data or technology in training. Therefore, training in such competences is essential, as seen in data analysis courses introduced by associations in other countries, such as the Football Association in the UK. Coaches need to familiarise themselves with data analytics and new technologies to be able to communicate effectively and constructively with team data analysts.

3- Recommendations for Team Managers and Policy Makers

Policy makers shall institutionalise the use of data analytics and KD logics in the ecosystem using different mechanisms. According to (Scott, 2010), institutional pressures can be mimetic, normative, or coercive. Mimetic pressures can occur when local teams compete against international teams several times, which will push coaches and data analysts to study what others are doing, potentially leading them to mimic these international teams and become more advanced. Currently, because the SPL lacks a data analytic and data knowledge culture, the attitude towards using data analytics for discovering knowledge is less than it should be. The next pillar of institutionalising KD logics is normative pressure, which pertains to the establishment of certification for practicing data analysis in football and coaching professions in Saudi Arabia. This accreditation would help increase the competence of human resources, enabling them to better cope with advances in technologies for absorption into the planning processes. Policy makers should therefore establish standards in coaching and data analysis informed by research, in an attempt to encourage adoption of KD principals in this industry. The last pressure is coercive, which means the use of laws and regulations to encourage data utilisation and protection of knowledge between teams.

7.6 Research Challenges

7.6.1 Acceptance of research practices

One of the main challenges for this research was the collaboration with organisational bodies, which was problematic due to conservativeness and the competitive environment. As a result, communication and response took a very long time, which was exacerbated by the busy schedule of the target audiences (i.e. coaches and analysts). The large distance between cities also made it challenging to travel around and accept other invitation. Travel from Jeddah to Riyadh by air takes a minimum of 2 hours and about 3 hours to Dammam. Additionally, the cost in funding, time and effort was very high. This research involved more than 3 overseas trips and more than 15 domestic trips (i.e. by car or internal flights) to collect, validate and verify the data collected. More than 10 virtual meetings were also done, supplemented by numerous telephone calls.

7.6.2 The collaborative culture between higher education and sports bodies

From the experiences of this research, it is apparent that no clear research collaboration exists between academia and sport authorities, perhaps due to minimal interest or a lack of clear guidelines in this area. However, I found great support and collaboration after visiting sport authorities in Saudi Arabia, who were very collaborative and supportive. Therefore, while the initial stages were very challenging, after meeting and discussing my research with representatives from these bodies, the process became easier.

7.6.3 Collaborations, Connections and Transparency

Relationships and connections are very critical in this kind of research, requiring extensive effort from the researcher in terms of travel and arranging meetings. Collaboration tended to be slow due to the busy schedules of the professionals in this field, which made scheduling data collection difficult. Transparency is also an extremely important concern in this field, due to the competition, sharing of information, education background and the tolerance for the impact of research in this field.

7.6.4 Cost: Inability to meet all coaches/data analysts

The cost for overseas and domestic travel required extensive investment of funds and time from the researcher. This part of the research was primarily funded by the researcher himself. Traveling time and distance (e.g. a minimum of 2 hours) made it challenging to address all invitations. This was exacerbated by the difficulties of arranging calls with the participating professionals.

7.7 Research Limitations

This research is interpretive in nature, aiming to understand and taxonomise the various factors contributing to the realisation of value from investment in KD technologies. As the study adopted a social construction of reality and qualitative, there are three inherent research limitations: generalisability/applicability in different sports/country contexts, the ability to test the results objectively, and the quantification of the impacts of different factors on performance.

7.7.1 Generalisability/applicability of findings

This research targeted top teams in different leagues in Saudi Arabia to develop a constructive framework that could serve as a roadmap for other teams to obtain value from investment in KD resources. Generalisability and applicability of the findings are limited to top teams in SPL. Precautions should be taken if the research recommendations

are applied to bottom teams, other sports in Saudi Arabia, or to football teams in other Arabic speaking countries in the Middle East.

The comparison was made between top teams that had different levels of adoption for KD resources and therefore different levels of realising value from them. Even though the researcher had the opportunities to meet with a local Olympic team, with the national Olympic team and with high level sport authorities, there were clear indications that there is extremely limited usage of technologies outside the national team. Considering KD practices in teams from low leagues teams would be an excellent opportunity to see analysis affects their practices and enables their progression to higher leagues.

This research has not studied or investigated the challenges faced by teams from lower leagues, who are likely to have less access to data analysis due to limited financial resources for hiring international coaches, data analysts, or even buying KD resources. This research therefore offers insights into the teams in the football sector in Saudi Arabia, but not bottom teams. The interviewees were also exclusively involved in the football industry, therefore care should be taken in the adaptation of findings or methodologies to other sports. Football has more financial resources than other sports in Saudi Arabia, as well as more competition and higher levels of care from sports policy makers. Application of the findings to other sports in Saudi Arabia could therefore entail other challenges.

Finally, there is a limited degree of generalisability from Saudi Arabia to other countries in the Middle East. For example, the level of English among locals is different in countries such as Jordan, Lebanon, the Emirates, and Egypt, potentially making it easier for teams in these countries to stay up-to-date with recent updates in the field. However, teams from these nations are also likely to have less access to financial resources than Saudi teams, making investment in KD more difficult. All of these factors can create differences in the results and recommendations.

7.7.2 Objective testing of results

Since this research is qualitative in nature, the ability to test results objectively was a challenge. There are two epistemological stances: social construction of reality and positivist. In positivist research, there is an assumption that respondents do not know and the role of the researcher is to test theories using quantitative and objective tools using correlational analysis and regression methods. Due to the limited number of participants in Saudi Arabia, the significance of the results will less meaningful. Having constructed

the framework in this study, future research could test the components in wider contexts, such as a Middle Eastern study of different coaches/data analysts, or even a global study.

7.7.3 Quantifications of the impacts of different factors on performance

Since this research is qualitative, the ability to quantify the impacts is limited. There are some intuitive based tools that can be used to quantify qualitative inquiries, such as Analytic Hierarchy Process (AHP). However, the AHP requires extensive time and effort from the respondents, which was not feasibly in the current research due to the extreme time limitations placed on coaches and data analysts, which already affected willingness to participate in this study. This limited time and availability of participants, it was preferable to benefit from gaps in their extremely tight schedules by validating and verifying the maturity model, rather than quantifying and weighting different factors.

7.8 Future Research

There are two directions for future research that have arisen from the limitations of this research: namely, to quantify and test research findings, and replicate this research on different contexts. In addition, there are three other research opportunities inspired by the research findings: understanding the role of cultural values on KD, investigating the institutionalisation process of KD logics in the football industry, designing a new software application to manage the communications, and lessons learned in the knowledge discovery process. These are outlined below.

7.8.1 Testing and Quantification of the research models

As noted in the research limitations, although these findings have been validated and verified, they have not been objectively quantified or tested due to limitations of sample size. For this reason, future research should test the research results using questionnaires distributed to coaches and data analysts at football teams in different countries in order to reach a significant sample size (e.g. 60) (Hayes, 2012). Quantification can also be done through AHP, fuzzy logics, Delphi approaches, pairwise analysis and other tools. However, as noted above, these methods require more engagement from the coaches and data analysts, so should be undertaken as a part of future research to improve the quality of findings and to make them useful for wider audiences, as discussed in the next section.

7.8.2 Replicability of the study on other contexts

The context of this research is football in Saudi Arabia. Different contexts include the level of team performance (top versus bottom league), country, or qualifications, background and education of the coach, background and education of the data analyst, and even the native language of the data analyst and coach. These factors may lead to different implications and recommendations.

Even though the research findings offered new insights into KD in performance, it would be interesting to see how cultural background influence these practices. In Saudi Arabia, there are different forms of education, coaching systems and certification, as well as variation in terms of competition strength. The study in the KSA context can be taken as a foundation for other studies to be undertaken in countries and continents. The different FIFA certification also might influence KD activities, since other countries have different coaching programmes and levels. Additionally, low league teams should also be studied, as they may face different challenges and issues, including minimal access to financial resources, which hinders their ability to access the technological and human resources required for knowledge discovery.

7.8.3 Investigating the institutionalisation process of the knowledge discovery logics in the football industry

The recommendations of this research propose that institutional pressures are used to push teams to use KD tools, techniques, and approaches in dealing with data. Integrating knowledge discovery into team planning logics has not been studied in the literature. Indeed, there is a general lack of research into the values governing KD logics in the football industry as a whole, perhaps because practices are not sufficiently mature. Globally, KD and data analytics in general are still relatively new and need further investigation.

This research illustrates the need for discovering insights into plans, the need for the communication and value co-creation process to be structured, the need for decisions to be augmented by verified knowledge, and the need for new dimensions of realities to be explored through the use of indicators and analytic models. There may be additional values past this initial list, so techniques and tools should be used to routinize and structure KD logics into the practices of contemporary teams to facilitate the rapid absorption of developments in technology.

7.8.4 Designing a new software application to manage the communications and lessons learned in the knowledge discovery process

KD is a value co-creation process that uses constructive communications and available technological platforms with statistical and analytic models to gain new insights. Constructive communications can therefore help in the creation of new analytic models. Research should be conducted to systemize the communication process to enable the value co-creation process to become easier, faster, and more efficient in utilising lessons. Research should investigate the design of a library of questions, indicators, and analytic models that have the level of errors, relationships, acceptance criteria, and retrospectives required to track the evolution of KD. The technical perspective requires the development of new information architecture to map this knowledge for fast, efficient retrieval, filtering, optimisation, and evolved to develop new knowledge in shorter time and with less efforts.

7.8.5 Benefits management techniques and approaches in Knowledge Discovery

Benefits management is a new approach to the realisation of value from investment in change (Badewi, 2016; Breese, 2015; Serra and Kunc, 2016). It defines the ownership, identification, planning, realisation, auditing and exploitation of benefits (Ward and Daniel, 2011). The value from investing in knowledge discovery technology may be increased by structuring and framing KD based on benefits management. For instance, if a coach identifies the benefits from investment in KD technologies, this may be an opportunity to obtain benefits before procuring the technology. If benefits are planned before data analysts are hired, the job description and specifications for this position could be clarified and made more transparent and effective. Planning for benefits also creates a sense of ownership in terms of the perception of the need to change the practices to realise these benefits. Additionally, planning and framing the process for realising benefits could help a team manager audit and control the realisation process. Finally, exploiting benefits rould help data analysts and coaches to brainstorm ways to realise new benefits from the current technologies, increasing utilisation of these technologies and raising the potential performance of the team.

References

Aamodt, A. and Nygård, M. (1995) 'Different roles and mutual dependencies of data, information, and knowledge — An AI perspective on their integration', *Data & Knowledge Engineering*. North-Holland, 16(3), pp. 191–222. doi: 10.1016/0169-023X(95)00017-M.

Abrahamsson, P. *et al.* (2002) 'Agile software development methods: Review and analysis', *Espoo, Finland: Technical Research Centre of Finland, VTT Publications*, p. 112. doi: 10.1076/csed.12.3.167.8613.

Adjei, D. *et al.* (2013) 'Performance on agile teams: Relating iteration objectives and critical decisions to project management success factors', *International Journal of Project Management*. doi: 10.1002/pmj.

Agile Alliance (2018a) *What are User Stories?* / *Agile Alliance*. Available at: https://www.agilealliance.org/glossary/user-

stories/#q=~(filters~(postType~(~'page~'post~'aa_book~'aa_event_session~'aa_experi ence_report~'aa_glossary~'aa_research_paper~'aa_video)~tags~(~'user*20stories))~se archTerm~'~sort~false~sortDirection~'asc~page~1 (Accessed: 21 June 2018).

Agile Alliance (2018b) *What is Kanban?*, *Agile Alliance*. Available at: https://www.agilealliance.org/glossary/kanban/#q=~(filters~(postType~(~'page~'post~' aa_book~'aa_event_session~'aa_experience_report~'aa_glossary~'aa_research_paper~ 'aa_video)~tags~(~'kanban))~searchTerm~'~sort~false~sortDirection~'asc~page~1) (Accessed: 6 July 2018).

Agrawal, R. and Shafer, J. C. (1996) 'Parallel mining of association rules', *IEEE Transactions on Knowledge and Data Engineering*, 8(6), pp. 962–969. doi: 10.1109/69.553164.

Ahlemann, F., Schroeder, C. and Teuteberg, F. (2005) Kompetenz- und Reifegradmodelle für das Projektmanagement: Grundlagen, Vergleich und Einsatz, ISPRIArbeitsbericht. Ahmad, M. O., Markkula, J. and Oivo, M. (2013) 'Kanban in software development: A systematic literature review', in *2013 39th Euromicro Conference on Software Engineering and Advanced Applications*. IEEE, pp. 9–16. doi: 10.1109/SEAA.2013.28. Ajzen, I. and Fishbein, M. (1980) 'Theory of Reasoned Action', *Social Psychology*. doi: 10.4135/9781483346427.n552.

Aliseda, A. (2006) *ABDUCTIVE REASONING*. 1st edn. Dordrecht: Kluwer Academic Publishers (Synthese Library). doi: 10.1007/1-4020-3907-7.

Alliance, A. (2018) *Sprint Planning Rules*. Available at: http://www.sprintplanning.com/SprintPlanningRules.aspx (Accessed: 25 June 2018).

Alsultanny, Y. (2011) 'Selecting a suitable method of data mining for successful forecasting', *Journal of Targeting, Measurement and Analysis for Marketing*. Nature Publishing Group, 19(3–4), pp. 207–225. doi: 10.1057/jt.2011.21.

Ambler, S. (2009) 'The Agile Scaling Model (ASM): Adapting Agile Methods for Complex Environments', *Environments*.

APM (2015) *The Practical Adoption of Agile Methodologies*. Available at: https://www.apm.org.uk/resources/find-a-resource/practical-adoption-of-agile-methodologies/ (Accessed: 6 July 2018).

Arksey, H. and Knight, P. T. (1999) Interviewing for Social Scientists: An Introductory Resource with Examples, Interviewing for Social Scientists: An Introductory Resource with Examples. doi: 10.4135/9781849209335.

Armatas, V. and Pollard, R. (2014) 'Home advantage in Greek football', *European Journal of Sport Science*, 14(2), pp. 116–122. doi: 10.1080/17461391.2012.736537.

Assfalg, J. *et al.* (2003a) 'Semantic annotation of soccer videos: Automatic highlights identification', *Computer Vision and Image Understanding*, 92(2–3), pp. 285–305. doi: 10.1016/j.cviu.2003.06.004.

Assfalg, J. *et al.* (2003b) 'Semantic annotation of soccer videos: Automatic highlights identification', *Computer Vision and Image Understanding*. doi: 10.1016/j.cviu.2003.06.004.

Auh, S. *et al.* (2007) 'Co-production and customer loyalty in financial services', *Journal of Retailing*. JAI, 83(3), pp. 359–370. doi: 10.1016/J.JRETAI.2007.03.001.

Awad, E. M. and Ghaziri, H. (2004) *Knowledge management*. Prentice Hall. Available at:

https://books.google.co.uk/books/about/Knowledge_Management.html?id=F4uCQgAA CAAJ (Accessed: 25 July 2018).

Badewi, A. (2016) 'The impact of project management (PM) and benefits management (BM) practices on project success: Towards developing a project benefits governance

framework', *International Journal of Project Management*. Pergamon, 34(4), pp. 761–778. doi: 10.1016/J.IJPROMAN.2015.05.005.

Badewi, A. *et al.* (2018) 'ERP benefits capability framework: orchestration theory perspective', *Business Process Management Journal*. Emerald Publishing Limited , 24(1), pp. 266–294. doi: 10.1108/BPMJ-11-2015-0162.

Baird, A. and Riggins, F. J. (2012) 'Planning and Sprinting: Use of a Hybrid Project Management Methodology within a CIS Capstone Course', *Journal of Information Systems Education*, 23(3), pp. 243–257. doi: 10.1002/spip.

Bampouras, T. M. *et al.* (2012) 'Performance analytic processes in elite sport practice: An exploratory investigation of the perspectives of a sport scientist, coach and athlete', *International Journal of Performance Analysis in Sport*. Routledge, 12(2), pp. 468–483. doi: 10.1080/24748668.2012.11868611.

Bandaru, S., Ng, A. H. C. and Deb, K. (2017) 'Data mining methods for knowledge discovery in multi-objective optimization: Part A - Survey', *Expert Systems with Applications*. doi: 10.1016/j.eswa.2016.10.015.

De Baranda, P. S. and Lopez-Riquelme, D. (2012) 'Analysis of corner kicks in relation to match status in the 2006 World Cup', *European Journal of Sport Science*, 12(2), pp. 121–129. doi: 10.1080/17461391.2010.551418.

Barbour, J. B., Treem, J. W. and Kolar, B. (2017) 'Analytics and expert collaboration: How individuals navigate relationships when working with organizational data', *Human Relations*, (c), p. 001872671771123. doi: 10.1177/0018726717711237.

Barney, J. (1991) 'Firm Resources and Sustained Competitive Advantage', *Journal of Management*. doi: 10.1177/014920639101700108.

Barreira, D., Garganta, J. J., *et al.* (2014) 'Ball recovery patterns as a performance indicator in elite soccer', *Proceedings of the Institution of Mechanical Engineers, Part P: Journal of Sports Engineering and Technology*, 228(1), pp. 61–72. doi: 10.1177/1754337113493083.

Barreira, D., Garganta, J., *et al.* (2014) 'Effects of ball recovery on top-level soccer attacking patterns of play', *Revista Brasileira de Cineantropometria & Desempenho Humano*, 16(1), pp. 36–46. doi: 10.5007/1980-0037.2014v16n1p36.

Barros, C. P. and Leach, S. (2006) 'Analyzing the Performance of the English F.A. Premier League With an Econometric Frontier Model', *Journal of Sports Economics*, 7(4), pp. 391–407. doi: 10.1177/1527002505276715.

Barros, C. P. and Leach, S. (2006) 'Performance evaluation of the English Premier Football League with data envelopment analysis', *Applied Economics*, 38(12), pp. 1449–1458. doi: 10.1080/00036840500396574.

Baškarada, S. and Koronios, A. (2013) 'Data, information, knowledge, wisdom (DIKW): A semiotic theoretical and empirical exploration of the hierarchy and its quality dimension', *Australasian Journal of Information Systems*, 18(1), pp. 5–24. doi: 10.3127/ajis.v18i1.748.

Bate, R. (1988) *Football chance: Tactics and strategy, Science and Football.* Available at:

https://books.google.com/books?hl=en&lr=&id=vCDBicFbsioC&oi=fnd&pg=PA293& ots=VDl4jMYTTH&sig=FETlmE-G7dwsHpKS5rXYIT3yk44 (Accessed: 31 July 2018).

BBC News (2017) 'Champions League: Firms pitch future football technology', *BBC News*, June. Available at: https://www.bbc.com/news/uk-wales-40122749 (Accessed: 30 September 2018).

Beck, K. et al. (2001) Manifesto for Agile Software Development. Available at: http://agilemanifesto.org/ (Accessed: 4 July 2018).

Beck, K. and Andres, C. (2004) 'Extreme Programming Explained: Embrace Change (2nd Edition)'.

Becker, J., Knackstedt, R. and Pöppelbuß, J. (2009a) 'Developing Maturity Models for IT Management', *Business & Information Systems Engineering*. SP Gabler Verlag, 1(3), pp. 213–222. doi: 10.1007/s12599-009-0044-5.

Becker, J., Knackstedt, R. and Pöppelbuß, J. (2009b) 'Developing Maturity Models for IT Management', *Business & Information Systems Engineering*. SP Gabler Verlag, 1(3), pp. 213–222. doi: 10.1007/s12599-009-0044-5.

Benbasat, I. *et al.* (1984) 'A critque of the stage hypothesis: theory and empirical evidence', *Communications of the ACM*, 1 May, pp. 476–485. doi: 10.1145/358189.358076.

Bendoly, E., Rosenzweig, E. D. and Stratman, J. K. (2007) 'Performance metric portfolios: A framework and empirical analysis', *Production and Operations Management*.

Berman, S. L., Down, J. and Hill, C. W. L. (2002) 'TACIT KNOWLEDGE AS A SOURCE OF COMPETITIVE ADVANTAGE IN THE NATIONAL BASKETBALL ASSOCIATION.', *Academy of Management Journal*. Academy of Management, 45(1), pp. 13–31. doi: 10.2307/3069282.

Bialkowski, A. *et al.* (2014) 'Win at Home and Draw Away ": Automatic Formation Analysis Highlighting the Differences in Home and Away Team Behaviors', in *8th annual MIT sloan sports analytics conference*. Available at: https://www.semanticscholar.org/paper/Win-at-Home-and-Draw-Away-"-%3A-Automatic-Formation-Bialkowski-

Lucey/c932441f8ecc9348e2dd54ec749984c8f4abf224 (Accessed: 25 August 2018).

Bloomfield, J., Polman, R. and O'Donoghue, P. (2004) 'The "Bloomfield Movement Classification": Motion Analysis of Individual Players in Dynamic Movement Sports', *International Journal of Performance Analysis in Sport*. Routledge, 4(2), pp. 20–31. doi: 10.1080/24748668.2004.11868300.

Bocij, P. et al. (2015) Business information systems: technology, development and management for the e-business.

Boddy, D., Boonstra, A. and Kennedy, G. (2005) *Managing information systems : an organisational perspective*. Financial Times Prentice Hall.

Boehm, B. (1996) 'Anchoring the software process', *IEEE Software*, 13(4), pp. 73–82. doi: 10.1109/52.526834.

Booroff, M., Nelson, L. and Potrac, P. (2016) 'A coach's political use of video-based feedback: a case study in elite-level academy soccer', *Journal of sports sciences*. Routledge, 34(2), pp. 116–124. doi: 10.1080/02640414.2015.1039464.

Bowen, L. *et al.* (2017) 'Accumulated workloads and the acute:chronic workload ratio relate to injury risk in elite youth football players', *British Journal of Sports Medicine*. doi: 10.1136/bjsports-2015-095820.

Bradley, P. S. *et al.* (2011) 'The effect of playing formation on high-intensity running and technical profiles in English FA Premier League soccer matches', *Journal of Sports Sciences*, 29(8), pp. 821–830. doi: 10.1080/02640414.2011.561868.

Bradley, P. S. *et al.* (2014) 'The influence of situational variables on ball possession in the English Premier League', *Journal of Sports Sciences*, 32(20), pp. 1867–1873. doi: 10.1080/02640414.2014.887850.

Bramer, M. A. (1999) Knowledge discovery and data mining. Iet.

Bray, S. R. and Brawley, L. R. (2002) 'Role Efficacy, Role Clarity, and Role Performance Effectiveness', *Small Group Research*. Sage PublicationsSage CA: Thousand Oaks, CA, 33(2), pp. 233–253. doi: 10.1177/104649640203300204.

Brooks, P., El-Gayar, O. and Sarnikar, S. (2015) 'A framework for developing a domain specific business intelligence maturity model: Application to healthcare', *International Journal of Information Management*. Pergamon, 35(3), pp. 337–345. doi: 10.1016/J.IJINFOMGT.2015.01.011.

De Bruin, T. *et al.* (2009) 'Understanding the Main Phases of Developing a Maturity Assessment Model', in. QUT ePrints. Available at: https://eprints.qut.edu.au/25152/ (Accessed: 15 July 2018).

Camerino, O. F. *et al.* (2012) 'Dynamics of the game in soccer: Detection of T-patterns', *European Journal of Sport Science*, 12(3), pp. 216–224. doi: 10.1080/17461391.2011.566362.

Canales, F. S. (2014) 'Automated Semantic Annotation of Football Games from TV Broadcast'.

Capgemini (2012) *Measuring Organizational Maturity in Predictive Analytics: the First Step to Enabling the Vision, Capgemini Worldwide.* Available at: https://www.capgemini.com/resources/measuring-organizational-maturity-in-predictiveanalytics-the-first-step-to-enabling-the-vision/# (Accessed: 27 September 2018).

Caralli, R. A., Knight, M. and Montgomery, A. (2012) *Maturity Models 101: A Primer for Applying Maturity Models to Smart Grid Security, Resilience, and Interoperability.* Available at: https://resources.sei.cmu.edu/library/asset-view.cfm?assetid=58916 (Accessed: 13 July 2018).

Carling, C. et al. (2005) Handbook of Soccer Match Analysis. doi: 10.4324/9780203448625.

Carling, C. *et al.* (2008) 'The Role of Motion Analysis in Elite Soccer', *Sports Medicine*, 38(10), pp. 839–862. doi: 10.2165/00007256-200838100-00004.

Carling, C. *et al.* (2013) 'Comment on "Performance analysis in football: A critical review and implications for future research", *Journal of sports sciences*. Routledge, 31(January 2014), pp. 37–41. doi: 10.1080/02640414.2013.807352.

Carling, C. *et al.* (2014) 'Squad management, injury and match performance in a professional soccer team over a championship-winning season.', *European journal of sport science*. Taylor & Francis, 0(0), pp. 1–10. doi: 10.1080/17461391.2014.955885.

Carling, C. and Dupont, G. (2011a) 'Are declines in physical performance associated with a reduction in skill-related performance during professional soccer match-play?', *Journal of Sports Sciences*. Routledge , 29(1), pp. 63–71. doi: 10.1080/02640414.2010.521945.

Carling, C. and Dupont, G. (2011b) 'Are declines in physical performance associated with a reduction in skill-related performance during professional soccer match-play?', *Journal of Sports Sciences*, 29(1), pp. 63–71. doi: 10.1080/02640414.2010.521945.

Carling, C., Gall, F. L. and Reilly, T. P. (2010) 'Effects of Physical Efforts on Injury in Elite Soccer', *International Journal of Sports Medicine*, 31(03), pp. 180–185. doi: 10.1055/s-0029-1241212.

Carling, C., Le Gall, F. and Dupont, G. (2012) 'Analysis of repeated high-intensity running performance in professional soccer', *Journal of Sports Sciences*, 30(4), pp. 325–336. doi: 10.1080/02640414.2011.652655.

Carling, C., Williams, A. M. (A. M. and Reilly, T. (2005) *Handbook of soccer match analysis : a systematic approach to improving performance*. Routledge. Available at: https://books.google.co.uk/books/about/Handbook_of_Soccer_Match_Analysis.html?id =A5ueI0tZuDYC (Accessed: 27 July 2018).

Casamichana, D. *et al.* (2014) 'Effect of number of touches and exercise duration on the kinematic profile and heart rate response during small-sided games in soccer', *Journal of Human Kinetics*, 41(1), pp. 113–123. doi: 10.2478/hukin-2014-0039.

Castellano, J., Casamichana, D. and Lago, C. (2012) 'The Use of Match Statistics that Discriminate Between Successful and Unsuccessful Soccer Teams', *Journal of Human Kinetics*, 31(1), pp. 139–147. doi: 10.2478/v10078-012-0015-7.

Castro-Martinez, M. P. and Jackson, P. R. (2015) 'Collaborative value co-creation in community sports trusts at football clubs', *Corporate Governance: The international journal of business in society*. Edited by D. Tim Breitbarth, Dr Stefan Walzel, D. Emerald Group Publishing Limited , 15(2), pp. 229–242. doi: 10.1108/CG-05-2014-0066.

Chae, H.-C., Koh, C. E. and Park, K. O. (2018) 'Information technology capability and firm performance: Role of industry', *Information & Management*. North-Holland, 55(5), pp. 525–546. doi: 10.1016/J.IM.2017.10.001.

Chambers, R. *et al.* (2015) 'The Use of Wearable Microsensors to Quantify Sport-Specific Movements', *Sports Medicine*. Springer International Publishing, 45(7), pp. 1065–1081. doi: 10.1007/s40279-015-0332-9.

Charmaz, K. (2006) Constructing grounded theory: a practical guide through qualitative analysis, SAGE Publications Ltd. doi: 10.1016/j.lisr.2007.11.003.

Du Chatenier, E. *et al.* (2009) 'The Challenges of Collaborative Knowledge Creation in Open Innovation Teams', *Human Resource Development Review*. SAGE PublicationsSage CA: Los Angeles, CA, 8(3), pp. 350–381. doi: 10.1177/1534484309338265.

Chen, L. and Nath, R. (2018) 'Business analytics maturity of firms: an examination of the relationships between managerial perception of IT, business analytics maturity and success', *Information Systems Management*. Taylor & Francis, 35(1), pp. 62–77. doi: 10.1080/10580530.2017.1416948.

Chen, S. Y. and Liu, X. (2009) 'The contribution of data mining to information science', *Journal of Information Science*, 30(6), pp. 550–558. doi: 10.1177/0165551504047928.

Chi, M. T. H., Slotta, J. D. and De Leeuw, N. (1994) 'From things to processes: A theory of conceptual change for learning science concepts', *Learning and Instruction*. Pergamon, 4(1), pp. 27–43. doi: 10.1016/0959-4752(94)90017-5.

Churchill, N. C., Kempster, J. H. and Uretsky, M. (1969) 'Computer based information systems for management: A survey.', *National Association of Accountants*.

Clemente, F. M. *et al.* (2013) 'Measuring Tactical Behaviour Using Technological Metrics: Case Study of a Football Game.', *International Journal of Sports Science & Coaching*, 8, pp. 723–740.

Clemente, F. M. et al. (2014) 'Developing a football tactical metric to estimate the sectorial lines: A case study', Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics), 8579 LNCS(PART 1), pp. 743–753. doi: 10.1007/978-3-319-09144-0_51.

CMMI (2010) 'CMMI for Services, Version 1.3', *Sei*. doi: CMU/SEI-2010-TR-033 ESC-TR-2010-033. CMMI Institute (2018) *Capability Maturity Model Integration*. Available at: https://cmmiinstitute.com/cmmi (Accessed: 9 July 2018).

CMMI Product Team (2010) *CMMI for Acquisition*, *SEI*. doi: CMU/SEI-2010-TR-033 ESC-TR-2010-033.

CMMI Product Team (2018) *CMMI Development*, *CMMI Institute*. Available at: https://cmmiinstitute.com/products/cmmi/dev (Accessed: 9 July 2018).

Cobb, N. M., Unnithan, V. and McRobert, A. P. (2018) 'The validity, objectivity, and reliability of a soccer-specific behaviour measurement tool', *Science and Medicine in Football*. Routledge, 00(00), pp. 1–7. doi: 10.1080/24733938.2017.1423176.

Cochrane, D. (2011) 'The sports performance application of vibration exercise for warmup, flexibility and sprint speed', *European Journal of Sport Science*, (June 2015), pp. 1– 16. doi: 10.1080/17461391.2011.606837.

Cohn, M. (2004) User stories applied : for agile software development. Addison-Wesley. Available at:

https://books.google.co.uk/books/about/User_Stories_Applied.html?id=SvIwuX4SVigC &redir_esc=y (Accessed: 5 July 2018).

Collet, C. (2013a) 'The possession game? A comparative analysis of ball retention and team success in European and international football, 2007–2010', *Journal of Sports Sciences*. Routledge, 31(2), pp. 123–136. doi: 10.1080/02640414.2012.727455.

Collet, C. (2013b) 'The possession game? A comparative analysis of ball retention and team success in European and international football, 2007–2010', *Journal of Sports Sciences*. doi: 10.1080/02640414.2012.727455.

Collet, C. (2013c) 'The possession game? A comparative analysis of ball retention and team success in European and international football, 2007–2010', *Journal of Sports Sciences*, 31(2), pp. 123–136. doi: 10.1080/02640414.2012.727455.

Collier, K. and Highsmith, J. (2010) 'Agile Data Warehousing: Incorporating Agile Principles', *Business Intelligence Journal*.

Collyer, S. *et al.* (2010) 'Aim, fire, aim-Project planning styles in dynamic environments', *Project Management Journal.* doi: 10.1002/pmj.20199.

Conboy, K. (2009) 'Agility from first principles: Reconstructing the concept of agility in information systems development', *Information Systems Research*. doi: 10.1287/isre.1090.0236.

Conforto, E. C. *et al.* (2016) 'The agility construct on project management theory', *International Journal of Project Management*. Pergamon, 34(4), pp. 660–674. doi: 10.1016/J.IJPROMAN.2016.01.007.

Constantinou, A. and Fenton, N. (2017) 'Towards smart-data: Improving predictive accuracy in long-term football team performance', *Knowledge-Based Systems*. Elsevier B.V., 124, pp. 93–104. doi: 10.1016/j.knosys.2017.03.005.

Cosic, R., Shanks, G. and Maynard, S. (2012) 'Towards a business analytics capability maturity model', ACIS 2012: Location, location, location: Proceedings of the 23rd Australasian Conference on Information Systems 2012.

Costa, M. F. da *et al.* (2018) 'Perceived competitive advantage of soccer clubs: a study based on the resource-based view', *RAUSP Management Journal*. Elsevier, 53(1), pp. 23–34. doi: 10.1016/J.RAUSPM.2016.08.001.

Coyle, S. *et al.* (2009) 'Textile-Based Wearable Sensors for Assisting Sports Performance', in 2009 Sixth International Workshop on Wearable and Implantable Body Sensor Networks. IEEE, pp. 307–311. doi: 10.1109/BSN.2009.57.

Creswell, J. W. and Miller, D. L. (2000) 'Determining Validity in Qualitative Inquiry', *Theory Into Practice*. Lawrence Erlbaum Associates, Inc., 39(3), pp. 124–130. doi: 10.1207/s15430421tip3903_2.

Cushion, C. (2001) 'THE COACHING PROCESS IN PROFESSIONAL YOUTH FOOTBALL: AN ETHNOGRAPHY OF PRACTICE A thesis submitted for the degree of Doctor of Philosophy awarded by', *Transition*, (December), pp. 1–303.

Cushion, C. J., Armour, K. M. and Jones, R. L. (2006) 'Locating the coaching process in practice: models "for" and "of" coaching', *Physical Education & Sport Pedagogy*. Routledge, 11(1), pp. 83–99. doi: 10.1080/17408980500466995.

Cushion, C. and Lyle, J. (2016) 'Conceptualising Sport-Coaching: Some Key Questions and Issues', in *Coaching-Praxisfelder*. Wiesbaden: Springer Fachmedien Wiesbaden, pp. 117–134. doi: 10.1007/978-3-658-10171-8_7.

Dalsgaard, C. and Sterrett, R. (2014) 'White paper on smart textile garments and devices: a market overview of smart textile wearable technologies', *Market Opportunities for Smart Textiles, Ohmatex, Denmark.*

Davenport, T. H. and Harris, J. G. (2007) Competing on analytics : the new science ofwinning.HarvardBusinessSchoolPress.Availableat:

https://books.google.co.uk/books?hl=en&lr=&id=n7Gp7Q84hcsC&oi=fnd&pg=PR4&d q=Davenport+%26+Harris+2007&ots=9uVJKN-EKI&sig=kz4aL_15jONig-

a0bv_kvSD3hpg&redir_esc=y#v=onepage&q=Davenport %26 Harris 2007&f=false (Accessed: 13 August 2018).

Dawson, C. W. (2009) Projects in Computing and Information Systems, Information Systems Journal.

Dawson, P. and Dobson, S. (2002) 'Managerial efficiency and human capital: an application to English association football', *Managerial and Decision Economics*. Wiley-Blackwell, 23(8), pp. 471–486. doi: 10.1002/mde.1098.

Dellaserra, C. L., Gao, Y. and Ransdell, L. (2014) 'Use of integrated technology in team sports: a review of opportunities, challenges, and future directions for athletes.', *Journal of strength and conditioning research / National Strength & Conditioning Association*, 28(2), pp. 556–73. doi: 10.1519/JSC.0b013e3182a952fb.

DeLone, W. H. and McLean, E. R. (1992) 'Information Systems Success: The Quest for the Dependent Variable', *Information Systems Research*, 3(1), pp. 60–95. doi: 10.1287/isre.3.1.60.

Dimitrijević, S., Jovanovic, J. and Devedžić, V. (2015) 'A comparative study of software tools for user story management', *Information and Software Technology*, 57(1), pp. 352–368. doi: 10.1016/j.infsof.2014.05.012.

Ding, Y. and Fan, G. (2006) 'Camera view-based american football video analysis', *ISM* 2006 - 8th IEEE International Symposium on Multimedia, pp. 317–322. doi: 10.1109/ISM.2006.42.

Dingsøyr, T. *et al.* (2012) 'A decade of agile methodologies: Towards explaining agile software development', *Journal of Systems and Software*, 85(6), pp. 1213–1221. doi: 10.1016/j.jss.2012.02.033.

Doyle, P. (2007) 'On second thoughts ... Egil Olsen', *The Guardian | Sport* . Available at: https://www.theguardian.com/sport/blog/2007/apr/04/onsecondthoughtsegilolsen (Accessed: 31 July 2018).

Drazin, R. and de Ven, A. H. Van (1985) 'Alternative Forms of Fit in Contingency Theory', *Administrative Science Quarterly*. Sage Publications, Inc.Johnson Graduate School of Management, Cornell University, 30(4), p. 514. doi: 10.2307/2392695. Drury, M., Conboy, K. and Power, K. (2012) 'Obstacles to decision making in Agile software development teams', *Journal of Systems and Software*. doi: 10.1016/j.jss.2012.01.058.

Duffield, S. and Whitty, S. J. (2015) 'Developing a systemic lessons learned knowledge model for organisational learning through projects', *International Journal of Project Management*. Pergamon, 33(2), pp. 311–324. doi: 10.1016/J.IJPROMAN.2014.07.004.

Düking, P., Holmberg, H.-C. and Sperlich, B. (2017) 'Instant Biofeedback Provided by Wearable Sensor Technology Can Help to Optimize Exercise and Prevent Injury and Overuse', *Frontiers in Physiology*, 8, p. 167. doi: 10.3389/fphys.2017.00167.

Dwyer, D. B. and Gabbett, T. J. (2012) 'Global Positioning System Data Analysis: Velocity Ranges and a New Definition of Sprinting for Field Sport Athletes', *Journal of Strength and Conditioning Research*, 26(3), pp. 818–824. doi: 10.1519/JSC.0b013e3182276555.

Eckerson, W. (2004) 'Gauge Your Data Warehouse Maturity', DM Review, November,pp.P34-51.Availableat:

http://web.a.ebscohost.com/ehost/detail/detail?vid=0&sid=be6d606e-61ea-463f-a4b7-

b94fddf9809e%40sessionmgr4007&bdata=JnNpdGU9ZWhvc3QtbGl2ZQ%3D%3D#A N=14964672&db=bth (Accessed: 27 September 2018).

Eckerson, W. W. (2009) 'No TitleEckerson, Wayne W. "TDWI's Business Intelligence Maturity Model.', *The Data Warehousing Institute*. Chatsworth.

Ekin, A., Tekalp, A. M. and Mehrotra, R. (2003) 'Automatic soccer video analysis and summarization', *IEEE Transactions on Image Processing*, 12(7), pp. 796–807. doi: 10.1109/TIP.2003.812758.

El-Gayar, O. *et al.* (2011) 'Toward a Maturity Model for DSS Development Processes.', in *AMCIS*. Available at: https://pdfs.semanticscholar.org/573c/340a3b646fc70e8d95f1e3aebb2d98c5e5dc.pdf (Accessed: 28 September 2018).

El-Hodiri, M. and Quirk, J. (1971) 'An Economic Model of a Professional Sports League', *Journal of Political Economy*. The University of Chicago Press, 79(6), pp. 1302–1319. doi: 10.2307/1830103.

Eppler, M. J. (2014) 'Software Development Process Models: A Technique for Evaluation and Decision-Making', *Knowledge and Process Management*, 1(1), pp. 42–53. doi: 10.1002/kpm.

Eys, M. A. *et al.* (2005) 'The relationship between role ambiguity and intention to return the following season', *Journal of Applied Sport Psychology*, 17(3), pp. 255–261. doi: 10.1080/10413200591010148.

Farin, D. (2005) 'Current and Emerging Topics in Sports Video Processing', 2005 IEEE International Conference on Multimedia and Expo, (August 2005), pp. 526–529. doi: 10.1109/ICME.2005.1521476.

Fayyad, U., Piatetsky-Shapiro, G. and Smyth, P. (1996) 'Knowledge discovery and data mining: Towards a unifying framework', *Knowledge Discovery and Data Mining*, pp. 82–88. Available at: http://www.aaai.org/Papers/KDD/1996/KDD96-014.pdf.

Felin, T. and Hesterly, W. S. (2007) 'The knowledge-based view, nested heterogeneity, and new value creation: Philosophical considerations on the locus of knowledge', *Academy of Management Review*. Academy of Management, pp. 195–218. doi: 10.5465/AMR.2007.23464020.

Fernandez-Echeverria, C. *et al.* (2017) 'Match analysis within the coaching process: a critical tool to improve coach efficacy', *International Journal of Performance Analysis in Sport.* Routledge, 17(1–2), pp. 149–163. doi: 10.1080/24748668.2017.1304073.

Fernandez-Navarro, J. *et al.* (2016) 'Attacking and defensive styles of play in soccer: analysis of Spanish and English elite teams', *Journal of Sports Sciences*, 34(24), pp. 2195–2204. doi: 10.1080/02640414.2016.1169309.

Fernandez-Navarro, J. *et al.* (2018a) 'Influence of contextual variables on styles of play in soccer', *International Journal of Performance Analysis in Sport*. Routledge, 18(3). doi: 10.1080/24748668.2018.1479925.

Fernandez-Navarro, J. *et al.* (2018b) 'Influence of contextual variables on styles of play in soccer', *International Journal of Performance Analysis in Sport*. Routledge, 18(3), pp. 423–436. doi: 10.1080/24748668.2018.1479925.

Flanagan, J. C. (1954) 'The critical incident technique', *Psychological Bulletin*. doi: 10.1037/h0061470.

Fletcher, D. and Arnold, R. (2011) 'A Qualitative Study of Performance Leadership and Management in Elite Sport', *Journal of Applied Sport Psychology*. Taylor & Francis Group , 23(2), pp. 223–242. doi: 10.1080/10413200.2011.559184.

Franks, I. M. and Miller, G. (1986) 'Eyewitness Testimony in Sport', *Journal of Sport Behaviour*.

Franks, I. M. and Miller, G. (1991) 'Training coaches to observe and remember', *Journal of Sports Sciences*, 9(3), pp. 285–297. doi: 10.1080/02640419108729890.

Fraser, P., Moultrie, J. and Gregory, M. (2003) 'The use of maturity models/grids as a tool in assessing product development capability', in *IEEE International Engineering Management Conference*. IEEE, pp. 244–249. doi: 10.1109/IEMC.2002.1038431.

Frencken, W. G. P., Lemmink, K. A. P. M. and Delleman, N. J. (2010) 'Soccer-specific accuracy and validity of the local position measurement (LPM) system', *Journal of Science and Medicine in Sport*, 13(6), pp. 641–645. doi: 10.1016/j.jsams.2010.04.003.

Fuss, F. K., Düking, P. and Weizman, Y. (2018) 'Discovery of a sweet spot on the foot with a smart wearable soccer boot sensor that maximizes the chances of scoring a curved kick in soccer', *Frontiers in Physiology*. doi: 10.3389/fphys.2018.00063.

Garcia-del-Barrio, P. and Szymanski, S. (2009) 'Goal! Profit Maximization Versus Win Maximization in Soccer', *Review of Industrial Organization*. Springer US, 34(1), pp. 45–68. doi: 10.1007/s11151-009-9203-6.

García-peñalvo, F. J. and Conde-gonzález, M. A. (2017) 'Statistical Implicative Analysis Approximation to KDD and Data Mining':, (May), pp. 70–77.

Garganta, J. (2009a) 'Match analysts must be able to check the relevance and descriptive power of performance indicators and to distinguish the core features of the game.', *Rev Port Cien Desp*, 9(1), pp. 81–89. Available at: http://www.fade.up.pt/rpcd/_arquivo/artigos_soltos/vol.9_nr.1/2.01.pdf.

Garganta, J. (2009b) 'Trends of tactical performance analysis in team sports: bridging the gap between research, training and competition.', *Revista Portuguesa de Ciências do Desporto*, 9(1), pp. 81–89. doi: 10.1080/10400410701397420.

Gaudino, P. *et al.* (2013) 'Monitoring Training in Elite Soccer Players: Systematic Bias between Running Speed and Metabolic Power Data', *International Journal of Sports Medicine*, 34(11), pp. 963–968. doi: 10.1055/s-0033-1337943.

Gibson Moreira, P. *et al.* (2015) 'Relationship between tactical and technical performance in youth soccer players', *Revista Brasileira de Cineantropometria e Desempenho Humano*, 17(2), pp. 136–144. doi: 10.5007/1980-0037.2015v17n2p136.

Gottschalk, P. (2009) 'Maturity levels for interoperability in digital government', *Government Information Quarterly*. JAI, 26(1), pp. 75–81. doi: 10.1016/J.GIQ.2008.03.003.

GOV.UK (2018) *Digital Service Standard - Service Manual - GOV.UK*. Available at: https://www.gov.uk/service-manual/service-standard#criterion-4 (Accessed: 3 July 2018).

Grant, R. M. (1991) 'The Resource-Based Theory of Competitive Advantage: Implications for Strategy Formulation', *California Management Review*. SAGE PublicationsSage CA: Los Angeles, CA, 33(3), pp. 114–135. doi: 10.2307/41166664.

Grant, M. J. and Booth, A. (2009), A typology of reviews: an analysis of 14 review types and associated methodologies. Health Information & Libraries Journal, 26: 91-108. doi:<u>10.1111/j.1471-1842.2009.00848.x</u>

Gregor, S. and Hevner, A. R. (2013) 'Positioning and Presenting Design Science Research for Maximum Impact', *MIS Quarterly*. Society for Information Management and The Management Information Systems Research Center, 37(2), pp. 337–355. doi: 10.25300/MISQ/2013/37.2.01.

Gregson, W. *et al.* (2010) 'Match-to-Match Variability of High-Speed Activities in Premier League Soccer', *International Journal of Sports Medicine*, 31(04), pp. 237–242. doi: 10.1055/s-0030-1247546.

Groom, N. R. (2012) TOWARDS AN UNDERSTANDING OF THE USE OF VIDEO-BASED PERFORMANCE ANALYSIS IN THE COACHING PROCESS.

Groom, R. and Cushion, C. J. (2004) 'COACHES PERCEPTIONS OF THE USE OF VIDEO ANALYSIS : A CASE STUDY', *Insight: The FA Coaches Association Journal*, 7(3), pp. 56–58. Available at: https://www.academia.edu/3752589/Groom_and_Cushion_2004_Coaches_perceptions_ of_the_use_of_video_analysis_A_case_study_-_FA_Insight?auto=download (Accessed: 15 July 2018).

Groom, R., Cushion, C. J. and Nelson, L. J. (2012) 'Analysing coach-athlete "talk in interaction" within the delivery of video-based performance feedback in elite youth

soccer', *Qualitative Research in Sport, Exercise and Health.* Routledge, 4(3), pp. 439–458. doi: 10.1080/2159676X.2012.693525.

Groom, R., Cushion, C. and Nelson, L. (2011) 'The Delivery of Video-Based Performance Analysis by England Youth Soccer Coaches: Towards a Grounded Theory', *Journal of Applied Sport Psychology*. Taylor & Francis Group , 23(1), pp. 16–32. doi: 10.1080/10413200.2010.511422.

GSA (2018) General Sport Authority, GSA. Available at: https://www.gsa.gov.sa/en/Pages/default.aspx (Accessed: 14 August 2018).

Gullo, F. (2015) 'From patterns in data to knowledge discovery: What data mining can do', *Physics Procedia*. Elsevier B.V., 62, pp. 18–22. doi: 10.1016/j.phpro.2015.02.005.

Gyarmati, L. (2016) 'QPass : a Merit-based Evaluation of Soccer Passes Which player is delivering passes into critical parts of the Partitioning the field In case of soccer analytics , the de facto method of aggre- Field value Using the partitioning , we compute the field val', pp. 0–3.

Haghighat, M., Rastegari, H. and Nourafza, N. (2013) 'A Review of Data Mining Techniques for Result Prediction in Sports', *Advances in Computer Science*, 2(5), pp. 7–12.

Halper, F. and Stodder, D. (2014) *TDWI Analytics Maturity Model Guide*. Available at: https://tdwi.org/whitepapers/2014/10/tdwi-analytics-maturity-model-guide.aspx (Accessed: 27 September 2018).

Hamil, S. and Walters, G. (2010) 'Financial performance in English professional football: "an inconvenient truth", *Soccer & Society*, 11(4), pp. 354–372. doi: 10.1080/14660971003780214.

Hammer, M. (2007) 'The Process Audit', *Harvard Business Review*. Available at: https://hbr.org/2007/04/the-process-audit (Accessed: 27 September 2018).

Harriott, J. (2013) '7 pillars for successful analytics implementation: A leader's guide to incorporating big data across the organization.', *Marketing Insights*, pp. 35–40.

Den Hartigh, R. J. R. *et al.* (2018) 'Selection procedures in sports: Improving predictions of athletes' future performance', *European Journal of Sport Science*. Routledge, pp. 1–8. doi: 10.1080/17461391.2018.1480662.

Hatton, L. (2013) Data analysis - shaping football [and Bolton] as they enter newterritory,SBNation.Availableat:

https://lionofviennasuite.sbnation.com/2013/6/10/4414800/data-analysis-shaping-football-and-bolton-as-they-enter-new-territory (Accessed: 26 July 2018).

Hazır, Ö. (2015) 'A review of analytical models, approaches and decision support tools in project monitoring and control', *International Journal of Project Management*. Pergamon, 33(4), pp. 808–815. doi: 10.1016/J.IJPROMAN.2014.09.005.

Hegland, M. (2001) 'Data mining techniques', *Acta Numerica*, 10, pp. 313–355. doi: 10.1017/S0962492901000058.

Heldman, K. (2011) PMP : project management professional exam : study guide. 9th edn, Sybex serious skills. 9th edn. John Wiley & Sons, Incorporated.

Helfat, C. E. and Peteraf, M. A. (2003) 'The dynamic resource-based view: capability lifecycles', *Strategic Management Journal*. Wiley-Blackwell, 24(10), pp. 997–1010. doi: 10.1002/smj.332.

Hevner, A. and Chatterjee, S. (2010) 'Design Research in Information Systems', 22, pp. 9–23. doi: 10.1007/978-1-4419-5653-8.

Hevner, A. R. (2007) 'A Three Cycle View of Design Science Research', *Scandinavian Journal of Information Systems*, 19(2), p. 4. doi: http://aisel.aisnet.org/sjis/vol19/iss2/4.

Hiscock, D. *et al.* (2012) 'Game movements and player performance in the Australian Football League', *International Journal of Performance Analysis in Sport*, 12(3), pp. 531–545. doi: 10.1080/24748668.2012.11868617.

Ho, A. (2017) Beyond the Dataset: Understanding Sociotechnical Aspects of the Knowledge Discovery Process Among Modern Data Professionals, UWSpace. University of Waterloo. Available at: https://uwspace.uwaterloo.ca/handle/10012/11835 (Accessed: 11 August 2018).

Hobbs, B. and Petit, Y. (2017) *Agile Methods on Large Projects in Large Organizations*, *Project Management Journal*. SAGE PublicationsSage CA: Los Angeles, CA. doi: 10.1177/875697281704800301.

Hoch, T. *et al.* (2017) 'A knowledge discovery framework for the assessment of tactical behaviour in soccer based on spatiotemporal data', *Mathematical and Computer Modelling of Dynamical Systems*, 23(4), pp. 384–398. doi: 10.1080/13873954.2017.1336634.

Hodges, N. J. and Franks, I. M. (2002) 'Modelling coaching practice: the role of instruction and demonstration', *Journal of Sports Sciences*. Taylor & Francis , 20(10), pp. 793–811. doi: 10.1080/026404102320675648.

Hoernig, M. et al. (2016) 'Detection of individual ball possession in soccer', Advances in Intelligent Systems and Computing, 392, pp. 103–107. doi: 10.1007/978-3-319-24560-7_13.

Holcomb, T. R., Holmes Jr., R. M. and Connelly, B. L. (2009) 'Making the most of what you have: managerial ability as a source of resource value creation', *Strategic Management Journal*. Wiley-Blackwell, 30(5), pp. 457–485. doi: 10.1002/smj.747.

Horton, M. *et al.* (2014) 'Classification of Passes in Football Matches using Spatiotemporal Data'. Available at: http://arxiv.org/abs/1407.5093.

Huang, K.-Y. Y. and Chang, W.-L. L. (2010) 'A neural network method for prediction of 2006 World Cup Football Game', in *Neural Networks (IJCNN), The 2010 International Joint Conference*, pp. 1–8. doi: 10.1109/IJCNN.2010.5596458.

Hughes, E. (Mike) and Franks, I. M. (2004) NOTATIONAL ANALYSIS OF SPORT : SYSTEMS FOR BETTER COACHING AND PERFORMANCE IN SPORT, Routledge. Routledge. doi: 10.4324/9780203641958.

Hughes, M. (2004) 'Notational analysis – a mathematical perspective. Mike Hughes, CPA, UWIC, Cyncoed, Cardiff, CF26 2XD.', *International Journal of Performance Analysis in Sport*, 4, pp. 97–139.

Hughes, M. *et al.* (2012) 'Moneyball and soccer - An analysis of the key performance indicators of elite male soccer players by position', *Journal of Human Sport and Exercise*, 7(SPECIALISSUE.2), pp. 402–412. doi: 10.4100/jhse.2012.72.06.

Hughes, M. D. and Bartlett, R. M. (2002) 'The use of performance indicators in performance analysis', *Journal of Sports Sciences*, 20(10), pp. 739–754. doi: 10.1080/026404102320675602.

Hughes, M. and Franks, I. (2005a) 'Analysis of passing sequences, shots and goals in soccer', *Journal of Sports Sciences*, 23(5), pp. 509–514. doi: 10.1080/02640410410001716779.

Hughes, M. and Franks, I. (2005b) 'Analysis of passing sequences, shots and goals in soccer', *Journal of Sports Sciences*, 23(5), pp. 509–514. doi: 10.1080/02640410410001716779.

Hughes, M. and Franks, I. M. (2007) The essentials of performance analysis: An introduction. Routledge.

Humana, F. D. M. (2011) 'Decision making behaviour in team sports: informational constraints and the dynamics of interpersonal coordination in rugby union'.

Ibáñez, S. J. *et al.* (2018) 'The impact of scoring first on match outcome in women's professional football', *International Journal of Performance Analysis in Sport*. Routledge, 18(2), pp. 318–326. doi: 10.1080/24748668.2018.1475197.

Inayat, I. *et al.* (2015) 'A systematic literature review on agile requirements engineering practices and challenges', *Computers in Human Behavior*. Elsevier Ltd, 51, pp. 915–929. doi: 10.1016/j.chb.2014.10.046.

Infoholic Research (2016) *Worldwide Sports Analytics Market* (2016-2022) / *Analytical Research Cognizance*. Available at: http://www.arcognizance.com/report/worldwide-sports-analytics-market-2016-2022 (Accessed: 15 July 2018).

Ingle, S. (2013) The Numbers Game: Why Everything You Know About Football IsWrong,TheGuardian.Availablehttps://www.theguardian.com/books/2013/may/24/numbers-game-everything-football-wrong (Accessed: 16 July 2018).

InStat (2018) *InStat, InStat.* Available at: http://instatsport.com/en/ (Accessed: 2 August 2018).

Iversen, J., Nielsen, P. A. and Norbjerg, J. (1999) 'Situated assessment of problems in software development', *ACM SIGMIS Database*. doi: 10.1145/383371.383376.

Ives, J. C., Straub, W. F. and Shelley, G. A. (2002) 'Enhancing Athletic Performance Using Digital Video in Consulting', *Journal of Applied Sport Psychology*, 14(3), pp. 237–245. doi: 10.1080/10413200290103527.

Jacklin, P. B. (2005) 'Temporal changes in home advantage in English football since the Second World War: What explains improved away performance?', *Journal of Sports Sciences*. Taylor & Francis Ltd , 23(7), pp. 669–679. doi: 10.1080/02640410400021948. Jagadish, H. V. *et al.* (2014) 'Big data and its technical challenges', *Communications of the ACM*, 57(7), pp. 86–94. doi: 10.1145/2611567.

James, N. (2006) 'Notational analysis in soccer: past, present and future.', *International Journal of Performance Analysis in Sport*. Routledge, 6(2), pp. 67–81. doi: 10.1080/24748668.2006.11868373.

Jashapara, A. (2011) *Knowledge management : an integrated approach*. Pearson/Financial Times/Prentice Hall.

Jenner, S. and APMG International. (2014) *Managing benefits : optimizing the return from investments*. 2nd ed edition. Stationery Office.

Jessup, L. M. and Valacich, J. S. (2003) Information systems today. Prentice Hall. Available at:

https://books.google.co.uk/books/about/Information_Systems_Today.html?id=RV5GA AAAYAAJ&redir_esc=y (Accessed: 25 July 2018).

John Moores, L. and Reilly Building, T. (2010) 'Match-to-Match Variability of High-Speed Activities ...', *Int J Sports Med*, 31, pp. 237–242. doi: 10.1055/s-0030-1247546.

Jones, R. L., Armour, K. M. and Potrac, P. (2004) *Sports Coaching Cultures: From Practice to Theory*, *Routledge*. Routledge. Available at: https://www.routledge.com/Sports-Coaching-Cultures-From-Practice-to-

Theory/Armour-Jones-Potrac/p/book/9780203390955 (Accessed: 13 August 2018).

Jones, R. L., Harris, R. and Miles, A. (2009) 'Mentoring in sports coaching: a review of the literature', *Physical Education & Sport Pedagogy*. Routledge , 14(3), pp. 267–284. doi: 10.1080/17408980801976569.

Kalambe, Y. S., Pratiba, D. and Shah, P. (2015) 'Big Data Mining Tools for Unstructured Data: a Review', *Ijitr*, 3(2), pp. 2012–2017. Available at: http://www.ijitr.com/index.php/ojs/article/view/610.

Kamble, P. R., Keskar, A. G. and Bhurchandi, K. M. (2017) 'Ball tracking in sports: a survey', *Artificial Intelligence Review*. Springer Netherlands, pp. 1–51. doi: 10.1007/s10462-017-9582-2.

Kaplan, R. S. and Norton, D. P. (1996) 'The Balanced Scorecard Translating Strategy In Action (Kaplan & Norton, 1996, Harvard Business School Press).pdf', *Proceedings of the IEEE*. doi: 10.1109/JPROC.1997.628729.

Kelly, S. (2008) 'Understanding the Role of the Football Manager in Britain and Ireland: A Weberian Approach', *European Sport Management Quarterly*, 8(4), pp. 399–419. doi: 10.1080/16184740802461652.

Kempton, T. *et al.* (2015) 'Match-to-match variation in physical activity and technical skill measures in professional Australian Football', *Journal of Science and Medicine in Sport*. Sports Medicine Australia, 18(1), pp. 109–113. doi: 10.1016/j.jsams.2013.12.006.

Kerr, J. L. and Jackofsky, E. F. (1989) 'Aligning managers with strategies: Management development versus selection', *Strategic Management Journal*. Wiley-Blackwell, 10(S1), pp. 157–170. doi: 10.1002/smj.4250100712.

Khajah, M., Lindsey, R. V. and Mozer, M. C. (2016) 'How deep is knowledge tracing?' Available at: http://arxiv.org/abs/1604.02416.

Kilpatrick, D. (2018) 'Steve McClaren took QPR job without knowing club's full financial position', *London Evening Standard*. Available at: https://www.standard.co.uk/sport/football/steve-mcclaren-took-qpr-job-without-

knowing-clubs-full-financial-position-a3845776.html (Accessed: 12 August 2018).

Kim, D. and Grant, G. (2010) 'E-government maturity model using the capability maturity model integration', *Journal of Systems and Information Technology*. Emerald Group Publishing Limited, 12(3), pp. 230–244. doi: 10.1108/13287261011070858.

King, J. L. and Kraemer, K. L. (1984a) 'Evolution and organizational information systems: an assessment of Nolan's stage model', *Communications of the ACM*, 27(5), pp. 466–475. doi: 10.1145/358189.358074.

King, J. L. and Kraemer, K. L. (1984b) 'Evolution and organizational information systems: an assessment of Nolan's stage model', *Communications of the ACM*. ACM, 27(5), pp. 466–475. doi: 10.1145/358189.358074.

King, N. and Horrocks, C. (2010) *Interviews in qualitative research*. SAGE. Available at: https://vufind.lboro.ac.uk/Record/490854 (Accessed: 21 August 2018).

Kite, C. S. and Nevill, A. (2017) 'The Predictors and Determinants of Inter-Seasonal Success in a Professional Soccer Team.', *Journal of human kinetics*. De Gruyter Open, 58, pp. 157–167. doi: 10.1515/hukin-2017-0084.

Koehn, S. and Morris, T. (2012) 'The effect of performance context and skill level on the frequency of flow experiences', *European Journal of Sport Science*, 14(sup1), pp. S478–S486. doi: 10.1080/17461391.2012.718364.

Kotter, J. (1995) 'Leading Change: Why Transofrmation Efforts Fail', *Harvard Business Review*. OTTAWA ONT.: CANADA COMMUNICATION Group, 73(2), pp. 59–67. Available at: https://www.worldcat.org/title/leading-change-why-transofrmation-efforts-fail/oclc/932948552 (Accessed: 2 August 2018).

Kuper, S. and Szymanski, S. (2018) Soccernomics : why England lose, why Germany, Spain and France win, and why one day the rest of the world will finally catch up, HarperCollins.

Kuznets, S. (1966) 'Economic growth and structure: selected essays'.

Kwasnik, B. H. (1999) 'The role of classification in knowledge representation and discovery', *Library Trends*, 48(1), pp. 22–47. doi: 10.1086/250095.

Lago-Ballesteros, J. and Lago-Peñas, C. (2010) 'Performance in Team Sports: Identifying the Keys to Success in Soccer', *Journal of Human Kinetics*, 25(1), pp. 85–91. doi: 10.2478/v10078-010-0035-0.

Lago-Peñas, C., Lago-Ballesteros, J. and Rey, E. (2011) 'Differences in performance indicators between winning and losing teams in the UEFA Champions League', *Journal of Human Kinetics*, 27(1). doi: 10.2478/v10078-011-0011-3.

Lago, C. (2009) 'The influence of match location, quality of opposition, and match status on possession strategies in professional association football.', *Journal of sports sciences*, 27(13), pp. 1463–1469. doi: 10.1080/02640410903131681.

Lago, C. and Martín, R. (2007) 'Determinants of possession of the ball in soccer', *Journal of Sports Sciences*. Routledge , 25(9), pp. 969–974. doi: 10.1080/02640410600944626.

Lahrmann, G. *et al.* (2011) 'Business Intelligence Maturity: Development and Evaluation of a Theoretical Model', in *2011 44th Hawaii International Conference on System Sciences*. IEEE, pp. 1–10. doi: 10.1109/HICSS.2011.90.

Laird, P. and Waters, L. (2008) 'Eyewitness Recollection of Sport Coaches', *International Journal of Performance Analysis in Sport*. Routledge, 8(1), pp. 76–84. doi: 10.1080/24748668.2008.11868424.

Lara, J. A. *et al.* (2014) 'A system for knowledge discovery in e-learning environments within the European Higher Education Area – Application to student data from Open University of Madrid, UDIMA', *Computers & Education*. Pergamon, 72, pp. 23–36. doi: 10.1016/J.COMPEDU.2013.10.009.

Larose, D. T. and Larose, C. D. (2014) Discovering Knowledge in Data: An Introduction to Data Mining, Wiley Series on Methods and Application in Data Mining. doi: 10.1017/CBO9781107415324.004.

Laudon, K. C. and Laudon, J. P. (2017) *Management information systems : managing the digital firm*. 15th edn. Pearson Education Ltd.
Laudon, K. C. and Laudon, J. P. (Jane P. (2006) *Management information systems : managing the digital firm*. Pearson/Prentice Hall. Available at: https://books.google.co.uk/books/about/Management_Information_Systems.html?id=m 84eAQAAIAAJ (Accessed: 25 July 2018).

Lechner, C. and Gudmundsson, S. V. (2012) 'Superior value creation in sports teams: Resources and managerial experience', *Management (France)*, 15(3), pp. 283–312. doi: 10.3917/mana.153.0284.

Lemire, J. (2018) *Catapult Sports Broadens Strategy, 'Isn't A Wearable Company' Anymore, www.sporttechie.com.* Available at: https://www.sporttechie.com/catapultsports-broadens-strategy-isnt-wearable-company-anymore/ (Accessed: 15 July 2018).

Leser, R., Baca, A. and Ogris, G. (2011) 'Local positioning systems in (game) sports', *Sensors*, 11(10), pp. 9778–9797. doi: 10.3390/s111009778.

Li, R. T. *et al.* (2016) 'Wearable Performance Devices in Sports Medicine', *Sports Health*, 8(1), pp. 74–78. doi: 10.1177/1941738115616917.

Lincoln, Y. S. and Guba, E. G. (1985) *Naturalistic inquiry*. Sage Publications. Available at:

https://books.google.co.uk/books?id=2oA9aWlNeooC&dq=naturalistic+inquiry&lr=&s ource=gbs_navlinks_s (Accessed: 24 August 2018).

Lincoln, Y. S., Lynham, S. A. and Guba, E. G. (2011) 'Pragmatic controversies, contradictions, and emerging confluences, revisited', in *The SAGE Handbook of Qualitative Research*.

Liu, H. *et al.* (2013) 'Inter-operator reliability of live football match statistics from OPTA Sportsdata', *International Journal of Performance Analysis in Sport*. Routledge, 13(3), pp. 803–821. doi: 10.1080/24748668.2013.11868690.

Liu, H. *et al.* (2015) 'Performance profiles of football teams in the UEFA Champions League considering situational efficiency', *International Journal of Performance Analysis in Sport*, 15(APRIL), pp. 371–390.

Liu, H., Gómez, M.-A., *et al.* (2016) 'Technical performance and match-to-match variation in elite football teams.', *Journal of sports sciences*. Routledge, 34(6), pp. 509–518. doi: 10.1080/02640414.2015.1117121.

Liu, H., Gómez, M. A., *et al.* (2016) 'Technical performance and match-to-match variation in elite football teams', *Journal of Sports Sciences*. Routledge, 34(6), pp. 509–518. doi: 10.1080/02640414.2015.1117121.

Liu, J. *et al.* (2013) 'Tracking Sports Players with Context-Conditioned Motion Models', in *Computer Vision and Pattern Recognition (CVPR), 2013 IEEE Conference on*, pp. 1830–1837. doi: 10.1109/CVPR.2013.239.

Lorains, M., Ball, K. and MacMahon, C. (2013) 'Performance analysis for decision making in team sports', *International Journal of Performance Analysis in Sport*. Routledge, 13(1), pp. 110–119. doi: 10.1080/24748668.2013.11868635.

Lyle, J. (2003) 'Stimulated recall: a report on its use in naturalistic research', *British Educational Research Journal*. Wiley-Blackwell, 29(6), pp. 861–878. doi: 10.1080/0141192032000137349.

Lyons, M., Al-Nakeeb, Y. and Nevill, A. (2006) 'Performance of Soccer Passing Skills Under Moderate and High-Intensity Localized Muscle Fatigue', *The Journal of Strength and Conditioning Research*, 20(1), p. 197. doi: 10.1519/R-17114.1.

Machado, J. C., Barreira, D. and Garganta, J. (2014) 'The influence of match status on attacking patterns of play in elite soccer teams', *Revista Brasileira de Cineantropometria e Desempenho Humano*, 16(5), pp. 545–554. doi: 10.5007/1980-0037.2014v16n5p545.

Mackenzie, R. and Cushion, C. (2013) 'Performance analysis in football: A critical review and implications for future research', *Journal of sports sciences*, 31:6, pp. 639–676.

Macutkiewicz, D. and Sunderland, C. (2011) 'The use of GPS to evaluate activity profiles of elite women hockey players during match-play', *Journal of Sports Sciences*, 29(9), pp. 967–973. doi: 10.1080/02640414.2011.570774.

Magee, C. (2015) Investors Swing For The Fences With \$1B In Sports Tech Deals /TechCrunch,techcrunch.com.Availableat:

https://techcrunch.com/2015/04/24/investors-swing-for-the-fences-with-1b-in-sports-tech-deals/?guccounter=1 (Accessed: 15 July 2018).

Mahony, L. E., Wheeler, K. W. and Lyons, K. (2012) 'Analysis of Factors Determining Invasion into Attacking Areas and the Creation of Goal-Scoring Opportunities in the Asian Cup Football Competition', *Asian Journal of Exercise & Sports Science*, 9(1), pp. 53–66. Available at: http://search.ebscohost.com/login.aspx?direct=true&db=s3h&AN=78130038&site=eho st-live.

Maier, A., Moultrie, J. and Clarkson, P. (2009) 'Developing maturity grids for assessing organisational capabilities: practitioner guidance', in *4th International Conference on Management Consulting, Academy of Management (MCD'09)*. Available at: http://publications.eng.cam.ac.uk/324215/ (Accessed: 15 July 2018).

Maksai, A., Wang, X. and Fua, P. (2015) 'What Players do with the Ball: A Physically Constrained Interaction Modeling'. Available at: http://arxiv.org/abs/1511.06181 (Accessed: 29 July 2018).

Manco, G. *et al.* (2016) 'Rialto: A Knowledge Discovery suite for data analysis', *Expert Systems with Applications*. Elsevier Ltd, 59, pp. 145–164. doi: 10.1016/j.eswa.2016.04.022.

Mann, C. and Maurer, F. (2005) 'A case study on the impact of scrum on overtime and customer satisfaction', in *Proceedings - AGILE Confernce 2005*. IEEE Comput. Soc, pp. 70–79. doi: 10.1109/ADC.2005.1.

Mara, J. *et al.* (2017) 'The Accuracy and Reliability of a New Optical Player Tracking System for Measuring Displacement of Soccer Players', *International Journal of Computer Science in Sport*, 16(3), pp. 175–184. doi: 10.1515/ijcss-2017-0013.

March, S. T. and Smith, G. F. (1995) 'Design and natural science research on information technology', *Decision Support Systems*, 15(4), pp. 251–266. doi: 10.1016/0167-9236(94)00041-2.

Maren, A. *et al.* (2005) 'Knowledge discovery system'. Available at: https://patents.google.com/patent/US20050278362A1/en (Accessed: 1 August 2018).

Marin-Perianu, R. *et al.* (2013) 'A performance analysis of a wireless body-area network monitoring system for professional cycling', *Personal and Ubiquitous Computing*, 17(1), pp. 197–209. doi: 10.1007/s00779-011-0486-x.

Mariscal, G., Marbán, Ó. and Fernández, C. (2010) 'A survey of data mining and knowledge discovery process models and methodologies', *The Knowledge Engineering Review*, 25(02), pp. 137–166. doi: 10.1017/S0269888910000032.

Martin, D. *et al.* (2017) 'The use, integration and value of performance analysis to GAA coaches', *Journal of Human Sport and Exercise*, 12(Proc2). doi: 10.14198/jhse.2017.12.Proc2.02.

Martin, D. *et al.* (2018) 'The use, integration and perceived value of performance analysis to professional and amateur Irish coaches', *International Journal of Sports Science & Coaching*, 0(0), p. 174795411775380. doi: 10.1177/1747954117753806.

Martindale, R. J. J. *et al.* (2010) 'Development of the Talent Development Environment Questionnaire for Sport', *Journal of Sports Sciences*, 28(11), pp. 1209–1221. doi: 10.1080/02640414.2010.495993.

Maslow, A. H. (1970) *Motivation and personality*. 2nd ed. Harper &Row,. Available at: https://vufind.lboro.ac.uk/Record/52042 (Accessed: 13 August 2018).

Matharu, G. S. *et al.* (2015) 'Empirical Study of Agile Software Development Methodologies', *ACM SIGSOFT Software Engineering Notes*. ACM, 40(1), pp. 1–6. doi: 10.1145/2693208.2693233.

McCormack, K. *et al.* (2009) 'A global investigation of key turning points in business process maturity', *Business Process Management Journal*, 15(5), pp. 792–815. doi: 10.1108/14637150910987946.

McFarlan, F. W., McKenney, J. L. and Pyburn, P. (1983) 'The information archipelago - plotting a course.', *Harvard Business Review*, 61(1), pp. 145–156. Available at: http://www.ncbi.nlm.nih.gov/pubmed/10299000 (Accessed: 28 September 2018).

McGarry, T., O'Donoghue, P., & Sampaio, J. (2013) *Routledge handbook of sports performance analysis*, *Routledge*. Routledge. Available at: https://books.google.co.uk/books?hl=en&lr=&id=fbNvlG9ww1EC&oi=fnd&pg=PA176 &dq=Wiltshire,+2013+coach&ots=af0YttZdm5&sig=aUs0ZZndk8B8z_RJYWvhHC0y 0hY&redir_esc=y#v=onepage&q=Wiltshire%2C 2013 coach&f=false (Accessed: 13 August 2018).

McHugh, O., Conboy, K. and Lang, M. (2012) 'Agile Practices: The Impact on Trust in Software Project Teams', *IEEE Software*, 29(3), pp. 71–76. doi: 10.1109/MS.2011.118. McHugh, O., Conboy, K. and Lang, M. (2014) 'Agile Practices: The Impact on Trust in Software Project Teams', *IEEE Software*. doi: 10.1109/MS.2011.118.

McKenna, M. *et al.* (2018a) 'Neophyte experiences of football (soccer) match analysis: a multiple case study approach', *Research in Sports Medicine*. Routledge, 26(3), pp. 306–322. doi: 10.1080/15438627.2018.1447473.

McKenna, M. *et al.* (2018b) 'Neophyte experiences of football match analysis: a multiple case study approach', *Research in Sports Medicine*. Routledge, 00(00), pp. 1–17. doi: 10.1080/15438627.2018.1447473.

McLean, S. *et al.* (2017) 'What's in a game? A systems approach to enhancing performance analysis in football', *Plos One*, 12(2), p. e0172565. doi: 10.1371/journal.pone.0172565.

Medeiros, J. (2017) 'How analytics killed the Premier League's long ball game and inspired gegenpressing and tiki taka', *WIRED UK*, August. Available at: https://www.wired.co.uk/article/premier-league-stats-football-analytics-prozone-

gegenpressing-tiki-taka (Accessed: 11 August 2018).

Mehravari, N. (2014) *Everything You Always Wanted to Know About Maturity Models*, *Software Engineering Institute*. Available at: https://resources.sei.cmu.edu/library/assetview.cfm?assetid=293858 (Accessed: 14 July 2018).

Melville, B. N. and Kraemer, K. (2004) 'Review: Technology Information An Performance: Organizational Integrative Model of IT Business', *MIS Quarterly*, 28(2), pp. 283–322.

Mertens, D. M. (2007) 'Transformative Paradigm', *Journal of Mixed Methods Research*. Sage PublicationsSage CA: Los Angeles, CA, 1(3), pp. 212–225. doi: 10.1177/1558689807302811.

Mettler, T. and Rohner, P. (2009) 'Situational maturity models as instrumental artifacts for organizational design', in *Proceedings of the 4th International Conference on Design Science Research in Information Systems and Technology - DESRIST '09.* New York, New York, USA: ACM Press, p. 1. doi: 10.1145/1555619.1555649.

Mettler, T., Rohner, P. and Winter, R. (2010) 'Towards a Classification of Maturity Models in Information Systems', in *Management of the Interconnected World*. Heidelberg: Physica-Verlag HD, pp. 333–340. doi: 10.1007/978-3-7908-2404-9_39.

Min, B. *et al.* (2008) 'A compound framework for sports results prediction: A football case study', *Knowledge-Based Systems*, 21(7), pp. 551–562. doi: 10.1016/j.knosys.2008.03.016.

Mishra, D., Akman, I. and Mishra, A. (2014) 'Theory of Reasoned Action application for Green Information Technology acceptance', *Computers in Human Behavior*. doi: 10.1016/j.chb.2014.03.030.

Mohr, M., KRUSTRUP, P. and BANGSBO, J. (2003) 'Match performance of highstandard soccer players with special reference to development of fatigue.', *Journal of sports sciences*, 21(7), pp. 519–528. doi: 10.1080/0264041031000071182.

Moll, T., Jordet, G. and Pepping, G.-J. (2010) 'Emotional contagion in soccer penalty shootouts: Celebration of individual success is associated with ultimate team success', *Journal of Sports Sciences*, 28(9), pp. 983–992. doi: 10.1080/02640414.2010.484068.

Mooney, M. *et al.* (2011) 'The relationship between physical capacity and match performance in elite Australian football: A mediation approach', *Journal of Science and Medicine in Sport.* doi: 10.1016/j.jsams.2011.03.010.

Nambisan, S. and Baron, R. A. (2009) 'Virtual Customer Environments: Testing a Model of Voluntary Participation in Value Co-creation Activities', *Journal of Product Innovation Management*. Wiley/Blackwell (10.1111), 26(4), pp. 388–406. doi: 10.1111/j.1540-5885.2009.00667.x.

Nash, C. and Collins, D. (2006) 'Tacit Knowledge in Expert Coaching: Science or Art?', *Quest*, 58(4), pp. 465–477. doi: 10.1080/00336297.2006.10491894.

Ndlec, M. *et al.* (2012) 'Recovery in Soccer: Part I-post-match fatigue and time course of recovery', *Sports Medicine*, 42(12), pp. 997–1015. doi: 10.2165/11635270-00000000-00000.

Nelson, L., Cushion, C. and Potrac, P. (2013) 'Enhancing the provision of coach education: The recommendations of UK coaching practitioners', *Physical Education and Sport Pedagogy*, 18(2), pp. 204–218. doi: 10.1080/17408989.2011.649725.

Nelson, L. J., Potrac, P. and Groom, R. (2014) 'Receiving video-based feedback in elite ice-hockey: a player's perspective', *Sport, Education and Society*, 19(1), pp. 19–40. doi: 10.1080/13573322.2011.613925.

Nevo, S. and Wade, M. (2011) 'Firm-level benefits of IT-enabled resources: A conceptual extension and an empirical assessment', *Journal of Strategic Information Systems*. doi: 10.1016/j.jsis.2011.08.001.

Nevo and Wade (2010) 'The Formation and Value of IT-Enabled Resources: Antecedents and Consequences of Synergistic Relationships', *MIS Quarterly*. doi: 10.2307/20721419. Nicholls, S. B. and Worsfold, P. R. (2016) 'The observational analysis of elite coaches within youth soccer: The importance of performance analysis', *International Journal of*

Sports Science & Coaching. SAGE PublicationsSage UK: London, England, 11(6), pp. 825–831. doi: 10.1177/1747954116676109.

Nolan, R. L. (1979) 'Managing Crises of Data Processing', *Harvard business review*, 3(4).

Nolan, R. L. and L., R. (1973) 'Managing the computer resource: a stage hypothesis', *Communications of the ACM*, 1 July, pp. 399–405. doi: 10.1145/362280.362284.

Nolan, R. L. and L., R. (1975) 'Thoughts about the fifth stage', *ACM SIGMIS Database*. ACM, 7(2), pp. 4–10. doi: 10.1145/1017570.1017571.

O'Donoghue, P. (2006) 'The use of feedback videos in sport.', *International Journal of Performance Analysis in Sport*. Routledge, 6(2), pp. 1–14. doi: 10.1080/24748668.2006.11868368.

O'Donoghue, P. (2008) 'Principal Components Analysis in the selection of Key Performance Indicators in Sport', *International Journal of Performance Analysis in Sport*. Routledge, 8(3), pp. 145–155. doi: 10.1080/24748668.2008.11868456.

O'Donoghue, P. and Cullinane, A. (2011) 'A regression-based approach to interpreting sports performance', *International Journal of Performance Analysis in Sport*. Routledge, 11(2), pp. 295–307. doi: 10.1080/24748668.2011.11868549.

O'Donoghue, P. and Holmes, L. (2014) *Data Analysis in Sport*. doi: 10.1017/CBO9781107415324.004.

O'Donoghue, P. and Robinson, G. (2016) 'The effect of dismissals on work-rate in English FA Premier League soccer', *International Journal of Performance Analysis in Sport*. Routledge, 16(3), pp. 898–909. doi: 10.1080/24748668.2016.11868937.

Ofoghi, B. *et al.* (2013) 'Data Mining in Elite Sports: A Review and a Framework', *Measurement in Physical Education and Exercise Science*, 17(3), pp. 171–186. doi: 10.1080/1091367X.2013.805137.

De Oliveira Bueno, M. J. *et al.* (2014) 'Analysis of the distance covered by Brazilian professional futsal players during official matches', *Sports Biomechanics*, 13(3), pp. 230–240. doi: 10.1080/14763141.2014.958872.

Olsen, E. and Larsen, O. (1997) 'Use of Match Analysis by Coache', in *The Third World Congress of Science and Football III*, pp. 200–220. Available at: https://books.google.co.uk/books?hl=en&lr=&id=3TnKAgAAQBAJ&oi=fnd&pg=PA2 09&dq=Olsen,+E.,+and+Larsen,+O.+(1997).+Use+of+match+analysis+by+coaches.+In +T.+Reilly,+J.+Bangsbo,+and+M.+Hughes+(Eds.),+Science+and+football+III+(pp.+20 9-220).+London:+E.+and+F.+Spo (Accessed: 31 July 2018).

Olson, D. L. (2018) Data Mining Models, Second Edition. Business Expert Press. Available at:

https://books.google.co.uk/books?id=T99XDwAAQBAJ&dq=discovering+patterns,+gr ouping,+correlating+olson+2018+%22data+mining%22&source=gbs_navlinks_s (Accessed: 11 August 2018).

Olszak, C. M. (2016) 'Toward Better Understanding and Use of Business Intelligence in Organizations', *Information Systems Management*. Taylor & Francis, 33(2), pp. 105–123. doi: 10.1080/10580530.2016.1155946.

OPTA (2018) *OPTA*, *Opta*. Available at: https://www.optasports.com/services/data-feeds/ (Accessed: 2 August 2018).

Ordanini, A. and Pasini, P. (2008) 'Service co-production and value co-creation: The case for a service-oriented architecture (SOA)', *European Management Journal*. Pergamon, 26(5), pp. 289–297. doi: 10.1016/J.EMJ.2008.04.005.

Orłowski, C., Ziółkowski, A. and Paciorkiewicz, G. (2017) 'Quantitative Assessment of the IT Agile Transformation', *Procedia Engineering*. The Author(s), 182, pp. 524–531. doi: 10.1016/j.proeng.2017.03.147.

Orth, D. *et al.* (2014) 'Effects of a defender on run-up velocity and ball speed when crossing a football', *European Journal of Sport Science*, 14(sup1), pp. S316–S323. doi: 10.1080/17461391.2012.696712.

Parmar, N. *et al.* (2017) 'Team performance indicators that predict match outcome and points difference in professional rugby league', *International Journal of Performance Analysis in Sport*, 17(6), pp. 1044–1056. doi: 10.1080/24748668.2017.1419409.

Patton, J. (2014) User story mapping.

Paulk, M. C. *et al.* (1993) 'Capability maturity model, version 1.1', *IEEE Software*, 10(4), pp. 18–27. doi: 10.1109/52.219617.

Paulk, M. C. *et al.* (1993) *Key Practices of the Capability Maturity Model Version 1.1*.
Available at: https://resources.sei.cmu.edu/library/asset-view.cfm?assetid=11965
(Accessed: 12 July 2018).

Peffers, K. *et al.* (2007) 'A Design Science Research Methodology for Information Systems Research', *Journal of Management Information Systems*. Routledge, 24(3), pp. 45–77. doi: 10.2753/MIS0742-1222240302.

Peppard, J., Ward, J. and Daniel, E. (2007) 'Managing the realization of business benefits from IT investments', *MIS Quarterly Executive*, 6(1), pp. 1–11. doi: 0657.

Perin, C., Vuillemot, R. and Fekete, J. D. (2013) 'SoccerStories: A kick-off for visual soccer analysis', *IEEE Transactions on Visualization and Computer Graphics*, 19(12), pp. 2506–2515. doi: 10.1109/TVCG.2013.192.

Peteraf, M. A. and Barney, J. B. (2003) 'Unraveling the resource-based tangle', *Managerial and Decision Economics*. Wiley-Blackwell, 24(4), pp. 309–323. doi: 10.1002/mde.1126.

Petter, S., DeLone, W. and McLean, E. (2008) 'Measuring information systems success: models, dimensions, measures, and interrelationships', *European Journal of Information Systems*. Palgrave Macmillan UK, 17(3), pp. 236–263. doi: 10.1057/ejis.2008.15.

PhridviRaj, M. S. B. and GuruRao, C. V. (2014) 'Data Mining – Past, Present and Future – A Typical Survey on Data Streams', *Procedia Technology*. Elsevier B.V., 12, pp. 255–263. doi: 10.1016/j.protcy.2013.12.483.

Piccoli and Ives (2005) 'Review: IT-Dependent Strategic Initiatives and Sustained Competitive Advantage: A Review and Synthesis of the Literature', *MIS Quarterly*. doi: 10.2307/25148708.

Plumley, D., Wilson, R. and Ramchandani, G. (2017) 'Towards a model for measuring holistic performance of professional Football clubs', *Soccer & Society*. Routledge, 18(1), pp. 16–29. doi: 10.1080/14660970.2014.980737.

Poli, R. and Obrst, L. (2010) 'The Interplay Between Ontology as Categorial Analysis and Ontology as Technology', in *Theory and Applications of Ontology: Computer Applications*. Dordrecht: Springer Netherlands, pp. 1–26. doi: 10.1007/978-90-481-8847-5_1.

Pollard, R. (2002) 'Charles Reep (1904-2002): pioneer of notational and performance analysis in football', *Journal of Sports Sciences*. Taylor & Francis , 20(10), pp. 853–855. doi: 10.1080/026404102320675684.

Pollard, R. and Reep, C. (1997) 'Measuring the effectiveness of playing strategies at soccer', *Journal of the Royal Statistical Society Series D: The Statistician*. WileyRoyal Statistical Society, 46(4), pp. 541–550. doi: 10.1111/1467-9884.00108.

Popovič, A. *et al.* (2012) 'Towards business intelligence systems success: Effects of maturity and culture on analytical decision making', *Decision Support Systems*. North-Holland, 54(1), pp. 729–739. doi: 10.1016/J.DSS.2012.08.017.

Popovič, A., Coelho, P. S. and Jaklič, J. (2009) 'The Impact of Business Intelligence System Maturity on Information Quality', *Information Research*, 14(4). Available at: https://papers.ssrn.com/sol3/papers.cfm?abstract_id=1625573 (Accessed: 27 September 2018).

Pöppelbuß, J. and Röglinger, M. (2011) 'WHAT MAKES A USEFUL MATURITY MODEL?A FRAMEWORK OF GENERAL DESIGN PRINCIPLES FORMATURITY MODELS AND ITS DEMONSTRATION INBUSINESS PROCESS MANAGEMENT', in *ECIS 2011 Proceedings*. AIS Electronic Library (AISeL). Available at: http://aisel.aisnet.org/ecis2011/28 (Accessed: 15 July 2018).

Potrac, P. and Jones, R. (2009) 'Power, Conflict, and Cooperation: Toward a Micropolitics of Coaching', *Quest*. Taylor & Francis Group, 61(2), pp. 223–236. doi: 10.1080/00336297.2009.10483612.

Potrac, P., Jones, R. and Armour, K. (2002) "'It's All About Getting Respect": The Coaching Behaviors of an Expert English Soccer Coach', *Sport, Education and Society*, 7(2), pp. 183–202. doi: 10.1080/1357332022000018869.

Poulter, D. R. (2009) 'Home advantage and player nationality in international club football', *Journal of Sports Sciences*. Routledge , 27(8), pp. 797–805. doi: 10.1080/02640410902893364.

Prahalad, C. K. and Ramaswamy, V. (2004) 'Co-creation experiences: The next practice in value creation', *Journal of Interactive Marketing*, 18(3), pp. 5–14. doi: 10.1002/dir.20015.

Prananto, A., McKay, J. and Marshall, P. (2003) 'A Study of the Progression of E-Business Maturity in Australian SMEs: Some Evidence of the Applicability of the Stages of Growth for E-Business Model', *undefined*. Available at: https://www.semanticscholar.org/paper/A-Study-of-the-Progression-of-E-BusinessMaturity-Prananto-McKay/364000ba81d2c3a38ebfbe5e5d9ba5f819b12dcd (Accessed: 26 September 2018).

Pranicevic, D. G., Alfirevic, N. and Stemberger, M. I. (2011) 'Information System Maturity and the Hospitality Enterprise Performance', *Economic and Business Review for Central and South - Eastern Europe*.

Proença, D. and Borbinha, J. (2016) 'Maturity Models for Information Systems - A State of the Art', *Procedia Computer Science*. Elsevier, 100, pp. 1042–1049. doi: 10.1016/J.PROCS.2016.09.279.

Rajterič, I. H. (2010) 'Overview of Business Intelligence Maturity Models', *Management : journal of contemporary management issues*. Faculty of Economics, 15(1), pp. 47–67. Available at: https://hrcak.srce.hr/53606?lang=en (Accessed: 27 September 2018).

Rampinini, E. *et al.* (2007) 'Variation in top level soccer match performance', *International Journal of Sports Medicine*, 28(12), pp. 1018–1024. doi: 10.1055/s-2007-965158.

Rampinini, E. *et al.* (2009) 'Technical performance during soccer matches of the Italian Serie A league: Effect of fatigue and competitive level', *Journal of Science and Medicine in Sport*, 12(1), pp. 227–233. doi: 10.1016/j.jsams.2007.10.002.

Randers, M. B. *et al.* (2010) 'Application of four different football match analysis systems: A comparative study', *Journal of sports sciences*, 28(2), pp. 171–182.

Rangsee, P., Suebsombat, P. and Boonyanant, P. (2013) 'Simplified low-cost GPS-based tracking system for soccer practice', 13th International Symposium on Communications and Information Technologies: Communication and Information Technology for New Life Style Beyond the Cloud, ISCIT 2013, pp. 724–728. doi: 10.1109/ISCIT.2013.6645948.

Rayner, N. and Schlegel, K. (2008) 'Maturity model overview for business intelligence and performance management', *Gartner, Stamford*.

Redwood-Brown, A. *et al.* (2012) 'The effect of score-line on work-rate in English FA Premier League soccer', *International Journal of Performance Analysis in Sport*. Routledge, 12(2), pp. 258–271. doi: 10.1080/24748668.2012.11868598.

Reep, C. and Benjamin, B. (1968) 'Skill and Chance in Association Football', *Journal of the Royal Statistical Society. Series A (General)*. WileyRoyal Statistical Society, 131(4), p. 581. doi: 10.2307/2343726.

Rein, R. and Memmert, D. (2016) 'Big data and tactical analysis in elite soccer: future challenges and opportunities for sports science', *SpringerPlus*. Springer International Publishing, 5(1), p. 1410. doi: 10.1186/s40064-016-3108-2.

Rein, R., Raabe, D. and Memmert, D. (2017) "Which pass is better?" Novel approaches to assess passing effectiveness in elite soccer', *Human Movement Science*. Elsevier, 55(July), pp. 172–181. doi: 10.1016/j.humov.2017.07.010.

Rohde, M. and Breuer, C. (2017) 'The market for football club investors: a review of theory and empirical evidence from professional European football', *European Sport Management Quarterly*, 17(3), pp. 265–289. doi: 10.1080/16184742.2017.1279203.

Ronkainen, J. and Harland, A. (2010) 'Laser tracking system for sports ball trajectory measurement Laser tracking system for sports ball trajectory measurementg', *Journal of Sports Engineering and Technology*, 224(3), pp. 219–228. doi: 10.1243/17543371JSET67.

Rosemann, M. and Bruin, T. De (2005) 'Towards a Business Process Mangement Maturity Model', *ECIS 2005 Proceedings of the Thirteenth European Conference on Information Systems*. doi: 10.1109/EUROMICRO.2007.35.

Rowley, J. (2007) 'The wisdom hierarchy: representations of the DIKW hierarchy', *Journal of Information Science*. Sage PublicationsSage CA: Thousand Oaks, CA, 33(2), pp. 163–180. doi: 10.1177/0165551506070706.

Rubin, K. S. (2012) 'Essential Scrum : a practical guide to the most popular agile process', *IT Professional*. doi: 10.1007/978-3-658-01284-7.

Ruiz-Ruiz, C. *et al.* (2013) 'Analysis of entries into the penalty area as a performance indicator in soccer', *European Journal of Sport Science*, 13(3), pp. 241–248. doi: 10.1080/17461391.2011.606834.

Ryoo, M., Kim, N. and Park, K. (2018) 'Visual analysis of soccer players and a team', *Multimedia Tools and Applications*. Springer US, 77(12), pp. 15603–15623. doi: 10.1007/s11042-017-5137-4.

SAFF (2018) *Saudi Professional League*, *SAFF*. Available at: http://www.saff.com.sa/league/saudi-professional-league (Accessed: 14 August 2018).

Sagiroglu, S. and Sinanc, D. (2013) 'Big data: A review', in 2013 International Conference on Collaboration Technologies and Systems (CTS), pp. 42–47. doi: 10.1109/CTS.2013.6567202.

Saltz, J. S., Shamshurin, I. and Crowston, K. (2017) 'Comparing Data Science Project Management Methodologies via a Controlled Experiment', pp. 1013–1022. Available at: http://hdl.handle.net/10125/41273.

Di Salvo, V. *et al.* (2007) 'Performance characteristics according to playing position in elite soccer', *International Journal of Sports Medicine*, 28(3), pp. 222–227. doi: 10.1055/s-2006-924294.

Di Salvo, V. *et al.* (2009) 'Analysis of High Intensity Activity in Premier League Soccer', *International Journal of Sports Medicine*, 30(03), pp. 205–212. doi: 10.1055/s-0028-1105950.

Di Salvo, V. *et al.* (2010) 'Sprinting analysis of elite soccer players during European Champions League and UEFA Cup matches', *Journal of Sports Sciences*, 28(14), pp. 1489–1494. doi: 10.1080/02640414.2010.521166.

Sampaio, J. *et al.* (2015) 'Exploring Game Performance in the National Basketball Association Using Player Tracking Data.', *PloS one*. Public Library of Science, 10(7), p. e0132894. doi: 10.1371/journal.pone.0132894.

Sampaio, M. A. G. M. G.-L. C. L. & J. *et al.* (2012) 'Effects of game location and final outcome on game-related statistics in each zone of the pitch in professional football', *European Journal of Sport Science*, 12(5), pp. 393–398. doi: 10.1080/17461391.2011.566373.

Santos, S. *et al.* (2010) 'Coaches' perceptions of competence and acknowledgement of training needs related to professional competences', *Journal of Sports Science and Medicine*, 9(1), pp. 62–70.

Sarmento, H., Marcelino, R., Anguera, M. T., *et al.* (2014) 'Match analysis in football: a systematic review.', *Journal of sports sciences*, 32(20), pp. 1831–1843. doi: 10.1080/02640414.2014.898852.

Sarmento, H., Marcelino, R., Anguera, M. T., *et al.* (2014) 'Match analysis in football: a systematic review', *Journal of Sports Sciences*, 32(20), pp. 1831–1843. doi: 10.1080/02640414.2014.898852.

Sarmento, H., Pereira, A., *et al.* (2014) 'The Coaching Process in Football – A qualitative perspective', *J. Sports Sci. Med*, 3(1), pp. 9–16.

Sarmento, H. *et al.* (2017) 'What Performance Analysts Need to Know About Research Trends in Association Football (2012–2016): A Systematic Review', *Sports Medicine*, pp. 1–38. doi: 10.1007/s40279-017-0836-6.

Scheider, S., Ostermann, F. O. and Adams, B. (2017) 'Why good data analysts need to be critical synthesists. Determining the role of semantics in data analysis', *Future Generation Computer Systems*. Elsevier B.V., 72, pp. 11–22. doi: 10.1016/j.future.2017.02.046.

Schlipsing, M. *et al.* (2017) 'Adaptive pattern recognition in real-time video-based soccer analysis', *Journal of Real-Time Image Processing*, 13(2), pp. 345–361. doi: 10.1007/s11554-014-0406-1.

Schryen, G. (2013) 'Revisiting IS business value research: What we already know, what we still need to know, and how we can get there', *European Journal of Information Systems*, 22(2), pp. 139–169. doi: 10.1057/ejis.2012.45.

Scott, W. R. (2010) 'Reflections: The Past and Future of Research on Institutions and Institutional Change', *Journal of Change Management*. Routledge , 10(1), pp. 5–21. doi: 10.1080/14697010903549408.

Serra, C. E. M. and Kunc, M. (2015) 'Benefits Realisation Management and its influence on project success and on the execution of business strategies', *International Journal of Project Management*. The Authors, 33(1), pp. 53–66. doi: 10.1016/j.ijproman.2014.03.011.

Serrador, P. and Pinto, J. K. (2015) 'Does Agile work? — A quantitative analysis of agile project success', *International Journal of Project Management*. Pergamon, 33(5), pp. 1040–1051. doi: 10.1016/J.IJPROMAN.2015.01.006.

Serrador, P. and Rodney Turner, J. (2014) 'The Relationship between Project Success and Project Efficiency', *Procedia - Social and Behavioral Sciences*. Elsevier, 119, pp. 75–84. doi: 10.1016/J.SBSPRO.2014.03.011.

Seshadri, D. R. *et al.* (2017) 'Wearable Devices for Sports: New Integrated Technologies Allow Coaches, Physicians, and Trainers to Better Understand the Physical Demands of Athletes in Real time', *IEEE Pulse*. doi: 10.1109/MPUL.2016.2627240.

Shannon, C. E. (1948) 'A Mathematical Theory of Communication', *Bell System Technical Journal*, 27(3), pp. 379–423. doi: 10.1002/j.1538-7305.1948.tb01338.x.

Simonsson, M., Johnson, P. and Wijkström, H. (2007) 'Model-Based IT Governance Maturity Assessments with Cobit', *undefined*. Available at: https://www.semanticscholar.org/paper/Model-Based-IT-Governance-Maturity-

Assessments-with-Simonsson-Johnson/53801950a4dbb9c7422fcdc93fda7b6b950993c1 (Accessed: 27 September 2018).

Škegro, D., Milanović, D. and Sporiš, G. (2012) 'Performance Analysis in sport', *4th International Scientific Conference "Contemporary Kinesiology"*, pp. 1–8. Available at: http://bib.irb.hr/prikazi-rad?rad=591816.

Slater, S. *et al.* (2017) 'Tools for Educational Data Mining', *Journal of Educational and Behavioral Statistics*. SAGE PublicationsSage CA: Los Angeles, CA, 42(1), pp. 85–106. doi: 10.3102/1076998616666808.

Sloane, P. J. (1971) 'SCOTTISH JOURNAL OF POLITICAL ECONOMY:THE ECONOMICS OF PROFESSIONAL FOOTBALL: THE FOOTBALL CLUB AS A UTILITY MAXIMISER', *Scottish Journal of Political Economy*. Wiley/Blackwell (10.1111), 18(2), pp. 121–146. doi: 10.1111/j.1467-9485.1971.tb00979.x.

Smart, D. L. and Wolfe, R. A. (2003) 'The contribution of leadership and human resources to organizational success: An empirical assessment of performance in major league baseball', *European Sport Management Quarterly*. Taylor & Francis Group, 3(3), pp. 165–188. doi: 10.1080/16184740308721949.

Software Engineering Institute (2005) 'Capability Maturity Model ® Integration (CMMI ®) overview', *Carniege Mellon University*, pp. 1–58.

Software Engineering Institute (2010) *CMMI for Development, Version 1.3, Carnegie Mellon University.* doi: CMU/SEI-2010-TR-033 ESC-TR-2010-033.

Solli-Sæther, H. and Gottschalk, P. (2010) 'The Modeling Process for Stage Models', *Journal of Organizational Computing and Electronic Commerce*. Taylor & Francis Group , 20(3), pp. 279–293. doi: 10.1080/10919392.2010.494535.

Souza, T. F. de and Gomes, C. F. S. (2015) 'Assessment of Maturity in Project Management: A Bibliometric Study of Main Models', *Procedia Computer Science*. Elsevier, 55, pp. 92–101. doi: 10.1016/J.PROCS.2015.07.012.

STATS (2018) *STATS*, *STATS*. Available at: https://www.stats.com/data-feeds/ (Accessed: 2 August 2018).

STATSports (2018) *STATSports*, *STATSports*. Available at: https://statsports.com/ (Accessed: 2 August 2018).

van Steenbergen, M. et al. (2010) 'The Design of Focus Area Maturity Models', in *Proceedings of the 5th international conference on Global Perspectives on Design Science Research*. Springer-Verlag, pp. 317–332. doi: 10.1007/978-3-642-13335-0_22.

Stein, M. *et al.* (2015) 'Visual Soccer Analytics: Understanding the Characteristics of Collective Team Movement Based on Feature-Driven Analysis and Abstraction', *ISPRS International Journal of Geo-Information*, 4(4), pp. 2159–2184. doi: 10.3390/ijgi4042159.

Stein, M. *et al.* (2016) 'Director's Cut: Analysis and Annotation of Soccer Matches', *IEEE Computer Graphics and Applications*, 36(5), pp. 50–60. doi: 10.1109/MCG.2016.102.

Stein, M. *et al.* (2017a) 'How to Make Sense of Team Sport Data: From Acquisition to Data Modeling and Research Aspects', *Data*, 2(1), p. 2. doi: 10.3390/data2010002.

Stein, M. *et al.* (2017b) 'How to Make Sense of Team Sport Data: From Acquisition to Data Modeling and Research Aspects', *Data*, 2(1), p. 2. doi: 10.3390/data2010002.

Stein, M. *et al.* (2017c) 'How to Make Sense of Team Sport Data: From Acquisition to Data Modeling and Research Aspects', *Data*, 2(1), p. 2. doi: 10.3390/data2010002.

Stellman, A. and Greene, J. (2014) *Learning Agile*. O'Reilly Media, Inc. Available at: https://books.google.co.uk/books?id=XLxUBQAAQBAJ&dq=%22Learning+Agile:+U nderstanding+Scrum,+XP,+Lean,+and+Kanban%22+pdf&lr=&source=gbs_navlinks_s (Accessed: 7 July 2018).

Stratman, J. K. and Roth, A. V. (2002) 'Enterprise Resource Planning (ERP) Competence Constructs: Two-Stage Multi-Item Scale Development and Validation', *Decision Sciences*. doi: 10.1111/j.1540-5915.2002.tb01658.x.

Stratton, G. et al. (2004) Youth soccer: From Science to Performance, Youth Soccer: From Science to Performance. Routledge. doi: 10.4324/9780203644133.

Sugimoto, M. *et al.* (2012) 'An accurate 3D localization technique using a single camera and ultrasound', 2012 International Conference on Indoor Positioning and Indoor Navigation (IPIN), 9725(November), pp. 1–8. doi: 10.1109/IPIN.2012.6418874.

Sun, Z., Sun, L. and Strang, K. (2018) 'Big Data Analytics Services for Enhancing Business Intelligence', *Journal of Computer Information Systems*. Taylor & Francis, 58(2), pp. 162–169. doi: 10.1080/08874417.2016.1220239.

Suri, H. (2011) 'Purposeful Sampling in Qualitative Research Synthesis', *Qualitative Research Journal*, 11(2), pp. 63–75. doi: 10.3316/QRJ1102063.

Sutherland, J. and Schwaber, K. (2017) *Scrum Guide*. Available at: http://www.scrumguides.org/scrum-guide.html#events-planning (Accessed: 25 June 2018).

Szymanski, S. and Kuypers, T. (2000) *Winners and losers*. Penguin. Available at: https://books.google.co.uk/books?id=T0DCAAAACAAJ&dq=Winners+and+Losers+-

+The+Business+Strategy+of+Football&hl=en&sa=X&ved=0ahUKEwiNnOiGrejcAhW KJMAKHSMbACYQ6AEIJzAA (Accessed: 12 August 2018).

Szymanski, S. and Smith, R. (1997) 'The English Football Industry: profit, performance and industrial structure', *International Review of Applied Economics*, 11(1), pp. 135–153. doi: 10.1080/02692179700000008.

Taibi, D. *et al.* (2017) 'Comparing Requirements Decomposition Within the Scrum, Scrum with Kanban, XP, and Banana Development Processes', in. Springer, Cham, pp. 68–83. doi: 10.1007/978-3-319-57633-6_5.

Tan, C.-S., Sim, Y.-W. and Yeoh, W. (2011) 'A maturity model of enterprise business intelligence', *Communications of the IBIMA*. International Business Information Management Association, 2011(417812), pp. 1–11. Available at: http://dro.deakin.edu.au/view/DU:30036783 (Accessed: 27 September 2018).

Tashakkori, A. and Creswell, J. W. (2008) 'Editorial: Mixed Methodology Across Disciplines', *Journal of Mixed Methods Research*, 2(1), pp. 3–6. doi: 10.1177/1558689807309913.

Taylor, J. B. et al. (2008a) 'The influence of match location, quality of opposition, andmatch status on technical performance in professional association football',http://dx.doi.org/10.1080/02640410701836887.Routledge10.1080/02640410701836887.

Taylor, J. B. *et al.* (2008b) 'The influence of match location, quality of opposition, and match status on technical performance in professional association football', *Journal of Sports Sciences*. Routledge , 26(9), pp. 885–895. doi: 10.1080/02640410701836887.

Teddlie, C. and Tashakkori, A. (2012) 'Common "Core" Characteristics of Mixed Methods Research', *American Behavioral Scientist*. SAGE PublicationsSage CA: Los Angeles, CA, 56(6), pp. 774–788. doi: 10.1177/0002764211433795.

Tenga, A. *et al.* (2010a) 'Effect of playing tactics on achieving score-box possessions in a random series of team possessions from Norwegian professional soccer matches', *Journal of Sports Sciences*, 28(3), pp. 245–255. doi: 10.1080/02640410903502766.

Tenga, A. *et al.* (2010b) 'Effect of playing tactics on goal scoring in Norwegian professional soccer', *Journal of Sports Sciences*. Routledge , 28(3), pp. 237–244. doi: 10.1080/02640410903502774.

Tenga, A. (2010) 'Reliability and validity of match performance analysis in soccer : a multidimensional qualitative evaluation of opponent interaction'. doi: 171287.

Tenga, A., Mortensholm, A. and O'Donoghue, P. (2017) 'Opposition interaction in creating penetration during match play in elite soccer: evidence from UEFA champions league matches', *International Journal of Performance Analysis in Sport*. Routledge, 17(5), pp. 802–812. doi: 10.1080/24748668.2017.1399326.

Thuraisingham, B. (2014) Data Mining. CRC Press. doi: 10.1201/b16553.

Tovinkere, V. and Qian, R. J. (2001) 'Detecting semantic events in soccer games: Towards a complete solution', in *Proceedings - IEEE International Conference on Multimedia and Expo*, pp. 833–836. doi: 10.1109/ICME.2001.1237851.

Trewin, J. *et al.* (2017) 'The influence of situational and environmental factors on matchrunning in soccer: a systematic review', *Science and Medicine in Football*. Routledge, 1(2), pp. 183–194. doi: 10.1080/24733938.2017.1329589.

Tucker, W. *et al.* (2005) 'Game Location Effects in Professional Soccer: A Case Study', *International Journal of Performance Analysis in Sport*, 5(2), pp. 23–35. doi: 10.1080/24748668.2005.11868325.

Tunaru, R. S. and Viney, H. P. (2010) 'Valuations of Soccer Players from Statistical Performance Data', *Journal of Quantitative Analysis in Sports*. De Gruyter, 6(2). doi: 10.2202/1559-0410.1238.

Turban, E., Rainer, R. K. and Potter, R. E. (2005) *Introduction to information technology*. John Wiley & Sons.

Urquhart, C., Lehmann, H. and Myers, M. D. (2010) 'Putting the "theory" back into grounded theory: Guidelines for grounded theory studies in information systems', *Information Systems Journal*. doi: 10.1111/j.1365-2575.2009.00328.x.

van de Ven, A. H. and Poole, M. S. (1995) 'Explaining Development and Change in Organizations', *The Academy of Management Review*, 20(3), p. 510. doi: 10.2307/258786.

Venkatesh, V. and Bala, H. (2008) 'Technology Acceptance Model 3 and a Research Agenda on Interventions', *Decision Sciences*. Wiley/Blackwell (10.1111), 39(2), pp. 273–315. doi: 10.1111/j.1540-5915.2008.00192.x.

Venkatesh, V., Thong, J. Y. L. and Xu, X. (2012) 'Consumer Acceptance and Use of Information Technology: Extending the Unified Theory of Acceptance and Use of Technology', *MIS Quarterly*. Management Information Systems Research Center, University of Minnesota, 36, pp. 157–178. doi: 10.2307/41410412.

Vesset, D. *et al.* (2013) 'IDC MaturityScape: Big Data and Analytics - A Guide to Unlocking Information Assets'. IDC Research. Available at: https://scholar.google.com/scholar?hl=en&as_sdt=0%2C5&q=IDC+maturityscape%3A +Big+data+and+analytics+-

+A+guide+to+unlocking+information+assets.+IDC+Research.&btnG= (Accessed: 27 September 2018).

Vilar, L. *et al.* (2013) 'Science of winning soccer: Emergent pattern-forming dynamics in association football', *Journal of Systems Science and Complexity*. Academy of Mathematics and Systems Science, Chinese Academy of Sciences, 26(1), pp. 73–84. doi: 10.1007/s11424-013-2286-z.

Walsham, G. (1997) 'Actor-Network Theory and IS Research: Current Status and Future Prospects', in *Information Systems and Qualitative Research*. doi: 10.1007/978-0-387-35309-8_23.

Walsham, G. (2014) 'Empiricism in interpretive IS research: a response to Stahl', *European Journal of Information Systems*. Palgrave Macmillan UK, 23(1), pp. 12–16. doi: 10.1057/ejis.2012.57.

Walsham, G. and Waema, T. (1994) 'Information systems strategy and implementation: a case study of a building society', *ACM Transactions on Information Systems*. ACM, 12(2), pp. 150–173. doi: 10.1145/196734.196744. Wang, X., Conboy, K. and Cawley, O. (2012) "Leagile" software development: An experience report analysis of the application of lean approaches in agile software development', *Journal of Systems and Software*. doi: 10.1016/j.jss.2012.01.061.

Ward, J. and Daniel, E. (2012) *Benefits management : how to increase the business value of your IT projects*. Wiley. Available at: https://www.wiley.com/en-gb/Benefits+Management:+How+to+Increase+the+Business+Value+of+Your+IT+Proj ects,+2nd+Edition-p-9781119993261 (Accessed: 17 July 2018).

Watson, H., Ariyachandra, T. and Matyska, R. J. (2001) 'Data Warehousing Stages of Growth', *Information Systems Management*, 18(3), pp. 42–50. doi: 10.1201/1078/43196.18.3.20010601/31289.6.

Watts, R. L. and Wruck, K. H. (1988) 'Stock prices and top management changes', *Journal of Financial Economics*, 20, pp. 461–492. doi: 10.1016/0304-405X(88)90054-2. Wendler, R. (2012) 'The maturity of maturity model research: A systematic mapping study', *Information and Software Technology*. Elsevier, 54(12), pp. 1317–1339. doi: 10.1016/J.INFSOF.2012.07.007.

Williams, A. M. and Hodges, N. J. (2005) 'Practice, instruction and skill acquisition in soccer: Challenging tradition', *Journal of Sports Sciences*, 23(6), pp. 637–650. doi: 10.1080/02640410400021328.

Wilson, R., Plumley, D. and Ramchandani, G. (2013) 'The relationship between ownership structure and club performance in the English Premier League', *Sport, Business and Management: An International Journal*, 3(1), pp. 19–36. doi: 10.1108/20426781311316889.

Winter, C. and Pfeiffer, M. (2016) 'Tactical metrics that discriminate winning, drawing and losing teams in UEFA Euro 2012®', *Journal of Sports Sciences*, 34(6), pp. 486–492. doi: 10.1080/02640414.2015.1099714.

Winter, S. G. and Szulanski, G. (2002) 'Replication of Organizational Routines: Conceptualizing the Exploitation of Knowledge Assets', *The Strategic Management of Intellectual Capital and Organizational Knowledge*.

Woods, C. T. *et al.* (2016) 'The relationship between game-based performance indicators and developmental level in junior Australian football: Implications for coaching', *Journal of Sports Sciences.* Routledge, 34(23), pp. 2165–2169. doi: 10.1080/02640414.2016.1210816. Wooster, B. (2013) *Soccer Analytics - Presented by Prozone*, *MIT Sloan Analytics Conference*. Available at: http://www.sloansportsconference.com/content/soccer-analytics-3/ (Accessed: 27 July 2018).

Wright, C. *et al.* (2013) 'The role of performance analysts within the coaching process: Performance Analysts Survey "The role of performance analysts in elite football club settings."', *International Journal of Performance Analysis in Sport*. Routledge, 13(1), pp. 240–261. doi: 10.1080/24748668.2013.11868645.

Wright, C. et al. (2016) 'Elite football player engagement with performance analysis', *International Journal of Performance Analysis in Sport*, 16(3), pp. 1007–1032. doi: 10.1080/24748668.2016.11868945.

Wright, C., Atkins, S. and Jones, B. (2012) 'An analysis of elite coaches' engagement with performance analysis services (match, notational analysis and technique analysis)', *International Journal of Performance Analysis in Sport*. Routledge, 12(2), pp. 436–451. doi: 10.1080/24748668.2012.11868609.

Wright, C., Carling, C. and Collins, D. (2014) 'The wider context of performance analysis and it application in the football coaching process', *International Journal of Performance Analysis in Sport*, 14(3), pp. 709–733. doi: 10.1080/24748668.2014.11868753.

Wright, P. M., McMahan, G. C. and McWilliams, A. (1994) 'Human resources and sustained competitive advantage: a resource-based perspective', *The International Journal of Human Resource Management*. Routledge , 5(2), pp. 301–326. doi: 10.1080/09585199400000020.

Wu, X. *et al.* (2014) 'Data Mining with Big Data', *IEEE Trans. on Knowl. and Data Eng.*, 26(1), pp. 97–107. doi: 10.1109/TKDE.2013.109.

Xie, L. *et al.* (2002) 'Structure analysis of soccer video with hidden Markov models', 2002 IEEE International Conference on Acoustics, Speech, and Signal Processing, 4(1), pp. 4096–4099. doi: 10.1109/ICASSP.2002.5745558.

Xu, C. *et al.* (2008) 'Using webcast text for semantic event detection in broadcast sports video', *IEEE Transactions on Multimedia*, 10(7), pp. 1342–1355. doi: 10.1109/TMM.2008.2004912.

Xue, Y. *et al.* (2017) 'Automatic Video Annotation System for Archival Sports Video', in 2017 IEEE Winter Applications of Computer Vision Workshops (WACVW). IEEE, pp. 23–28. doi: 10.1109/WACVW.2017.11. Yang, G. *et al.* (2018) 'Key team physical and technical performance indicators indicative of team quality in the soccer Chinese super league', *Research in Sports Medicine*. Routledge, 26(2), pp. 158–167. doi: 10.1080/15438627.2018.1431539.

Zambom-Ferraresi, F., Lera-López, F. and Iráizoz, B. (2017) 'And if the ball does not cross the line? A comprehensive analysis of football clubs' performance', *Applied Economics Letters*. Routledge, 24(17), pp. 1259–1262. doi: 10.1080/13504851.2016.1270408.

Zambon Ferraresi, F., Lera López, F. and García Cebrián, L. I. (2017) 'Sports Results Measurement and Efficiency in UEFA Champions League', *Athens j. sports*. Available at: https://zaguan.unizar.es/record/64476?ln=en (Accessed: 26 September 2018).

Zerguini, Y. *et al.* (2007) 'Impact of Ramadan on physical performance in professional soccer players.', *British journal of sports medicine*. BMJ Publishing Group, 41(6), pp. 398–400. doi: 10.1136/bjsm.2006.032037.

Zhou, S. and Zhang, F. (2017) 'Research on the Influence of Communication and Competition Performance between Football Coach and Team Member', 55(11), pp. 436– 441.

Appendix (Questionnaire)

KDMMFA Questionnaire

This section includes the questionnaire used in developing the KDMMFDA.

Questionnaire

 $\ensuremath{\mathbb{O}}$ by the researcher

Q1.2 In your football club what is your primary position - Please select the one is close to your responsibilities.

Team Manager - Head coach - Assistance Coach (Second Coach) - Supporting Coach (Fitness, Physical, Goalkeeper and so on) - General Manager of Football - Data/Video Analyst - Other - Please specify

Q2.1 This part of the Knowledge Discovery Maturity Assessment Model in Football is the Information Technology (IT), tools, technologies and infrastructure used by the team or club by the coaching team for collecting, obtaining and analysing the data.

In your analysis practices, which, tools and technologies you often use?

Use of recorded videos and clips in your analysis

Use of professional football databases

Use of purchased or obtained reports by professional football websites

Use of LPS: Local Positioning System and GPS: Global Positioning System, technologies

Use of football field (pitch) cameras

Use of the body sensors

Use other sports wearable devices

Use specialised analytics software (e.g. SPSS and R) to enable you develop new algorithms (e.g. ANOVA, Correlations analysis and t test) Use of artificial intelligence analytics software (e.g. R software, Python) to develop new models (e.g. Neural Network and Genetic Algorithms)

Q2.2 Any comments about the IT environment?

Q3.1 The next part of the KDMAMF is focusing on the Data Analyst or (Video Analysts) Competency in analysing football data.

Q4.1 Data analyst(s) availability at the club or the team as:

Team of analysts - Part time team of analysts - One analyst - One part time analyst - No dedicated analysts

Q4.2 The scope of the following questions is related to the Data Analyst or Video Analysts competences. Where: 5= Very Competent - 4= Competent - 3= Average - 2= Weak competence - 1= Very weak competence Data analyst is able to import relevant data from professional databases to include them in spreadsheet applications or software (i.e. Microsoft Excel or Apple Numbers).

Data analyst is able to code the match key performance indicators and propose opponent teams' analysis.

Data analyst is able to use spreadsheet (i.e. "Microsoft Excel" and "Apple Numbers") in comparing players performance and team's performance.

Data analyst is able to use "Excel" for statistical modelling.

Data analyst is able to use sophisticated data analytic applications (i.e. SPSS and R) for doing structured statistical analysis such as (i.e. ANOVA, Correlations analysis and t test) for testing proposed relations.

Data analyst is able to use sophisticated data analytic applications (i.e. R software or Pythons) for utilising artificial intelligence (i.e. Neural Network and Genetic Algorithms) for discovering new pattern in the data.

Data analyst knowledge and expertise of the football concepts, strategies, techniques, and approaches.

Data analyst profound experience in football strategies (training, transferring, and match strategies).

Data analyst ability to interpret the reports and provide useful meaning of the data.

Data analyst ability to propose constructive and insightful football match analytical ideas from the data and data resources available.

To define the communication requirements from the team managers or head coach (i.e. which reports are required and when to hand them). Data analyst ability to produce reliable reports (level of error in reporting data is accepted).

Data analyst ability to define who shall be communicated, by what, and when (e.g. players, media, other members in the coaching team).

Data analyst ability to intemperate and explain the technical report with an accepted level of interpretation (i.e. not too much subjectivity, not too few explanation and interpretation).

Q4.3 Any comments about the Data analyst competencies

Q5.1 The following part of the KDMAMF is focusing on the Team Manager or Head Coach competency in the analysis practices of football data.

The Following questions are related to the Coach

Team manager/Coach is a fan of data in planning

Team manager/coach believes in the importance of statistics and analysis in improving the planning and developing match strategies

Team manager knowledge of statistical concepts and basic analytic models to compare between metrics (i.e. significance level and "t" test) Team manager knowledge of statistical models to explain and correlate different measures (i.e. when and why to use correlational analysis, regression, ANOVA, and cluster analysis)

Team manager knowledge of statistical models to discover new insightful patterns in the data (i.e. when and why to use Neural Network, cluster analysis, pattern analysis and genetic algorithms)

Q5.2 Any comments about the Team Manager / Head Coach environment and competencies?

Q6.1 The next part of the KDMAMF is focusing on the Value Co-Creation analysis practices within the coaching team for analysing football data.

The Following questions are reflecting the practices in the KD value co-creation process.

There are set of questions to work on in order to reach insights and vision of each meeting.

There are set of stories to work on in order to reach insights and vision of each meeting.

A set of predefined models to describe the problem and integrate them in the stories.

A set of predefined tools to describe the problem and integrate them in the stories.

Well documentation of stories to be used for lessons learned and improving the story models.

There are continuous and periodic meetings between the team manager/coach and analyst.

There is clear schedule of meetings with predefined acceptance criteria for what will be discussed and what will be handed.

A clear framework for the meetings so that Knowledge discovery process is integrated in the sprint process.

Lessons learned are taking into consideration for improving sprints.

There are meetings between the team manager/coach and analyst.

The communication channels are defined (Who, what, when, and how).

Clear communication channels with acceptance criteria of what the information require and why they are required.

Knowledge Discovery is integrated in the communication process through setting communication process quality (user questions, stories, stories/question mapping, and acceptance criteria).

Lessons learned are documented in the communications and taking into consideration for improving communications practices.

Q6.2 Any comments about the KD value co-creation process?

Q7.1 The following part of the KDMAMF is focusing on the analysis practices, models and outcomes within the coaching team.

The Following questions are reflecting the analysis outcomes of the analysis process and practices.

Use of videos and clips in analysis.

Use secondary data reports such as STATS, InStats and OPTA.

Use of coding systems/techniques to track different players' performance.

Use of analytic models to measure the significance of the differences between players such as "t" test.

Use of Advanced analytic tools to do context analysis.

Use of advanced analytic tools to do simulation analysis.

Use of advanced analytic tools to do structured analysis.

Q7.2 Any comments about the Analysis environment and outcomes?

Q8.1 The next part of the KDMAMF is to explore the awareness and use of football Key Performance Indicators (KPIs) during the analysis used by the coaching team within the team or club.

Q9.1 Physical KPIs - are those physiological and fitness measures for the players' abilities. Some of them are traits cannot be changed such as the height and ambidexterity while others can be improved by training such as speed, low/moderate/high intensity running, and recovery rate.PhysicalKPIsareclassifiedintospeed,movementanddistance.Do you agree with this definition?

Strongly Disagree - Disagree - Neither agree nor disagree - Somewhat agree - Strongly agree

Q9.2 Which indicators you most often currently use for evaluating physical performance? and if used or not what are their level of importance?

Player Speed: Average player speed per match in different modes (low, moderate, high speed and sprint) with and without the ball. Player speed in different running categories (low, moderate, high speed, and sprint) with the ball.

Player speed in different running categories (low, moderate, high speed, and sprint) without the ball.

Distance covered: The total distance run during the match.

Distance covered with the ball.

Distance covered without the ball.

Distance covered in different speed categories (low, medium, high speed, and sprint).

The maximum speed of shooting the ball.

The maximum distance of a throw in.

The maximum height of aerial action (i.e. jumping for header).

Q9.3 Any additional comments about the use of Physical KPIs in player or team analysis?

Q9.4 **Technical KPIs** are different individual football physical competencies required to control or to regain the control, to direct the ball, and to build constructive movements during the match. They are classified into off the ball competences (ability to regain the control) and on the ball competences (ability to direct the ball towards a constructive movement) *Do you agree with this definition*?

Strongly Disagree - Disagree - Neither - Agree - Strongly Agree

Q9.5 Which indicators you most often currently use? and if not, will you use it? - and what are their level of importance?

% Success rate of the attempts to regain control on the ball (e.g. tackles or interceptions).

% Success rate to take on (e.g. dribbling).

% the players ability to win aerial interactions.

Significant change in the speed with the ability to build a constructive pass or goal. i.e. 80% change in speed within 1 min leading to successful attack.

% of successful shots towards the goal in different situations (in plenty area, outside the plenty area, when marked by 1 person, by more than 1 person).

% of successful free kicks towards the goal from different zones (e.g. right, left, middle zones).

Number of innovative movements in the match (new dribbling, tackles, passes or movement).

Q9.6 Any additional comments about the use of Technical KPIs in player or team analysis?

Q9.7 Tactical KPIs - are defined as metrics to measure the players' ability to position himself in the pitch effectively and efficiently in such a way the probability of passing, possessing, scoring and intervening are improved. The tactical KPIs are measured by player, unit of play (set of players), tactical lines (e.g. attacking, defending, or midfield line), or by the team. They are classified into passes, possession, and playing style.

Do you agree with this definition?

Strongly Disagree - Disagree - Neither agree nor disagree - Agree - Strongly Agree

Q9.8 Which indicators you most often currently use for evaluating passing and intercepting performance? and if not, will you use it?

Overall all passes performance index: % of successful passes per match (e.g. spontaneous passes, 1 to 1 passes, unit passes, constructive passes, and long passes).

% of successful spontaneous passes: SP is defined as the passes without having a clear intention to build a constructive attack (i.e. due to pressure from opponents).

% of successful 1 to 1 passes: 1 to 1 passes is the several passes between two players only aiming to construct an attack, penetrate defensive line or shift the direction of the play.

% of successful unit passes: unit passes are the several repeated passes between more than two players.

% of successful constructive passes: constructive passes are more than 2 passes with more than 2 players aiming to construct an attack or shift the play direction. (e.g. second ball and third ball).

% of successful long passes: Long passes is a movement of the ball from a zone to another zone or from tactical lines (e.g. from back to front, from the left side to the right side or from defending to attacking line). E.g. successful crosses/ counter attack.

% of successful interceptions from short passes or long passes (crosses or counter attacks).

Q9.9 Which indicators you most often currently use for evaluating team playing style? and if not, will you use it?

Team/player minutes played with or without the ball.

Time played in offence, defence, and midfield.

Ball recovery time: Average time required to regain the ball.

Total Possession.

Distances between attackers and defenders: The average distance between the attacking and defending lines.

Maintaining distance between players (close down space): The % of the time that distance between players within the ball range is lower than the coach defined space in each zone.

Offside (Tolerance) Management (such as % of the successful deliberate offside (avoidance) and % of the successful avoiding opponent deliberate offside.

Q9.10 Which indicators you most often currently use for evaluating player positioning performance? and if not, will you use it?

% of successful (constructive change) changes of players' positions in the pitch during the match.

The duration of a player being in specific zone.

Player density: % of time played in the player specified zone.

Marking (man to man marking – Zonal marking) : The ability of a player to mark an opponent's players – or zonal area of the pitch.

Q9.11 Any additional comments about the use of Tactical KPIs in player or team analysis?

Q9.12 Psychological KPIs - is the ability to play in the standard performance under different psychological pressures. I.e. psychological resilience indicator.

Do you agree with this definition?

Strongly disagree - Somewhat disagree - Neither agree nor disagree - agree - Strongly agree

Q9.13 Which indicators you most often currently use for evaluating Psychological player performance? and their level of importance?

Resilience: % of change in the performance indicator in different contexts (e.g. opponents, home/away, fan support, get paid well).

Ethical indicators: number of cards or injuring other players.

Discipline indicators: % Body language/facial expressions of anger against the coach/referee decisions.

Manipulative indicators: number of free kicks against opponent.

Q9.14 Any additional comments about the use of Psychological KPIs in player or team analysis?



1- The role of physical/fitness indicators on the tactical indicators.

2- The role of physical/Fitness indicators on the technical indicators.

3- The role of physical/fitness indicators on match performance.

4- The role of technical indicators on tactical indicator.

5-The role of technical indicators on match performance.

6- The role of match tactical indicators on match result.

7- The role of psychological indicators on physical indicators.

8- The role of psychological indicators on technical indicators.

9- The role of psychological indicators on tactical indicators.

10- The role of psychological indicators on match performance.

Q11.4 Would you like to add any comments about the BSC?

Q12.1 In which country do you currently work?

Q13.1 Current Football/Soccer Team

Q13.2 Please add your email address - A report will be sent to you after analysing the Interview / Online survey.

Q13.3 First and last names. To contact you and address you in the report - if preferred.